

DEVELOPMENT OF AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER FIVE CHILDREN IN SIDAMA REGIONAL STATE, ETHIOPIA

By DERESE DESALEGN BUTA Submitted in accordance with the requirements for the degree of

DOCTOR OF PHILOSOPHY

In the subject Public Health

UNIVERSITY OF SOUTH AFRICA SUPERVISOR: Pof LI ZUNGU

CO-SUPERVISOR: Dr AYUK BETRAND

NOVEMBER 25/ 2021

Authored by

Derese Desalegn Buta *, Li Zungu, Ayuk Betrand

University of South Africa.

Published Date January 29, 2024

Published in the Journal of

Psychology and Mental Health Care

Auctores Publishing, LLC

16192 Coastal Highway

Lewes, DE 19958,

United States of America



······

Open Access

Psychology and Mental Health Care

Derese Desalegn Buta *

Research Article

Development of Awareness and Prevention Strategies on Pneumonia Among Under Five Children in Sidama Regional State, Ethiopia

Derese Desalegn Buta *, Li Zungu, Ayuk Betrand

University of South Africa.

*Corresponding Author: Derese Desalegn Buta, University of South Africa.

Received date: January 02, 2024; Accepted date: January 19, 2024; Published date: January 30, 2024.

Citation: Derese D. Buta, Li Zungu, Ayuk Betrand, (2024), Development of Awareness and Prevention Strategies on Pneumonia Among Under Five Children in Sidama Regional State, Ethiopia, *J. Psychology and Mental Health Care*, 8(1): **DOI:**10.31579/2637-8892/250

Copyright: © 2024, Derese Desalegn Buta. This is an open-access article distributed under the terms of The Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Abstract

Purpose: Aim of the study was to develop awareness and prevention strategies on pneumonia among under-five children in Sidama Regional State, Ethiopia.

Methods: Current study used mixed method approach followed with Delphi group comments and suggestions to attain the research goal.

Community based cross sectional study design and multistage sampling method were employed to recruit 843 mothers with under five children in quantitative phase. Strucured questionaries used to collect data. Data were entered in Epi data version 4.2.2 and exported to SPSS version 25; further analysis. Descriptive statistics, binary logistic and multivariate analysis were applied.

Furthermore, 6 FGDs were conducted among 40 mothers. Qualitative data were collected by using FGD guide, audio recorded, field note taken, and voice transcription, verbatim has done, three themes and six sub themes identified. Analysis supported by Atlas ti soft ware.

Result: The over all prevalence of childhood pneumonia was 22.2%. Factors like Medium and large family size, lack of formal education, birth intervals <2 years, age of 7-36 months, history of respiratory disease, disease history among children and history of measles disease last year prior to data collection, absence of latrine, lack of separate house for animal, using charcoal, chimney in the house, absence of window, severely underweight, lack of receiving vitamin A and immunization were found as significantly contributing factors of pneumonia.

Results also showed that 63.6% of respondents had poor knowledge, 38.2% with negative attitude and 42.5% improper practices towards childhood pneumonia prevention and control.

Majority of FGD discussants perceived as pneumonia is dangerous and seek health care, but showed gap of identifying the exact causes of childhood pneumonia, transimision way, sign and symptom as well as prevention mechanism.

Conclusion: Current study confirmed that, there is substainatially high prevalence of child hood pneumonia and remained as public health problem. Likewise poor knowledge and practice of communities' were also obtained towards pneumonia prevention and control mechanisms. Thus, researcher developed awareness and prevention strategies to assist and strengthen the prevention and control of childhood pneumonia. Health planners and managers recommended to utilise the strategies to reduce child morbidity and mortality.

key terms: attitude, awareness, ethiopia, knowledge, pneumonia, practice, prevention, sidama regional state, strategies, under-five children,

Competing interests

All authors declare that they have no competing interests.

Funding

The financial support for the current study was obtained from University of South Africa and NORAD project of Hawassa University and Derese Desalegn Buta is an author who received an award. The funded agency did not take part in study design, data collection and manuscript preparation process.

Development Of Awareness and Prevention Strategies on Pneumonia Among Under Five Children in Sidama Regionla State, Ethiopia

I declare that the above thesis is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references.

I further declare that I submitted the thesis to originality checking software and that it falls within the accepted requirements for originality.

I further declare that I have not previously submitted this work, or part of it, for examination at Unisa for another qualification or at any other higher education institution.



Date November 25/2021

DEVELOPMENT OF AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER FIVE CHILDREN IN SIDAMA REGIONAL STATE, ETHIOPIA

STUDENT NUMBER: 62136313 NAME OF STUDENT: DERESE DESALEGN BUTA DEGREE: DOCTOR OF PHILOSOPHY: DEPARTMENT OF PUBLIC HEALTH, UNIVERSITY OF SOUTH AFRICA SUPERVISOR: Prof LI ZUNGU CO-SUPERVISOR: Dr AYUK BETRAND

Acnowledgement

First and foremost infinite thanks goes to Almighty God who gave me vision with full strength, without him everything might be meaningless even my life too. His mercy, love and support make me joyful within each ups and downs.

My heartfelt thanks and appreciation goes to my dearest supervisor Professor LI Zungu for her tireless advice, guidance and support as well as understanding without her help this might not be happen really. I am proud of her patience and on time responses through the whole process of my research work.

My especially gratitude goes to my co-advisor Dr Ayuk Betrand for his amazing technical advice and guidance through all thesis chapters which I learnt a lot from his valuable comments and suggestions.

I am also thankful to UNISA for the educational chance, assigning advisor, granting the ethical clearance and the bursary offered. The UNISA staffs at all level including Ethiopian regional office deserve my appreciation and thanks for their unreserved commitment and support they did through each unclear steps and error without their involvement, which might take a time even to understand.

My thanks and love goes to my wife Ms Elsabet Shudura and my lovely daughter Kayo (Kukusha/Buchula/Mare) for your understanding and support during my study time. Dears I knew that my full engagement in the course work precluded you my time but your thoughtfulness and praying make me happier rather thinking the worries. Your understanding and patience supported me to manage and handle the every aspects safely. Much more love!

I would also like to recognize and acknowledge Hawassa University, Sidama development association, former Sidama Zone Administration Office and health department for financial and logistic support.

My thanks and appreciation goes to my friends and colleagues Mr Mengesha Boko, Mr Fitsum Dangura, Teshale Tamirat, Nafkot Birhanu, Mohamed, Getu Mulatu, Tadese Mekuria and Abiyot Fiseha for your unforgettable support and motivation you showed.

My unique and personal thanks goes to data collectors and supervisors as well as study participants for your especially commitment and engagement in my study, without your involvement and contributions this might not be happened.

I would like to appreciate and recognize scholars and experts involved in Delphi panels with sharing their plenty of experiences and practices through strategy development.

My gratitude and thanks goes to Mr Amanuel Yoseph, Dr Dejene Hailu and Dr Hunachew Beyene for your unreserved technical guidance and advice through my research work.

My especially thanks and recognition goes to all my families, brothers, sisters, father in law and mother in law. My father Desalegn Buta and my mother Almaz Echero I never forget your best wishes and praying always, which helped me yet. I pray God for your long live and health too.

Mr Solomon G/Medihin, Ms Rinnie Matlou, Dr Tsegaye Asres and Dr Tsegaye Alemu also deserve my acknowledgement for their unreserved contributions on language edition, formatting and technical assistance work respectively.

I would also like to acknowledge those all my friends who have been encouraging and supporting me in each steps, especially Mr Teklu Birhanu, Gosaye Kebato and Yohanes Esayas.

Last but not least my thanks goes to Tadese enjory, Central and Rori hotels and Gezhegn and Elfinesh resort for their patience and understanding while consuming Wifi services.

Dedication

This thesis work is dedicated to our son who went to heaven in his 8th day of birth. God is good all the time!

Table of Content

Error! Bookmark not defined.	
4	
12	
13	
14	
e study 15	
 DN	
O TO THE RESEARCH PROBLEM	
DF PROBLEM	
OF RESEARCH	
E OF STUDY	
MITATIONS	
TA-THEORETICAL GROUDING/THEORETICAL FRAMEWORK	18
retical assumptions	18
il framework	18
)F KEV CONCEPTS	
DESIGN AND METHODS	
ab Desian	
n Dongn nh sotting	
	Error! Bookmark not defined. 4 12 13 14 e study 15 N To the research problem of Research of Research f Key Concepts b Setting

1.10.3 Research population	19
1.10.4 Data collection procedure	
1.10.5 Data analysis	
1.10.6 Ethical considerations	
1.11 STRUCTURE OF THE DISSERTATION	
1.12 CONCLUSSION	
CHAPTER TWO 20	
LITERATURE REVIEW 21	
2.1 INTRODUCTION	
2.2 OVERVIEW OF PNEUMONIA	
2.3 BURDEN OF CHILDHOOD PNEUMONIA	
2.4 LITERATURE REVIEW FINDINGS ON FACTORS ASSOCIATED WITH CHILDHOOD PNEUMONI	A 22
2.4.1 Socio-demographic and economic factors contributing pneumonia among under five children	
2.4.1.1 Sex of the child as contributing factors of childhood pneumonia	
2.4.1.2 Household economic status as factor for the occurrence of childhood pneumonia	
2.4.1.3 Child's age as pneumonia contributing factors	
2.4.1.4 Parental education as factors of childhood pneumonia	
2.4.1.5 Family size as factor of childhood pneumonia	
2.4.2 Environmental factors associated with childhood pneumonia	
2.4.2.1 Solid fuels and indoor air pollution as contributing factors of childhood pneumonia	
2.4.2.2 Smoking as associated factors for childhood pneumonia	
2.4.2.3 Overcrowding and poor ventilation as contributing factors of pneumonia	
2.4.2.4 Housing conditions of households as contributing factors of childhood pneumonia	
2.4.2.5 Hygiene and sanitation practices as factors of childhood pneumonia	
2.4.3 Nutrition and immunization related factors associated with childhood pneumonia	
2.4.3.1 Malnutrition as contributing factors of childhood pneumonia	
2.4.3.2 Breast feeding and micronutrient deficiency as contributing factors of childhood pneumonia	
2.4.4 Immunization and related factors of childhood pneumonia	
2.5 KNOWLEDGE, ATTITUDES AND PRACTICES OF MOTHERS/CAREGIVERS TOWARDS CHILDHO PREVENTION AND CONTROL	OOD PNEUMONIA 25
2.5.1 Knowledge of mothers/caregivers towards childhood pneumonia prevention and control and its asso	ciated factors 25
2.5.2 Attitude of mothers/care givers towards childhood pneumonia prevention and control and its associa	ted factors 25
2.5.3 Practice of mothers/care givers towards childhood pneumonia prevention and control and its association association and control and its association and control and its association association and control and its association association as a specific as	ated factors 25
2.6 THE AVAILED CHILDHOOD PNEUMONIA PREVENTION AND CONTROL STRATEGIES	
2.6.1 Protective interventions for childhood pneumonia prevention and control	
2.6.1.1 Exclusive breastfeeding	
2.6.1.2 Adequate complementary feeding and continued breastfeeding	
2.6.1.3 Vitamin A supplementation:	
2.6.2 Preventative interventions for childhood pneumonia prevention and control	
2.6.2.1 Immunization	
2.6.2.2 Sanitation and hygiene	

25 25 25

	2.6.2.3	Reduced household air pollution	
	2.6.2.4	HIV prevention	
2.	6.3 T	reatment interventions for childhood pneumonia prevention and control	
	2.6.3.1	Improved care seeking and referral	
	2.6.3.2	Timely and accurate diagnosis mechanism to prevent childhood pneumonia deaths	
	2.6.3.3	Appropriate use of antibiotics	
	2.6.3.4	Oxygen therapy	
2.7	CONC	LUSSIONS	

CHAPTER 3 29

THEORETICAL FOUNDATIONS OF THE STUDY 29

3.1 INTRODUCTION	
3.2 A FRAMEWORK	
3.3 THEORETICAL FRAMEWORK	
3.4 CONCEPTUAL FRAMEWORK	
3.4.1 Health Belief Model (HBM)	
3.4.1.1 Perceived Susceptibility	
3.4.1.2 Perceived Seriousness/Severity	
3.4.1.3 Perceived Benefits	
3.4.1.4 Perceived Barriers	
3.4.1.5 Individual's perceptions	
3.4.1.6 Modifying Factors	
3.4.1.7 Benefits of taking the preventive measures	
3.4.2 Global Action Plan for Prevention and control of Pneumonia Framework	
3.5 CONCLUSSIONS	

CHAPTER 4 31

RESEARCH DESIGN AND METHODS 31

4.1 INT	RODUCTION	
4.2 RES	SEARCH DESIGN	
4.2.1	Quantitative Research	
4.2.	1.1 Cross-sectional study design	
4.2.2	Qualitative Research	
4.2.3	Mixed Research method	
4.3 RES	SEARCH METHOD	
4.3.1	Study area/setting	
4.3.2	Study Population	
4.3.3	Study sample and sampling techniques	
4.3	3.1 Sample size calculation	
4.3	3.2 Sampling techniques/procedures	
4.3	3.3 Eligibility criteria for being a respondent for interview	
4	4.3.3.3.1 Inclusion criteria	

4.3.3.3.2 Exclusion criteria	
4.3.3.4 Study Variables	
4.3.3.4.1 Dependent Variables	
4.3.3.4.2 Independent Variables	
4.3.4 Data collection process and tool	
4.3.4.1 Data collection approach	
4.3.4.2 Research data collection tool	
4.3.4.3 Training provided for data collectors and supervisors	
4.3.4.4 Pilot study	
4.3.4.5 Data collection process	
4.3.5 Ethical considerations	
4.3.5.1 Protecting the Rights of Institutions	
4.3.5.2 Protecting the rights of respondents	
4.3.5.3 Scientific integrity	
4.3.6 Processing and analysis of data	
4.3.6.1 Data processing	
4.3.6.2 Data analysis	
4.3.6.2.1 Analysis of quantitative part	
4.3.6.2.2 Analysis of qualitative part	
4.3.6.2.3 Analysis of Delphi method data	
4.3.7 Validity, Reliability and Trustworthiness	
4.3.7.1 Validity	
4.3.7.2 Reliability of the research instrument	
4.3.7.3 Trustworthiness	
4.3.7.3.1 Credibility	
4.3.7.3.2 Dependability	
4.3.7.3.3 Confirm-ability	
4.3.7.3.4 Transferability	
4.3.7.3.5 Authenticity	
4.4 CONCLUSSION	
CHAPTER 5 41	
ANALYSIS AND PRESENTATION OF THE RESEARCH FINDINGS 41	
5.1 INTRODUCTION	
5.2 DATA COLLECTION, MANAGEMENT AND ANALYSIS	
5.3 RESEARCH RESULT FOR QUANTITATIVE PART OF STUDY	
5.3.1 Response rate and residence of respondents	
5.3.2 Socio-demographic and economic characteristics of respondents	
5.3.2.1 Wealth index of households	
5.3.3 Socio-demographic characteristics of surveyed child	

5.3.4.1 Availability of latrine and solid waste management	
5.3.4.2 Housing characteristics of the respondents'	
5.3.4.3 Households energy source	
5.3.5 Nutrition related characteristics of surveyed child	
5.3.6 Vitamin A supplementation, deworming and immunization status of child	
5.3.7 Behavioural and disease status of child's family	
5.3.8 Prevalence of childhood pneumonia and associated factors in Sidama Regional State	
5.3.8.1 Prevalence of pneumonia among under-five children	
5.3.8.2 Bivariate analysis of factors associated with pneumonia among children under- five years old	
5.3.8.2.1 The association between the socio-demographic and economic factors and the occurrence of pneumonia among under year's children	-5
5.3.8.2.1.1 Households' socio-demographic and economic characteristics and association with the occurrence of pneumonic among under-5 year's children	ı
5.3.8.2.1.2 Socio-demographic characteristics of child and association with the occurrence of pneumonia among under-5 ye old children 50	ar's
5.3.8.2.2 Bivariate analysis of behavioural and disease history of household members and occurrence of pneumonia among und five year's children	er-
5.3.8.2.3 Bivariate analysis of environmental characteristics of households and the occurrence of pneumonia among under-five year's old children	
5.3.8.2.4 Bivariate analysis of nutritional and immunization characteristics of child and the occurrence of pneumonia among un five year's old children	ıder-
5.3.8.4 Multivariate analysis of factors and association with the occurrence of pneumonia among under-five year's old children 53	t
5.3.8.4.1 Multivariate analysis of socio-demographic and economic factors and the occurrence of pneumonia among under-five children 53	
5.3.8.4.2 Multivariate analysis of behavioural and disease history of households and the occurrence of pneumonia among under children 54	r-five
5.3.8.4.3 Multivariate analysis of environmental factors of households and the occurrence of pneumonia among under-five child 56	lren
5.3.8.4.4 Multivariate analysis of nutritional and immunization related factors and the occurrence of pneumonia among under-j children 57	īve
5.3.9 Knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control and associated factors w integration of qualitative finding	vith
5.3.9.1 Heard about pneumonia and source of information	
5.3.9.2 Descriptive analysis of knowledge of mothers/caregivers on sign and symptom of pneumonia with integration of qualitat finding 58	ive
5.3.9.3 Knowledge of mothers/caregivers on causes of pneumonia with integration of qualitative findings 59	
5.3.9.4 Knowledge of mothers/caregivers towards transmission method of pneumonia with integration of qualitative findings 60	
5.3.9.5 Knowledge of mothers/caregivers on prevention method of pneumonia with integration of qualitative findings 60	
5.3.9.6 Overall level of knowledge of mothers/caregivers towards childhood pneumonia prevention and control61	
5.3.9.7 Attitude of mothers/caregivers towards pneumonia prevention and control with integration of qualitative findings 61	
5.3.9.8 Overall attitude level of mothers/caregivers towards prevention and control of childhood pneumonia 62	
5.3.9.9 Practices of mothers/caregivers towards pneumonia prevention and control with integration of qualitative findings 62	
5.3.9.10 Overall level practice of mothers/caregivers towards childhood pneumonia prevention and control 63	
5.3.9.11 Factors associated with KAP of mothers/care givers towards childhood pneumonia prevention and control 64	

5.3.9.11.	1 Factors associated with knowledge of mothers/caregivers towards childhood pneumonia prevention and contr 64	ol
5.3.9.11.	2 Factors associated with attitude of mothers/caregivers towards childhood pneumonia prevention and control	65
5.3.9.11.	3 Factors associated with the practice of mothers/caregivers towards childhood pneumonia prevention and cont 66	rol
5.4 CONCLUS	SIONS	
CHAPTER 6 67		
DISCUSSION OF T	THE RESEARCH FINDINGS 67	
6.1 INTRODUC	CTION	
6.2 PREVALE	NCE OF CHILDHOOD PNEUMONIA AND ASSOCIATED FACTORS	
6.2.1 Preva	lence of pneumonia among under five children in Sidama Regional State	
6.2.2 The a	ssociation between socio-demographic and economic factors and the occurrence of childhood pneumonia	67
6.2.3 The a 68	ssociation between behavioural and history of disease related factors and the occurrence of childhood pneumon	nia
6.2.4 The a	ssociation between environmental factors and the occurrence of childhood pneumonia	
6.2.5 The a	ssociation between nutritional and immunization related factors and the occurrence of childhood pneumonia	69
6.2.6 Level	of knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control69	
6.2.7 The a	ssociation of factors with knowledge of mothers/caregivers towards pneumonia prevention and control69	
6.2.8 The a	ssociation of factors with attitude of mothers/caregivers towards pneumonia prevention and control 70	
6.2.9 The a	ssociation of factors with practice of mothers/caregivers towards pneumonia prevention and control 70	
6.3 CONCLUS	<i>SION</i>	
CHAPTER 7 70		
CONCLUSIONS, R.	<i>ECOMMENDATIONS AND LIMITATIONS</i> 70	
7.1 INTRODUC	CTION	
7.2 SUMMARY	AND INTERPRETATION OF THE RESEARCH FINDINGS	
7.2.1 Socio-de	mographic and economic characteristics of respondents	
7.2.2 Socio	- demographic characteristics of child	
7.2.3 Preva	lence of pneumonia among under-five children71	
7.2.4 Key fo	actors associated with the occurrence of pneumonia among under-five children	
7.2.4.1 Fi 71	ndings related to socio-demographic and economic factors and association with the occurrence of childhood pn	eumonia
7.2.4.2 Fii five childrei	ndings related to behavioural and disease related history of households and the occurrence of pneumonia amor n	ıg under-
7.2.4.3 Fi	ndings related to environmental factors and occurrence of pneumonia among under-five children71	
7.2.4.4 Fi 71	ndings related to nutritional and immunization related factors and occurrence of pneumonia among under-five	children
7.2.5 Know	ledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control	71
7.2.6 Factors of 71	associated with knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and cont	rol
7.2.6.1 Fa children 71	ctors associated with knowledge of mothers/caregivers towards pneumonia prevention and control among unde	er five
7.2.6.2 Fa children 71	ctors associated with attitude of mothers/caregivers towards pneumonia prevention and control among under f	ive

7.2.6.3 Factors associated with practice of mothers/caregivers towards pneumonia prevention and control children 71	among under five
7.3 CONCLUSIONS	72
7.4 RECOMMENDATIONS	72
7.4.1 Recommendations for woreda health offices, health centers and health posts	72
7.4.2 Recommendations for Sidama Regional Health Bureau	72
7.4.3 Recommendations for health sector partners and professional associations	72
7.4.4 Recommendations to the Federal Ministry of Health	72
7.4.5 Recommendations for further researchers	72
7.5 CONTRIBUTIONS OF THE STUDY	73
7.6 LIMITATIONS OF THE STUDY	73
7.7 CONCLUDING REMARKS	73
CHAPTER 8 73	
DEVELOPMENT OF AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER-F	IVE CHILDREN IN
SIDAMA REGIONAL STATE 73	
8.1 INTRODUCTION	73
8.2 GUIDING PRINCIPLES OF THE STRATEGIES	73
8.3 VALIDATION MECHANISM	73
8.4 PURPOSE AND OBJECTIVES OF THE STRATEGIES	73
8.4.1 Purpose of the strategies	73
8.4.2 Objective of the strategies	73
8.5 SCOPE OF THE STRATEGIES	73
8.6 BASIS FOR THE DEVELOPMENT OF THE STRATEGIES	73
8.6.1 Delphi method	74
8.7 INTEGRATION OF THE FINDINGS WITH THE THEORETICAL FOUNDATION	74
8.7.1 Individual's perceptions	74
8.7.2 Modifying factors	74
8.7.3 Benefits of taking the preventive measures	75
8.8 AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER-FIVE CHILDR REGIONAL STATE	EN IN SIDAMA 75
8.8.1 Development of strategies	76
8.8.1.1 Strategy one: Health communication and promotion	76
8.8.1.1.1 Community mobilization	76
8.8.1.1.2 Health education	77
8.8.1.1.3 Advocacy	77
8.8.1.1.4 Social marketing	78
8.8.1.2 Strategy two: Protecting children from being diseased from pneumonia through adequate nutrition	ι 78
8.8.1.3 Strategy three: Preventing children from being diseased by pneumonia through timely immunization environment.	o n and healthy 78
8.8.1.4 Strategy four: Treating children who become ill with pneumonia	78
8.8.1.5 Strategy five: Improving the utilization of maternal and child health services	78
8.8.1.6 Strategy six: Engaging multi-sector partnership to facilitate and contribute pneumonia prevention	and control 78

<u>ه</u> ۵	8.8.1.7 Strategy seven: Strengthening health system to deliver strong health care in order to com	nbat pneumonia79
8.9	CONCEPTUAL FRAME WORK FOR AWARENESS AND FREVENTION STRATEGIES OF C 79	HILDHOOD PNEUMONIA
8.9	PRIORITY AREAS FOR ACTION	
8.10	MONITORING AND EVALUATION	
8.11	IMPLIMENTATION STRATEGIES	
8.12	STRATEGIES DISSEMINATION PLAN	
8.13	CONCLUSIONS	
REFER	RANCES 83	

ANNEXURES 88

List of Tables

Table 5. 1: Socio-demographic and economic characteristics of respondents in Sidama Regional State: 2020 (N=843) 42
Table 5. 2: Socio-demographic characteristics of surveyed children in Sidama Regional State: 2020 (N=843)
Table 5. 3: Availability of latrine and solid waste management in Sidama Regional State, Ethiopia: 2020 (N=843)
Table 5. 4: Housing characteristics of respondents in Sidama Regional State, Ethiopia: 2020 (N=843)
Table 5. 5: Households' energy sources and related characteristics in Sidama regional state, Ethiopia: 2020 (N=843)
Table 5. 6: Child breastfeeding and complementary feeding practices among mothers in Sidama regional state, Ethiopia: 2020, (N=843)46
Table 5. 7: Vitamin A supplementation, deworming and immunization status of surveyed children in Sidama regional state, Ethiopia: 2020 (N=843)
Table 5. 8: Behavioural and disease history of child's family member in Sidama Regional State, Ethiopia: 2020, (N=843) 48
Table 5. 9: Bivariate analysis of mother's/caregivers' socio-demographic and economic characteristics of mothers/care givers in Sidama regional state: 2020 49
Table 5. 10: Socio-demographic characteristics of child and the occurrence of pneumonia among under five children in Sidama Regional State: 2020 50
Table 5. 11: Behavioural and disease history and the occurrence of pneumonia among under-five year's children in Sidama Regional State: 2020
Table 5. 12: Bivariate analysis of environmental characteristics of households in Sidama Regional State: 2020
Table 5. 13: Bivariate analysis of nutritional and immunization related status of children and the occurrence of pneumonia among under-fiveyear's children in Sidama Regional State: 2020
Table 5. 14: Omnibus Test of model coefficients for the socio-demographic and economic variables and the occurrence of childhood pneumonia,in Sidama Regional State, Ethiopia: 2020
Table 5. 15: Multivariate analysis of household's socio-demographic and economic factors and occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020
Table 5. 16: Omnibus Test of model coefficients for the behavioral and disease history of households' family member and the occurrence ofchildhood pneumonia, in Sidama Regional State, Ethiopia: 2020
Table 5. 17: Hosmer and Lemeshow Test for the socio-demographic variables and the occurrence of childhood pneumonia, in Sidama Regional State, Ethiopia: 2020
Table 5. 18: Multivariate analysis of behavioural and disease history of households and the occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843)

Table 5. 19: Multivariate analysis of environmental factors and the occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843) 56
Table 5. 20: Multivariate analysis of nutritional and immunization related factors and the occurrence of pneumonia among under-five children inSidama Regional State, Ethiopia: 2020 (N=843)
Table 5. 21: Knowledge and source of information of mothers/caregivers about childhood pneumonia in Sidama Regional State, Ethiopia: 2020
Table 5. 22: Knowledge of mothers/caregivers on sign and symptom of pneumonia in Sidama Regional State, Ethiopia: 202059
Table 5. 23: Knowledge of mothers/caregivers on causes of pneumonia in Sidama Regional State, Ethiopia: 2020 59
Table 5. 24: Knowledge of mothers/caregivers on transmission method of pneumonia in Sidama Regional State, Ethiopia: 202060
Table 5. 25: Knowledge of mothers/caregivers on prevention methods of childhood pneumonia in Sidama Regional State, Ethiopia: 2020
Table 5. 26: Attitude of mothers/caregivers towards pneumonia prevention and control in Sidama Regional State, Ethiopia: 202062
Table 5. 27: Mothers'/caregivers' practices towards prevention and control of childhood pneumonia in Sidama Regional State, Ethiopia: 2020, (N=843) 63
Table 5. 28: Socio-demographic and economic factors associated with knowledge of mothers/caregivers towards pneumonia prevention and control in Sidama Regional State: 2020, (N=843)
Table 5. 29: Factors associated with attitude of mothers/caregivers towards pneumonia prevention and control in Sidama Regional State: 2020, (N=843) 65
Table 5. 30: Factors associated with practice of mothers/caregivers towards pneumonia prevention and control in Sidama Regional State: 2020, (N=843) 66
Table 8.1: Summary of major findings of the research in Sidama Regional State, Ethiopia: 2020 2020
Table 8.2: Priority areas of action of awareness creation and prevention strategies on pneumonia prevention and control in Sidama Regional State: 2020 80
Table 8.3: Indicators to evaluate progress in achieving key operational targets for awareness and prevention strategies on childhood pneumonia

List of Figures

Figure 2.1: APP framework for protecting, prevent and treating childhoodr pneumonia (WHO and UNICEF 2009a:4) 27
Figure 3.1: Conceptual framework developed by researcher (adapted and modified from HBM (Turner 2004: 34) and GAPP framework (WHO and UNICEF 2009a: 4)
Figure 4.1: Map of study site, Ethiopia, 2020
Figure 4.2: Schematic presentation of how to select sample population for quantitative part of study in Sidama Regional State: 202035
Figure 5.1: Wealth index of households in Sidama Regional State, Ethiopia: 2020 (N=843)
Figure 5.2: The prevalence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843) 48
Figure 5.3: Knowledge of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020
Figure 5.4: Attitude of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020
Figure 5.5: Practice of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020 (N=843)
Figure 8.1: Diagrammatic presentation of Delphi group data collection and analysis

 Figure 8.2: Continuum of care approach: time of care giving and place and approach of care giving (FMOH 2015:36).
 78

 Figure 8.3: The WHO health system framework/the six building blocks of health system (WHO 2007:3).
 79

 Figure 8.4: Conceptual framework developed for implementation of proposed awareness and prevention strategies on childhood pneumonia in Sidama Regional State: 2020
 80

List of Annexures

ANNEXURE A: ETHICAL CLEARANCE GRANTED FROM UNIVERSITY OF SOUTH AFRICA	88
ANNEXURE B: LETTER OBTAINED FROM SNNPR HEALTH BUREAU TO CONDUCT RESEARCH	90
ANNEXURE C: LETTER OF PERMISSION FROM FORMER SIBAMA ZONE HEALTHDEPARTMENT	91
ANNEXURE D: INFORMATION SHEET AND CONSENT FORM	91
ANNEXURE E: DATA COLLECTION TOOL FOR QUATITATIVE PART OF STUDY (ENLISH VERSION	92
ANNEXURE F: INFORMATION SHEET, CONSENT FORM AND DATA COLLECTION TOOL FOR QUATITATIVE PA (SIDAAMU AFOO) VERSION	ART OF STUDY 99
ANNEXURE G: FGD GUIDE AND CONSENT FORM FOR QUALITATIVE DATA COLLECTION (ENGLISH VERSION	N)108
ANNEXURE H: FGD GUIDE AND CONSENT FORM FOR QUALITATIVE DATA COLLECTION (SIDAAMU AFOO V	VERSION) 109
ANNEXURE I: COMMENT AND FEEDBACK FORM FOR DELPHI GROUP	110
ANNEXURE J: BIBLIOGRAPHY OF DELPHI TECHNIQUE INDIVIDUALS	110
ANNEXURE K: DIRECTION TO ASSESS AND CLASSIFY THE CHILDHOOD PNEUMONIA ADAPTED AND MODIF BASED ON IMNCI CHART BOOKLET	TED BY THE PI 111
ANNEXURE L: LANGUAGE EDITORIAL CERTIFICATE	111
ANNEXURE M: SIMILARITY INDEX REPORT	112

List of Acronyms

AIDS:	Acquired immunodeficiency syndrome		
AOR:	Adjusted odds ratio		
ARI:	Acute respiratory infections		
CHERG: Child health epidemiology reference group			
CI:	Confidence interval		
COR:	Crude odds ratio		
CSA:	Central statistical agency		
DPT:	Diphtheria, tetanus and pertussis		
EDHS:	Ethiopian demography and health survey		
EPI:	Expanded programme on immunization		
FGD:	Focus group discussion		
FIRS:	Forum of international respiratory societies		
FMOH:	Federal Ministry of health		
GAPP:	Global action plan for pneumonia prevention		
GAPPC:	Global action plan for the prevention and control		
GVAP:	Global vaccine action plan		
HAD:	Health development army		

HBM:	Health Belief Model
HEW:	Health extension worker
HHs:	Households
Hib:	Homophiles influenza type b
HIV:	Human immunodeficiency virus
IMNCI:	Integrated management of childhood illness
IVAC:	International vaccine access centre
KAP:	Knowledge, attitude and practice
LMICs:	Lower middle income countries
MCV:	Measles vaccine
MDG:	Millennium development goal
MNCH:	Maternal and child health
MUAC:	Mid upper arm circumference
NEB:	National Electoral Board
NGO:	Non-governmental organizations
OCA:	Organizational capacity assessment
ORS:	Oral rehydration solution
PCV:	Pneumococcal conjugate vaccines
PMTCT:	Prevention of mother to child transmission
PPP:	Public Private Partnership
RR:	Respiration rate
SDG:	Sustainable development goals
SNNPR:	Southern nation nationalities and people's region
SPSS:	Statistical package for the social sciences
SRS:	Simple random sampling
TB:	Tuberculosis
UN:	United Nations
UNICEF	: United Nations Children's Fund
UNISA:	University of South Africa
URI:	Upper respiratory infections
USAID:	United States Agency for International development
VIP:	Ventilated improved latrine
WASH:	Water, sanitation and hygiene
WHO:	World health organization

CHAPTER 1

ORIENTATION TO THE STUDY

1.1 INTRODUCTION

Pneumonia is one of the acute lower respiratory tract infections which can be caused by bacteria or viruses or other micro-organisms those cause inflammation of the lungs, that makes difficult for the oxygen breathe in

Auctores Publishing LLC – Volume 8(1)-255 www.auctoresonline.org ISSN: 2637-8892

and get into bloodstream. Most of the severe or fatal pneumonia is caused by bacteria (Save the Children 2017:6).

The signs and symptoms of childhood pneumonia can be cough with or with out fever, chills and difficulty in breathing. Its severity could be ranged from mild to severe and might be affected by the factors like type of micro-organisms causing the lung infection, age of persons and their health status. The likely vulnerable groups are under five children and 65 or older people who have co-morbidies (WHO 2019a:1-2) Globally pneumonia remained as the leading cause of mortality among children under five years old; killing about 2,216 children in a day. In 2017 (WHO 2019b:1) childhood pneumonia accounted nearly 15 per cent of the 5.4 million under-five deaths worldwide and killing around 808,694 children. Most of its vulnerable age groups were less than 2 years old children.

Pneumonia is a major cause of illness and death in under five children in resource Low Countries. In the last decade there have been different improvements and new interventions, resulting in a significantly reduction in pneumonia incidence and improved outcome (WHO 2019a:1). In 2015 countries like India, Nigeria, Indonesia, Pakistan, and China alonenly contributed more than 54% of all global cases of pneumonia. Similarly 49% of global pneumonia deaths occurred in India, Nigeria, Pakistan, Democratic Republic of Congo and Ethiopia McAllister et al 2019: e47).

Although there have been improvements globally; the challenge posed by pneumonia has persisted in the above mentioned five countries where half of all under-five deaths registered in 2018 occurred in these countries (WHO 2019a:1).

Ethiopia is 6th leading country from the top 15 countries those estimated with the highest number of under five deaths due to pneumonia and diarrhoea, with the 15 deaths/1000 live births per year according to international vaccine access centre (2016:6). Pneumonia is a leading cause of under-five children in Ethiopia. It killed more than 32,000 children under-five in 2018 which mean more than 4 children every hour and shares 17% of all causes of deaths. Pneumonia can be prevented easily by different cost effective interventions like proper nutrition, immunizing the children, exclusive breastfeeding, appropriate complementary feeding and hand washing (Save, UNICEF & every breath count 2019:1-2).

According to the annual report of Health Bureau of southern nation, nationality and peoples regional State as well as Sidama zone health department (2018:25-32) pneumonia registered as the first leading cause of both illinesses and deaths in the study area. In the budget year of 2017/18 it caused 339,152 morbidity cases in the southern region and from these 42,830 cases registered in the Sidama zone.

According to the report of Save the Children (2017:34) the early recognition of pneumonia and its danger signs can help caregivers and or health care workers to prevent deaths of children due to childhood pneumonia. Thus, establishing the proper health information and communication systems to provide evidences on childhood pneumonia to the communities, can contribute to control burden of pneumonia. Mothers/caregivers can easily identify signs and symptoms of pneumonia and seek appropriate care for their children as if they got information and empowered very well.

The current research aimed to develop awareness and prevention strategies through assessing the overall prevalence of pneumonia among under five children and its contributing factors like socio-demographic, environmental, behavioural and nutritional as well as immunization related determinants. Additional the study was also explored the level of knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia.

The study gives an insight into the household determinants of childhood pneumonia and assists to recognize the actions to be prioritised and recommends prevention and awareness creation strategies for the better prevention and control of childhood pneumonia in the study area.

1.2 BACKGROUND TO THE RESEARCH PROBLEM

Global pneumonia occured as one of the most common infectious diseases which affects millions of life every day. It remained as immediate cause of death for young children in the poorest countries. The World Health Organization and the United Nations Children's Fund together developed and promoted the approach called Integrated Management of Childhood Illness (IMCI), for the better patient management and significantly reducation of morbidity and mortality those caused by common childhood diseases, including pneumonia. However deaths of children due to pneumonia are declining slowly than the other major killers of diseases, where the high-burden countries in sub-Saharan Africa registering the slowest progress (Save the Children 2017:5; Miller et al 2014:24).

According to the report of UNICEF (2016a:7), deaths of children due to pneumonia concentrated within the poorest countries. Low and lower-middle income countries account more than 90 percent of pneumonia and diarrhoea deaths together. Furthermore the poorest countries share a disproportionate burden of pneumonia deaths; where more than 30% of all pneumonia and diarrhoea deaths are concentrated. Morbidity and mortality due to pneumonia also disproportionately affects the youngest children whose about 80% of deaths associated with pneumonia.

Pneumonia remained a major cause of illinesses and deaths among under five children and deaths due to pneumonia has been declining more slowly than other diseases of child mortality such as malaria, measles and HIV in 2015. Accordingly, reports from UNICEF (2016b:7) and Save the children (2017: vii) revealed that around 735,000 childhood pneumonia deaths may be registered in 2030 if no significant progress is made.

Childhood morbidity due to pneumonia in developing countries can be associated with behavioural, socio-economic and environmental factors. Having a better knowledge on such underlying risk factors of childhood pneumonia is one of the strategy to protect children against pneumonia (Mihrete et al 2014:1; Diaz et al 2013:1). Improper nutritional status (micronutrient deficiency as well as lack of breastfeeding), HIV infection, exposure to indoor air pollution, overcrowding, premature birth, and poor living conditions have also been reported to influence the occurrence of childhood pneumonia (Dadi et al 2014:1; Fekadu et al 2014:1; Ghimire et al 2012:1). However, only limited data exist about the knowledge, attitude and practice of households towards childhood pneumonia prevention and control at the current study. Thus, there is a need for further research to reveal the influence of the above-mentioned factors on childhood pneumonia.

Improvements those showed global to reach the children who are losing their life because of pneumonia are not fast or fair enough across the globe. Children from the poorest countries are two times more likely to die before their 5th birthday compared to those from developed countries. This is because of undernutrition and lack of quality health services including immunizations, diagnosis and treatment of common childhood illnesses. Due to this, iInterventions that may save thousands of lives each year are not reaching children with the greatest need (Save, UNICEF & every breath count 2019:1).

World Health Organization (2013:5-9) reported that pneumonia is a common problem for everybody and its severity can be increased by different factors like medical conditions, health behaviours and environment. Most of the time infants and children below two years of age are at increased risk of pneumonia, because of lack of well developed immune systems.

In 2009 WHO and UNICEF launched the new approach called Global Action Plan for the Prevention and Control of Pneumonia framework (GAPP) global, with the aim to achieve that every child should be protected from pneumonia through a healthy environment and access to preventive and treatment measures. The GAPP brought a better opportunities to integrate activities as well as capture synergies and efficiencies (WHO/UNICEF 2013:5-7&38). It proposed the different interventions for preventing and controlling of pneumonia among under five children; through protecting them by establishing and promoting good health practices, preventing them from becoming ill by ensuring healthy environments and universal coverage of immunization, HIV prevention and treating children who are ill from pneumonia with appropriate treatment.

The GAPP framework also aimed to strengthen the countries to achieve impacts through analysing their local data, intervening on the results and continuously monitor and evaluate their progress towards clear and achievable goals by 2025. These goals helped to reduce child mortality due to pneumonia to less than 3/1000 live births and reduced the incidence of severe pneumonia by 75% compared with 2010 levels. Likewise it also helped to reduce the global burden of stunted children by 40% compared with 2010 levels. Additional targets like 90% full-dose coverage of relevant vaccines, 90% access to pneumonia case management, 50% coverage of exclusive breastfeeding and virtual elimination of paediatric HIV infections were also included in GAPP framework. Eventhough, there have been clear actions and results achieved, the progress made so far is not enough. Thus governments and the global community need to address pneumonia comprehensively to improve child survival.

In Ethiopia 9 deaths per 1000 live births were reported among under-five children due to pneumonia in 2018 and it was very far away from the goal of GAPP 2025 (3 deaths per 1000 live births planned). So, the concerted actions and efforts needed to save lives of children through improving policies, creating innovations, search and scale up evidence-based interventions. It is a must to combat childhood pneumonia and to leave no child behind (Save the children, UNICEF & Every Breath Counts 2019:1-2).

Current study aimed to develop awareness and prevention strategies on pneumonia among under five children through determining the prevalence of pneumonia and associated socio-demographic, environmental, behavioural, child related, nutrition related and immunization related factors and assessing the level of knowledge, attitude and practice of mothers/care givers towards the pneumonia prevention and control among under five years old children.

1.3 STATEMENT OF PROBLEM

Despite the different actions taken for the reduction of childhood pneumonia burdens, the challenge has not significantly reduced as compared to other diseases like malaria, measles, HIV/AIDS and diarrhoea. According to the estimation done by UNICEF, Globally including Ethiopia, pneumonia mortality fell by only 46% in 2000 and 2015, which was low improvement compared with 85% decline for measles and 61%, 58% and 57% decline for AIDS, malaria and diarrhoea respectively (Save the Children 2017:14).

Within sub-Saharan countries, childhood pneumonia mortality have not declined rather increased to the contrary. According to the IHME report most countries, especially Tanzania, Ethiopia and Uganda, failed to reduce childhood pneumonia mortality as expected, even though they had achieved large reductions of deaths from measles, diarrhoea, malaria and HIV/AIDS, in 1990 to 2013 (International Vaccine Access Center 2018:3-5).

In Ethiopia childhood pneumonia is among the most prevalent diseases (malaria, diarrhoea, measles and problems of the new born). Pneumonia, diarrhoea, malaria and measles accounted for more than 18% of the total deaths in under-five years of age children in 2014. Despite all the impressive efforts exerted by countries and the declining of two-thirds of childhood deaths and achievement of MDG-4; pneumonia accounts the first share among all deaths (FMOH 2015:2). Likewise, it is the most prevalent and accounts 20% share of other acute respiratory infections among under five years' children in Oromia region, Ethiopia (Dadi et al 2014:1).

With the collaborative effort of government and stakeholders a significant progress seen in decreasing child death in Ethiopia. According to the estimated report of UN inter Agency Group's 2013, in Ethiopia underfive, infant (below 1 year) and neonatal (up to 28 days) mortality rates were reported as 68, 44 and 28 per 1000 live births respectively. Similarly WHO/CHERG estimated that Ethiopia that acute respiratory infection or childhood pneumonia accounts 18% of under-five mortality, birth asphyxia 14%, premature births 11%, diarrhoea and sepsis each 9%, meningitis and injury alonely 6%, measles 3% and others including measles account 24% in 2014 (CSA/EDHS 2016:22, FMOH 2015:2 & UN 2013:14).

Despite those all efforts and impressive results achieved; Ethiopia has been ranked the 10th top pneumonia high burdened countries and included under "every breath counts coalition" approach; which was launched to prevent child pneumonia deaths by 2030 through strengthen the partnership to support national governments and gave the especial focus for countries with the largest populations of under five children (Every Breath Counts 2017:3). According to the study conducted by Lema et al (2018:10) childhood pneumonia continue to be a public health problem in Ethiopia.

Provision of frequent and relevant data on pneumonia and its associated factors is assumed to be vital for program planning and availing essential child health care services. It is known that childhood pneumonia is preventable and needs properly identification of contributing factors and take preventive measures accordingly. The existing study findings confirmed some of nutrition related determinants for pneumonia, such as stunting, underweight, sub-optimum breastfeeding, under-nutrition and zinc deficiency within its geographic codistribution of pneumonia (Bhutta et al 2013:1). But there is gap obtaining overall burden of pneumonia and associated factors at the current study area and gap of properly aligning study findings for action.

According to the annual reports of Sidama Zone Health department, childhood pneumonia was the first leading cause of morbidity for past decade and a facility base study conducted in Wondo Genet district, Sidama zone, southern Ethiopia showed that the prevalence of pneumonia among under-five children was 33.5% (Abuka 2017:1). However there was no well investigated study findings that can show the burden of childhood pneumonia and its community base contributing factors in the current study area with considering more than one districts. Similary there was no evidence about knowledge, attitudes and practice of mothers/caregivers about childhood pneumonia control. Not only that there have been no developed evidence base awareness and prevention strategies in place.

Success of public health interventions to reduce the burden of childhood pneumonia as well as improve care seeking behaviour and management of pneumonia depends on various factors. These factors may include identification of disease burden and associated factors with childhood pneumonia. Although they pose major economic and health burdens to the communities. The current study, therefore, aimed to develop awareness and prevention strategies on pneumonia among under five children at the current study area, through assessing pneumonia burden and associated factors. Furthermore, assessing the level of knowledge, attitude and practice of mothers/caregivers towards childhood prevention and control.

1.4 AIM/PURPOSE OF RESEARCH

The purpose of the current study was to develop awareness and prevention strategies on pneumonia among under-five years' children in Sidama Regional State, Ethiopia in line with the strategy to accelerate the goal of ending preventable pneumonia disease by 2025

1.5 OBJECTIVES

Phase 1

- To determine the prevalence of pneumonia among under-five children in Sidama Regional State, Ethiopia, 2020.
- To identify the socio-demographic and economic factors of pneumonia among children under-five years in Sidama Regional State, Ethiopia, 2020
- To assess the behavioural and health related risk factors of pneumonia among children under-five years in Sidama Regional State, Ethiopia, 2020
- To assess the environmental risk factors of pneumonia among children under-five years in Sidama Regional State, Ethiopia, 2020
- To assess the nutritional and expanded immunization related pneumonia among children under-five years in Sidama Regional State, Ethiopia, 2020
- To assess knowledge, attitude and practice of mothers/caregivers with under-five children towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia, 2020
- To assess factors associated with knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia, 2020

Phase 2

Findings from phase 1 were addressed by the following objectives for phase 2:

- Explore the Delphi groups' recommendations for the development of awareness and prevention strategies for pneumonia among under-five year's children in Sidama Regional State, Ethiopia, 2020
- Develop awareness and prevention strategies for pneumonia among under-five years old children in Sidama Regional State, Ethiopia based on study findings, 2020

1.6 SIGNIFICANCE OF STUDY

Current study has provided a breakthrough insight into prevalence and different associated factors of childhood pneumonia and KAP of mothers or care givers towards childhood pneumonia prevention and control. It explored the evidence of relationship between socio-economic, environmental, behavioural and nutritional related factors and the occurrence of childhood pneumonia. The result of this study has been used to develop awareness and prevention strategies for childhood pneumonia after properly analysis and consultation of concerned stakeholders. The awareness and prevention strategies will help to reduce the childhood pneumonia burden in Sidama Regional State, Ethiopia and The awareness and prevention strategies on pneumonia among under five children will help as a baseline reference that can be applied and implimented in different areas out of study area in Ethiopia as well as outside of the country. The study contribution will also create a pool for scientific knowledge and served as the basis for further researchs.

1.7 SCOPE AND LIMITATIONS

Current study covered the Sidama regional state, Ethiopia which has 30 rural districts, 6 town administrations and one city admiration with 8 sub cities. It has a population of more than 4.5 million and around 792,435 under five children. The findings of research will be operated within study areas and it can be also applied out of study area as the problem is important public health concern. The implementation of study findings will include households especially mothers/caregivers with under five children.

Limitation of the current study might be related with the study design which was cross sectional one and that can not pin down cause-and-effect relationship. The calibration error might be also another limitation while taking anthropometric measurements to assess nutritional status of children.

1.8 THEORIES/META-THEORETICAL GROUDING/THEORETICAL FRAMEWORK

In this sub-topic the Meta-theoretical assumptions, theoretical framework, Health Belief Model (HBM) and GAPP framework for pneumonia prevention and control those have been the foundations of the current study were discussed. The theory that has been used and thought as the most appropriate for this research was the HBM complemented with GAPPs Framework for pneumonia control. The HBM was initially developed and utilised since 1950s by social psychologists who had worked in Public health services, United States of America to explain why many people did not participate in massive public health programs such as TB or cervical cancer screening (Nancy & Anz 1984:2); whereas GAPPs framework was developed in 2009 (WHO & UNICEF 2009a:4). As applied to this research, HBM holds that the independent variables (socio-demographic and economic, knowledge/indigenous, beliefs, the environment) could influence or explain the dependent variables or the prevalence of childhood pneumonia and the effect of care givers KAP towards childhood pneumonia prevention and control. A person with good awareness and positive attitudes towards childhood pneumonia prevention strategies, is expected to likely to practice preventive measures and can thus reduce the risk of getting pneumonia and of treating patients with pneumonia effectively. GAPPs framework used as guiding principle while developing childhood pneumonia prevention and control strategies at large.

1.8.1 Meta-theoretical assumptions

According to a review conducted by Lor (2017:2-3), Meta theory is assumptions that can underpin investigators' decisions on methodologies for specific projects, or it is the assumption of which they may or may not be aware. This theory is also can be defined as presupposition which provides general perspectives based on assumptions about the nature of reality and human beings (ontology), the nature of knowing (epistemology), the purposes of theory and research (teleology); values and ethics (axiology); and the nature of power (ideology), (Musa 2013:42-3).

For the current study meta-theoretical assumptions used as assisting scholarly of the inquisiton in the delivery of a way of thinking and justfying the philosophical approach of study which would shape the action of the researcher in the selection of a study design and methods.

1.8.2 Theoretical framework

According to Glanz and Bishop (2010:400), interventions to improve health behaviour can be best designed with an understanding of relevant theories of behaviour change and the ability to use them skilfully. The science and art of using health behaviour theories glisten the mixed of approaches, methods and strategies from social and health sciences. These theories are very instrumental in understanding the most common contributing factors in community and directing our attention to the critical issues that can create a difference in the health of the community.

Most of the time pneumonia can affect those poor people and mainly it can be affected by factors like socio-demographic and economic, behavioural, environmental, nutritional and health services related (UNICEF 2016c:2).

The HBM and GAPP have been selected to guide the researcher in establishing the relationship between the explanatory and the outcome variables. According to Glanz and Bishop (2010:400), interventions those developed with a clear theoretical foundations are more effective than those lacking theoretical base and some strategies that combine multiple theories and concepts have larger effect. Thus, researcher used both complementary theories as base for present study to develop awareness and prevention strategies for childhood pneumonia by assessing all contributing factors of childhood pneumonia or study variables (socio-demographic and economic, environmental, behavioural and health related, child related nutritional and immunization) and KAP of respondents with its associated factors towards prevention and control of childhood pneumonia. The common pneumonia contributing factors indicated in the theoretical grounding were the focus areas, that our study has been addressed.

1.9 DEFINITION OF KEY CONCEPTS

Awareness

According to Sharma and Romas (2012:29), awareness defined as being conscious about an action, idea, person or situation or it is knowledge gained through means of information or one's own perceptions. For the current study awareness means level of community understanding or being familiar about childhood pneumonia prevention and control strategies. Identifying the causes of pneumonia, mode of transmission, prevention methods and management.

Knowledge

Knowledge is the capacity to acquire, retain and use of information (Badran 1995:1). In the current study, knowledge is considered as a mixture of comprehension, experience, discernment and skill of mothers/caregivers on sign and symptom of pneumonia, causes, transmission way and prevention of pneumonia.

Attitude

From different literatures attitude is leaning or tendency to react in some way to certain situations to look and interprete situations from the view of predispositions. (Badran 1995:1 & Sharma 2012:31). In the current study, attitude refers to as relatively common feeling of

mothers/caregivers towards pneumonia preventionand control mechanisms

Practice

Practice can be taken as practicing and taking actions based on certain standards or rules (Badran 1995:1-2). In the current study, practice was considered as the act of mothers/caregivers towards childhood pneumonia prevention and control mechanisms either to improve the health of children from the infection or to protect them.

Pneumonia

Childhood pneumonia is a common lung infection where the lungs' air sacs become inflamed. These sacs may also be filled with fluid, pus and cellular debris. Childhood pneumonia can be caused by viruses, fungi, or bacteria. It can be fast or difficult to breath and cough with or without fever (Sullivan 2017:1-2). For the current study, pneumonia was assessed and diagnosed based on respiration rate of individual children. The exact cuase of the disease did not identified.

Prevention

Prevention measures does not only mean to prevent disease occurrence, but it is also to control disease progress and further disability too. It an action which usually gushed from the health sector, to convise the individuals and communities as whole on certain identified contributing factors of disease and way preneting (WHO 1998:4; Jenkins 2003:6). For the current study prevention means the restricting of the children from being diseased with pneumonia or the efforts of community to intervene and address the pneumonia causing factors to minimize its prevalence among children aged under five and further control of its existences.

Strategy Development

It is the process for a policy to realize its intended impact, which should be operationalized into doable steps and way of considering different important resources like financial as well as human. It is way of collaborative working to improve the health services and quality life of the communities by moving them from a focus on illiness and disease to prevention and wellness (Organizational Capacity Assessment 2014:1-3). For the current study strategy development was used as process, methods or ways of preparing the guiding tool for the childhood pneumonia prevention and control by using the availed literatures, the study findings and Delphi group comments and suggestions.

Delphi method or technique

Delphi method is the arranged sequence of communication on certain issues among selected experienced people or experts who are more knowledgeable about the specific issues and who may from diverse geographic and discipline background. Delphi method of discussion helps to understand the issues from many panellists living in different areas and from diverse disciplines participated in a normative way (Geist 2010:148). For the current study Delphi method indicated the two round mail communications with 10 different experts via email in the area of childhood pneumonia prevention and control based on draft developed awareness and prevention strategies. They gave the comments and suggestions regarding the developed awareness and preventions strategies on pneumonia among children aged under five in Sidama Regional State.

1.10 RESEARCH DESIGN AND METHODS

Research methods are the tools and techniques for doing research; whereas a research design is a procedural plan that is adopted by the

researcher to answer questions validly, objectively, accurately and economically (Kumar 2011:345).

1.10.1 Research Design

Current study used a *community based* cross-sectional, explorative and descriptive study design for quantitative and focus group discussion (FGD) for qualitative study; to assess the prevalence of childhood pneumonia and associated factors and also to determine the level of KAP of mothers/caregivers towards childhood pneumonia prevention and control within its contributing factors.

1.10.2 Research setting

This research was conducted in Sidama Regional State, Ethiopia where there were 30 rural woreda, 6 towns and one city administration with 8 sub-cities. There are 547 smallest administrative units (rural kebele) and 135 urban kebele. The present study addresses 12 rural kebele from 5 woreda and one from town administration.

1.10.3 Research population

The current study considered all household mothers and caregivers with under-five children who live in Sidama Regional State as source population and household mothers or caregivers with under-five children living in six districts namely Dale, Aleta Chuko, Bensa, Bona Zuria, Loka Abaya districts and Aleta Wondo towns as target population. Furthermore the study samples were considered as randomly selected mothers/caregivers from households who with under-five children and included under the study

1.10.4 Data collection procedure

Data were collected by using the quantitative and qualitative research techniques after training the data collectors and supervisors. Structured interview and observation for the quantitative part and FGD with guideline for the qualitative part were used. The Delphi comments and suggestions were also collected from the experts for the development of awareness and prevention strategies.

1.10.5 Data analysis

The researcher used Epi Data version 4.2.2 for data entry and exported to SPSS version 25 for further cleaning and analysis. Descriptive statistics, bivariate logistic regression and multivariate analysis were conducted. Qualitative part of data were analysed manual.

1.10.6 Ethical considerations

The ethics clearance and approval to conduct the research was obtained from the research ethics committee, Department of Health Studies, University of South Africa. Institutional permissions were obtained from the SNNP Regional Health Bureau and former Sidama zone health department, Ethiopia. The individual consent were also obtained from each respondents.

1.11 STRUCTURE OF THE DISSERTATION

The thesis divided as or has eight chapters and discussed as follows:

CHAPTER 1 ORIENTATION OF THE STUDY

In this chapter the introduction and background of the study were discussed. Similarly the statement of the problem, significance of the study, theoretical framework, the study aim or purpose, objectives and scope of the research were included. Further the chapter addresses the issues about key conceptual definitions, methods of research and structure of dissertation.

CHAPTER 2 LITERATURE REVIEW

Chapter 2 describes about the literatures reviewed on childhood pneumonia burden, factors associated with, KAP of mother/caregivers towars childhood pneumonia prevention and control, the availed pneumonia prevention and control strategies.

CHAPTER 3 THEORETICAL FOUNDATIONS OF THE STUDY

The theoretical foundation of the study, conceptual framework, Health Belief Model and Global Action of Pneumonia Prevention and control framework were discussed in chapter.

CHAPTER 4 RESEARCH DESIGN AND METHOD

Chapter 4 discusses about methods and materials of the current study, (research design study settin, population, sampling method, data collection tool and procedures, issue of ethical considerations, data quality management and analysis, validity and reliability). **CHAPTER 5 RESULT PRESENTATION**

Chapter 5 presents the results of both qualitative and quantitative study.

CHAPTER 6 DISCUSSION OF RESEARCH FINDINGS

The discussion part of the current study presented in chapter 6, where research findings has been tried to integrate with the availed research findings.

CHAPTER 7 CONCLUSSION, RECOMMENDATION AND LIMITATIONS

In chapter 7 the issues about the conclusion and recommendations were presented. Similarly limitation of the study, conclussions and contributions of the study were presented accordingly.

CHAPTER 8 PROPOSED AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER FIVE CHILDREN IN SIDAMA REGIONAL STATE

Chapter 8 presents the proposed awareness and prevention strategies on pneumonia among under-five year's children in Sidama Regional State, Ethiopia. The chapter discusses about the the method of strategy development and detail contents of proposed strategy including monitoring evaluation way.

1.12 CONCLUSSION

This chapter presents the introduction and background to the research problem, the source of the research problem, significance of the study, the foundation, objectives and scope of the research.

The chapter also discusses about the theoretical foundation of the research, conceptual framework, health belief model and GAPP framework. Furthermore, the chapter also addresses the key conceptual definitions, arrangement of methods used and arrangement of chapters.

The next chapter will discuss about the literature review of the study.

Chapter 2

LITERATURE REVIEW

2.1 INTRODUCTION

The chapter presents the review of literature accessed from different sources like books, journals, published articles, governmental as well as non-governmental organizations' reports, EDHS, WHO factsheets and documents, UNICEF reports and global as well as national plans setted regarding childhood pneumonia prevention and control. Researcher reviewed or could access those materials via enternet, google scholar and pubmed, searched with using key words like, childhood pneumonia, prevalence of under five pneumonia, factors associated with childhood pneumonia, childhood pneumonia prevention and control strategies, KAP of mothers/caregivers towards under five pneumonia and other feasible sentences as well as phrases were used. The literature review starts with the general overview of childhood pneumonia, burden and its associated factors like socio-demographic, environmental, and nutritional. It also covers KAP of mothers towards childhood pneumonia prevention and control with its associated factors of and lastly the availed childhood pneumonia prevention and control strategies were also discussed under this chapter.

2.2 OVERVIEW OF PNEUMONIA

Pneumonia is one of the infectious disease of the children that can cause infection of the lungs and it can be caused by bacteria, viruses or fungi. The infections of pneumonia may causes the alveoli, to become inflamed and fill up with fluid or pus that can make it difficult for the oxygen we breathe in to and get into our bloodstream. The most common symptoms of pneumonia are cough, fever, chills, and trouble breathing. The severity of pneumonia can be categorised into mild, moderate and severe. The streptococcus pneumonias is one of the most common cause of bacterial pneumonia in children, followed by *Haemophilus influenzae type b;* whereas respiratory *syncytial virus* is the among most common viral cause of pneumonia; and *pneumocystis jiroveci* is responsible for at least one quarter of all pneumonia deaths in infants infected with HIV (WHO 2014:1-4).

Factors like germ causing the lung infection, age of idividuals and their overall health status can determine the overall seriousiness of pneumonia case. Pneumonia can affect the most vulnerable groups of communities those are infants and young children, adults 65 years or older, and people who have co-morbidies (WHO 2019a:1-2).

According to study conducted by Maranu and Zar (2019:4), the pneumonia caused aetiologies have been increasingly ascribed to multiple organisms as detected by molecular testing and they can be categorized as bacteria, virus and fungi. Regarding their classifications, bacteria types are *staphylococcus aureus*, *haemophilus influenza*, *streptococcus pneumonia*, *mycobacterium tuberculosis*, *bordetella pertussis*, *klebsiella* pneumonia. The viral categories are respiratory syncytial virus, rhinovirus influenza, human Meta, adenovirus, para influenza virus, rhinovirus measles virus herpes viruses and those fungal is *Pneumocystis jirovecii*.

The pneumonia causative infectious agents can be transimitted through airborne droplets such as coughs or sneezes inhaled into the lungs, direct oral contact with someone who has pneumonia through kissing and during or shortly after birth, the babies are also at a higher risk of developing pneumonia from coming into contact with infectious agents through birth canal, or from contaminated instruments if used for delivery (FMOH).

Approaches like integrated management of neonatal and childhood illnesses (IMNCI) and integrated community case management (ICCM) can be used to assess and diagnose the sick children with cough or difficult breathing those with symptoms of pneumonia. It is possible to classify pneumonia as severe or non severe based on presence or absence of danger signs like stridor, chest in-drawing and or inability to feed or drink. Health workers can use standard guidance for monitoring the child's breathing rate with a timer and observing the child for chest indrawing. Thus childhood pneumonia can be identified by a breathing rate of 50 or more breaths per minute, for children between the ages of 2-12 months and 40 breaths per minute for those children from 12 months to 5 years. If pneumonia diagnosed the child should be given antibiotics (Amoxicillin) and sent to home, or referred to a health facility for further care, depending on the severity of the symptoms (WHO 2014:2-3; FIRS 2017:17).

According to Ashraf (2014:63); pneumonia can be classified as the following:-

Pneumonia: For child (2 months to 5 years), can be defined as a cough or difficult breathing with or without fever and having only fast breathing, but no other signs and symptoms of severe disease like chest in-drawing, stridor and or unable to eat or drink.

Severe pneumonia/very severe disease: in child can be cough or difficult breathing or fast breathing and with one of the signs and symptoms of severe diseases (stridor, chest in-drawing, unable to feed/drink, lethargy and vomiting).

World Health Organization (2019a:3), proposed that under five pneumonia can be prevented through ensuring proper immunizations, adequate nutrition with exclusive breast feeding, reducing indoor air pollution, accessing good housing conditions, encouraging good hygiene practices and availing cotrimoxazole prophylaxis for children infected with HIV. Likewise engaging the community and encouraging their participation may also help to reduce morbidty as well as mortality of children due to pneumonia.

Early diagnosis and management of childhood pneumonia is one of the critical area which can help the reduction of morbidity and mortality of child. It can be achieved through increased community involvement and awareness. For this the improved diagnostic tests like better sampling procedures and rapid detection of infectious agents are needed which enables the targeted therapy. In contrary unwise use of antibiotics may agravative the situation of emergence and bacterial resistance. However the proper use of antibiotics may decrease the challenge related with antimicrobial drug resistance (FIRS 2017:18).

2.3 BURDEN OF CHILDHOOD PNEUMONIA

Of the 2.56 million pneumonia deaths global in 2017, majority of victims were under five children and it is the leading cause of mortality. Although too many children still die today even if there has been a three (03) fold reduction in child deaths due to pneumonia globally since 1990 (Dadonaite and Roser 2019:1).

Pneumonia remained as a leading cause of child illiness and death in low and middle income countries (Marangu and Zar 2019:3). The findings from the study conducted in Philippines showed that childhood pneumonia remained as the public health important issues and the results obtained as incidence rates of pneumonia episodes were 105, the episodes of severe pneumonia 61 and deaths associated with pneumonia were 0.9 per 1,000 person-years (Kosai et al 2015:1).

According to the research findings which was conducted in India, (AwasthiI et al 2019:1) about 24.6% children suffered from community acquired pneumonia within past 12 months preceding the survey and from these 4% of children were hospitalized. Another study conducted in two slums of India revealed that the prevalence of childhood pneumonia was 16.3% (Nirmoliaa et al 2017:1).

According to the missing piece world pneumonia day report (2018:23), almost 2.6 million mortality due to pneumonia were concentrated in sub-Saharan Africa, south Asia, and south East Asia. Sub-Saharan Africa alone share the largest number of pneumonia deaths, followed by South Asia and South East Asia. In sub-Saharan Africa, more than 60% of pneumonia deaths are concentrated among under five children, while in the developed region, 86% pneumonia deaths are concentrated among the older peoples whose age over 70 years. In contrast, South Asia and South East Asia carry a double burden due to pneumonia deaths, with significant deaths among both children and the elders. Furthermore in South Asia under five deaths account 40% and adults over 70 years account 34% of deaths due to pneumonia while in South East Asia 14% of pneumonia deaths were among children and 60% were among the older age group.

The result of systematic analysis conducted globally, regionally, and nationally to identify the causes of under-5 mortality between 2000–2015 showed that pneumonia was the leading cause of deaths among under five children in sub-Saharan Africa (Liu 2016:3027). Similarly the study conducted in Ghana (Osei et al 2018:1) showed that prevalence of underfive pneumonia was found to be 18.4% with a mortality rate of 12.7%.

Ethiopia is among high childhood pneumonia burdened countries. The report of Ethiopian Demography and Health Survey (2016:26) showed that the prevalence of acute respiratory infections was 7% with the significant differences of results across the regions. In addition a study result carriedout in Munesa District, Ethiopia revealed that the magnitude of community acquired pneumonia was 17.7% (Lema et al 2019:1). Another research employed in Northwest, Ethiopia revealed that prevalence of childhood pneumonia was 16.1% (Fekadu and Mamo 2014:1). Similarly, a study conducted in Southwest, Ethiopia showed that the prevalence of childhood pneumonia was 28.1% (Lema et al 2018:1).

Finally, previous findings by Yoseph and Fantahun (2019:2) indicated that the prevalence of Pneumonia was found to be 20.68% among under five children, which is high as compared to study findings from Wondo Genet district, Ethiopia which recorded a prevalence of 33.5% (Abuka 2017:10).

2.4 LITERATURE REVIEW FINDINGS ON FACTORS ASSOCIATED WITH CHILDHOOD PNEUMONIA

It is known that global morbidity of child as well as mortality due to pneumonia can be contributed by socio-demographic and economic factors like age of child, educational status and occupation of mothers/caregivers, economic status of families, residence and number of family in the household and environmental factors such as housing conditions, availability of toilet and source of water, poor management and disposal of the child's stool, (Dadonaite and Roser 2019:6-9; Amugsi et al 2015:1).

Evidences from the past studies revealed that different factors like child as well as maternal related, environmental, socio-economic status of households, co-morbid diseases like HIV/AIDS and malaria, lack of breastfeeding and poor nutritional status of the child are important risk factors for childhood pneumonia especially common in low resource countries (Dadi et al 2014:1; Banda et al 2016:153).

2.4.1 Socio-demographic and economic factors contributing pneumonia among under five children

2.4.1.1 Sex of the child as contributing factors of childhood pneumonia

Sex of child is one of the contributing factor for several illinesses. However, the contribution of sex for the occurrence of pneumonia remained unclear and no agreement has been reached in the literature (Ramachandran et al 2012:889, Muenchhoff & Goulder 2014:120). According to the literature review study conducted majority of the respiratory tract infections are more common and severe in males than females and also leading to higher deaths too. Commonly obtained that lower respiratory tract infections occurred in males. This was guessed as the greater resistance found in females due to their enhanced Th1 immune response (Muenchhoff and Goulder 2014:124-5; Falagas 2007:1845). Similar result was obtained from the case control study conducted in Urban part of Bangladesh, which revealed that being male was risk factor of childhood pneumonia (Ram 2014:968).

According to facility based study conducted in Sudan childhood pneumonia was more prevalent in male than female children, which revealed a relationship between gender and pneumonia (Gritly et al 2018:60). Another birth cohort research employed in South Africa reported that boys had a greater incidence of pneumonia than girls (Roux et al 2015:e95). The similar results were obtained from the studies done in Germany and Ethiopia which showed that being male increases the risk of having pneumonia (Lema 2019:1; Kolditz 2016:32). On the contrary, findings obtained from the case control study carriedout in Brazil revealed that being a male gender was found as protective factor for the occurrence of childhood pneumonia (Lima et al 2016:1).

2.4.1.2 Household economic status as factor for the occurrence of childhood pneumonia

Worldwide, poor economic status and living conditions can be taken as risk factors for illnesses and associated with poorly utilization of health care services.

Previous studies revealed that children who from their households' with poor economic status are more likely affected by pneumonia infections than children living with better economic status (Amugsi et al 2015:1; Hemagiri 2014:47). Over all, a better income may help to have better housing conditions, presence of clean fuel for cooking, better food, media access and health seeking behaviour. But the poor communities are likely to live in poor sanitation and hygienic practices, which might increase their exposure to infectious agents (Cardoso et al 2004:1). It is also stated that communities with poor economic status may have more family size and live in crowded houses where the housing conditions are suitable for the transmission of infectious agents (Amugsi et al 2015:1).

Different previous studies revealed that low economic status can contribute childhood pneumonia. For instance a studies conducted in India reported that children those from low socio economic status are at higher risk of developing childhood pneumonia (Gupta and Bhadrala 2019:949; Nirmolia et al 2017:1). Similar result was also obtained from the case control study conducted in Urban Dhaka, Bangladesh, which revealed that poor economic status was the predictors of pneumonia among under-five children (Ram 2014:968). In addition hospital based case control study conducted in Bellary and another descriptive study from Sudan revealed that poor economic status of the family was predictor of childhood pneumonia (Gritly et al 2018:60; Hemagiri et al 2014:47).

2.4.1.3 Child's age as pneumonia contributing factors

Pneumonia can affect the children regardless of all age. However, the evidences showed that younger age groups of children are at more risk of developing childhood pneumonia than others. For instance, a population based study conducted in Germany and a hospital based study conducted in Sudan revealed that less than one year age children were more affected with pneumonia than children of older age groups (Kolditz et al 2016:32, Gritly et al 2018:60). Similarly facility based research employed in Bellary reported that the younger age was found as contributing factor for childhood pneumonia (Hemagiri et al 2014:47).

In addition, a study done in Oromia zone of Amhara region revealed that children at age range 2-11 months were 85% more likely to suffer from pneumonia as compared to older age. Another facility based research employed in Sidama zone alluded that children aged between 2-12 months were 4 times more likely to develop pneumonia than older age group (Abuka 2017:19; Dadi et al 2014:1).

2.4.1.4 Parental education as factors of childhood pneumonia

Parental education is another important risk factors of children's health. The level of maternal education is inversely related to the occurrence of childhood illinesses as well as deaths, mothers with better educational status are more capable about taking care of their children than their contraries which require all necessary recognition and implementation of simple home care and also early presentation to health-care facilities (Gothankar et al 2018:1).

Education helps to improve the socio-economic lives of the families; hence helping in avoidance of some of the risk factors of acute respiratory infections including pneumonia (Ujunwa and Ezeonu 2014:95). However lack of education and poor awareness of the mothers may have negative impact the reduction both morbidity and mortality of children due to pneumonia which can be also the big challenge for public health interventions. A research carriedout in India revealed that gap of maternal education contribute the prevalence of pneumonia (Nirmolia et al 2017:1).

Well educated mothers relatively can understand the any risk factors of child health and may try to prevent them from the exposures while the reverse is true with uneducated mother or mothers with low level of education. An institution based study conducted in India as well as in Vietnam showed that low literacy status of mothers was the significant contributor for the occurrence of acute severe pneumonia in under five children (Hoang et al 2019:274; Gupta and Bhadrala 2019:949).

The study findings from different countries or areas showed that children with low educational level of mothers have high probability of developing childhood pneumonia. An Institution based case control study conducted in India and another hospital based descriptive study, from Sudan showed that low literacy level of mothers had significant association with childhood pneumonia (Gritly et al 2018:60 Hemagiri et al 2014:47).

The same is true with father's educational level as that the less educated mother's positively contributes for the occurrence of childhood pneumonia; where the households with high level educated fathers, there is protective associations. According to the result of study done in Ethiopia showed that the prevalence of pneumonia was significantly contributed with the father's lower level of education grades (1 - 4) and (5 - 8) as compared to higher educational level (Dadi 2014:1).

2.4.1.5 Family size as factor of childhood pneumonia

A household's family size is another important risk factors for childhood pneumonia. Three different studies conducted at different areas of India with different study designs and periods revealed that overcrowding, due to large family size, was found as a contributing factor for the occurrence of childhood pneumonia (Gupta and Bhadrala 2019:949, Gothankar et al 2018:1 & Hemagiri et al 2014:47.

A hospital based study which was done in Southeast Nigeria, reported that having large family size obtained as contributing factor for the occurrence of childhood pneumonia than those from small family size (Ujunwa and Ezeonu 2014:95).

2.4.2 Environmental factors associated with childhood pneumonia

Environmental factors are major determinants for different human diseases. According to the literature review study conducted on childhood pneumonia in developing countries; environmental factors like biomass combustion, exposures to wood combustion, tobacco smoking, day care attendance for children, any refugee environment and school resumption, were found as associated factors with pneumonia.

Positive upper respiratory infections, contact history of households and rainy/winter season were also found as contributing factors of pneumonia (Johnson and Abdulkarim 2013:5).

2.4.2.1 Solid fuels and indoor air pollution as contributing factors of childhood pneumonia

According to the WHO fact sheet (2018a:1), about three billion people has been using solid fuels like; wood, charcoal, crop wastes, coal and dung. Some anothers have been using kerosene in open fires and inefficient stoves for cooking and most of these people are poor and from developing countries. These cooking practices are inefficient use of fuels and technologies that can produce household air pollution which could be the suitable environment for childhood pneumonia transmission.

According to WHO (2018b:2) report of air pollutions and child health, the use of polluting fuels and technologies are common problem for the developing pollution which then contributes respiratorty diseases including pneumonia among under five children. Almost all low and middle income countries are highly burdened about air pollutions. For instanse 83% of the population has been affected in the African region, 59% in the South-East Asia region and 42% in the Western Pacific Region. The evidence obtained from systematic review study carriedout in LMIC countries revealed that solid fuel as a proxy for PM2.5 reported as major contributing factors for childhood pneumonia (Adaji et al 2019:3208).

The findings from four preceding studies conducted in different parts of India using different study designs and time revealed that using biomass fuel, cooking fuel, indoor air pollution and poor indoor household air pollution were strongly associated with the occurrence of childhood pneumonia (Gupta and Bhadrala 2019:949; Nirmolia et al 2018:1; Yeimo et al 2018:307 & Hemagiri et al 2014:47). The same findings were also obtained from a reviewed study done in India, which showed indoor air pollution was found as predictors of childhood pneumonia and recommended as proper attention should be given to minimize the use of such contributing factors (Yadav and Awasthi 2016:83).

A preceding study conducted in South Sudan, around 99% households used solid fuels for cooking and this put the children at risk of pneumonia related mortality, due to household air pollution. To minimize such risks the use of clean or improved stoves for cooking was highly advised in South Sudan (Abd-Elfaraga & Langoyab 2016:36).

As Ethiopian demography and health survey report (2016:11) indicated 95% of households in Ethiopia used the same type of solid fuel for cooking with virtually all of these households using wood. According to cross-sectional studies conducted at public hospitals in Southwest and Northwest Ethiopia, the use of wood and charcoal were the main fuel

source for cooking and as the major contributors of childhood pneumonia (Lema et al 2018:25 & Fekadu et al 2014:150).

2.4.2.2 Smoking as associated factors for childhood pneumonia

It is obtained that air pollution and passive smoking can elevate the risk of pneumonia occurrence. Studies revealed that the presence of indoor air pollution can double the probability of a child developing pneumonia and decreases its recovery. In 2017, indoor air pollution estimated to cause 29% childhood pneumonia deaths and outdoor air pollution shared another 18% of all deaths (Dadonaite & Roser 2019:7). Detailed findings reported that passive smoking children are more likely to develop pneumonia than non-exposing children. Furthermore, Dadonaite & Roser (2019:7) reported that exposure to second hand smoke led to 61,000 deaths due to lower respiratory disease among childhoods in 2017. Also, findings of a study conducted by Ujunwa & Ezeonu (2014:95) in Enugu, Southeast Nigeria showed that parental smoking was a risk factor for childhood pneumonia. These findings were inline with the findings of a case control study carried out in Ethiopia that showed parental smoking was contributing factor of childhood pneumonia (Geleta et al 2016:1).

2.4.2.3 Overcrowding and poor ventilation as contributing factors of pneumonia

Overcrowding is one of the means that may create fertile opportunities for communicable diseases like pneumonia.

Children living inside the home environment where there were inadequate ventilation conditions could be more susceptible to childhood pneumonia, while toddlers who live in the home environment with adequate room ventilations have a lower risk of the same disease. The findings of a cross sectional study from India and another hospital based case control study from Brazil reported that overcrowding was a contributing factor for childhood pneumonia (Lima et al 2016:1 & Hemagiri et al 2014:47). Similarly, the result of case control study conducted in Urban Dhaka, Bangladesh revealed that overcrowding was a contributing factor to pneumonia (Ram et al 2014:968). An Ethiopian study also confirmed that there was significant relationship between occurrence of pneumonia among under-five children and crowded living space (Fekadu et al 2014:150).

2.4.2.4 Housing conditions of households as contributing factors of childhood pneumonia

It is known that different housing conditions like lack of separate kitchen and windows in the kitchen, number of rooms, roofing material, general sanitation, rearing cattle inside the main house, latrine availability and lack of proper hand washing can contribute to the occurrence of childhood pneumonia. The finding from case-control study design conducted in Indonesia showed that poor environmental sanitation was predictors of pneumonia among under-five children (Yeimo et al 2018:307). A research result which was carriedout in Urban Bangladesh revealed that a tin roof in the living space was found as the contributing factors for the occurrence of childhood pneumonia (Ram et al 2014:968).

Having separate kitchen with the presence of windows can be protective from the occurrence of pneumonia. For instance, institution based crosssectional studies carriedout in different areas of Ethiopia revealed that cooking food in living room, lack of separate kitchen and lack of windows in the kitchen were risk factors of childhood pneumonia (Lema et al 2018:25; Abuka 2017:19). Similar result has also been reported in another literature reviewed study carriedout in Ethiopia which alluded that food cooking in the main house was associated for being a factor of pneumonia (Merkeb and Adane 2019:2).

Different preceding studies showed that rearing of cattle inside the main house was the determinant factor of pneumonia. A community based cross sectional and another institution-based unmatched case-control studies conducted at different areas of Northwest Ethiopia showed that rearing of cattle inside the main house found as the contributing factors of childhood pneumonia (Markos 2019:1 & Fekadu et al 2014:150)

2.4.2.5 Hygiene and sanitation practices as factors of childhood pneumonia

Improper hygienic and sanitation practices are major determinants of childhood pneumonia. Study carriedout in Rwanda revealed that household toilet type was one of the associated factors with the occurrence of childhood pneumonia (Harerimana et al 2016:1). Similarly result of study conducted in Ethiopia reported that improper hand washing practice of mothers was one of the contributing factors of the occurrence of childhood pneumonia (Markos 2019:1).

2.4.3 Nutrition and immunization related factors associated with childhood pneumonia

2.4.3.1 Malnutrition as contributing factors of childhood pneumonia

Proper nutrition is essential for the development human life and keeping healthy. The essentiality and necessity of proper and balanced nutrition gets started from the very beginning of foetal development. It is important and necessary for all-rounded growth and development of body organs and physiologic functions (WHO 2000:3).

According to WHO (2005:1), malnutrition (under-nutrition) is one of the health outcomes and contributing factors for disease and which can also escalate the probability of getting illiness and death of children. Malnutrition is determinant factors of childhood pneumonia (Awasthi 2016:83; Srivastava et al 2015:1). Similarly, studies conducted in Bellary and Pakistan revealed that malnutrition to be found as significant risk factor of pneumonia among under-five children (Hemagiri et al 2014:47; Aftab et al 2016:1).

According to a review study conducted in developing countries, malnutrition was significantly associated with pneumonia (Ramezani et al 2015:1173). Similar findings were also obtained in Gambia through a case control study and reported that malnutrition remains important tractable risk factor for pneumonia. In addition, severe malnutrition was found to be the strongest predictor of severe pneumonia (Howie et al 2016:1405). Another cross sectional study design conducted in Southwest Ethiopia, shown that moderate acute malnutrition was significant factor of childhood pneumonia (Lema et al 2018:25).

2.4.3.2 Breast feeding and micronutrient deficiency as contributing factors of childhood pneumonia

There were evidences in Indian which showed that inappropriate complimentary feeding, iron deficiency (anaemia), low birth weight and pre-lacteal feeding to be predictors of under five pneumonia (Yadav & Awasthi 2016:83). There was also strong evidence between breast feeding practice and the magnitude of childhood pneumonia. According to the research conducted in different parts of India, poor exclusive breast feeding was found as predisposing factors of childhood pneumonia (Awasthi 2016:63, Srivastava et al 2015:1 & Hemagiri et al 2014:47).

Different studies from developing countries revealed that nutritional factors like micronutrient deficiencies, low birth weight, lack of breast feeding, rickets, anaemia and stunting were found to be the contributing

factors of childhood pneumonia (Aftab et al 2016:1 Ramezani et al 2015:1173, Ujunwa and Ezeonu 2014:95 & Fekadu et al 2014:150). Likewise the result of literature review study revealed that suboptimal breastfeeding increase the probablity of children getting pneumonia disease and death in all age groups. Especially the death due to pneumonia was higher among those children did not breastfed compared to those with exclusively breastfed as well as propely breastfed for their ages (Lamberti 2013:1). The similarly findings were obtained from the institution based study employed in Ethiopia, which revealed that mixed breastfeeding was found as significant factor for the occurrence of childhood pneumonia (Lema et al 2018:25).

2.4.4 Immunization and related factors of childhood pneumonia

Improving the vaccination status of children can decrease the related diseases and deaths of children. There are several types of vaccines those can protect against diseases including childhood pneumonia. Commonly utilized vaccines are pneumococcal conjugate vaccines, Haemophilus influenzae type b, measles vaccine, diphtheria, tetanus and pertussis. The evidences showed that around 20 million children did not get immunized, with global coverage of three doses of DPT and one dose of the measles vaccines continuing to stall at 86% in 2018. There were also nearly 39 million children global who did not received the three doses of PCV, putting children at higher risk of pneumonia (Save the Children and UNICEF 2019:5).

Previous Observational studies conducted at different areas of India revealed that the absence or inappropriate immunization was found to be a significant predictors of pneumonia (Gupta and Bhadrala 2019:949; Yeimo et al 2018:307). Incomplete immunization for age was found as determinants of pneumonia (Srivastava et al 2015:1; Gothankar et al 2018:1). This inappropriate immunization was mostly due to the absence of vaccines especially BCG (Gupta and Bhadrala 2019:949; Yeimo et al 2018:307). Likewise, another case control studies conducted in Northeastern Pakistan and Vietnam revealed that lack of immunization was obtained as contributing factor of childhood pneumonia (Hoang et al 2019:274 & Aftab et al 2016:1).

2.5 KNOWLEDGE, ATTITUDES AND PRACTICES OF MOTHERS/CAREGIVERS TOWARDS CHILDHOOD PNEUMONIA PREVENTION AND CONTROL

As part of improving child health program; WHO/UNICEF identified three critical steps namely recognizing the sickness of child, timely seeking appropriate care and treating the child with appropriate antibiotics(WHO & UNICEF 2006:2). Knowledge and attitude of the mothers/caregivers towards childhood illiness are important factors to determine child health. The knowledge, attitude and practices of mothers can directly influence the over all health status of the children hence they are the main caregivers for children in almost all societies (Mutalik & Raje 2017:30; Kanté et al 2015:3-5).

2.5.1 Knowledge of mothers/caregivers towards childhood pneumonia prevention and control and its associated factors

A study conducted in India revealed that majority of mothers even those who obtained formal education had poor knowledge as the illiterate counterparts (Mutalik & Raje 2017:30). Two other studies from India showed that knowledge of mothers towards the prevention and control of pneumonia was not satisfactory (Pradhan et al 2016:35 & Jena 2014:25). Similarly a study conducted in Saudi Arabia showed that 54% of the mothers/caregivers had insufficient knowledge towards ARI, including pneumonia (Alluqmani et al 2017:1959).

A comprehensive health education on aetiology or causative agent of childhood pneumonia, its prevention and management can help to establish appropriate contact between the health services provided and the community. This will help the families to identify the danger signs of acute respiratory diseases in children and to encourage appropriate and early care seeking behaviours (Mutalik & Raje 2017:32). A cross sectional study employed in Uttar Pradesh; identified that there was lack of knowledge among mothers/caregivers about under five pneumonia (Minz et al 2019:62). Another community based behavioural trial study conducted in India revealed that the level of recognition of mothers/care givers towards pneumonia and its danger signs was poor (Awasthi et al 2018:1).

Majority of past studies have analysed low knowledge of mothers regarding childhood pneumonia. Studies from Nigeria and Ghana showed that mothers were less knowledgeable towards aetiology and danger signs of childhood pneumonia (Abbey et al 2016:1 & Ukwaja et al 2012:889). Likewise in resource Low Countries, caregivers poorly recognized the symptoms associated with pneumonia.

A qualitative research conducted in Uttar Pradesh, India showed that knowledge of mothers/caregivers towards majority of danger signs of childhood pneumonia were reported to be poor,however chest in-drawing was identified as sign of sever disease (Awasthi et al 2015:1).

Majority of illiterate mothers were found to be less knowledgeable about teaching their children regarding proper hygienic practices, even though they need to care for their families. Age and educational level of mothers' were found as the determinant factors of their knowledge towards pneumonia. Good knowledge of mothers was associated with ARI including pneumonia. The findings of the study also showed that short duration of marriage, young ages and higher education were significantly associated with better understanding and sufficient knowledge of mothers/caregivers (Alluqmani et al 2017:1959 & Pradhan et al 2016:35).

2.5.2 Attitude of mothers/care givers towards childhood pneumonia prevention and control and its associated factors

From the study conducted in Bangladesh, it was reported that majority of mothers could describe that childhood pneumonia as it is critically life costing disease, however they did not identify whether child had pneumonia or not. Different factors like presence of coughing mothers, drinking cold water and engulfing dust particles were found as the causes for childhood pneumonia but mothers did not considered the illness as serious enough. (Ferdous et al 2014:1). Majority of mothers whose children with respiratory infections often did not provide early care for their children and stick to their intention to try herbal medicine or antibiotic before hand or held confidence in traditional healer. The finding of cross sectional study which was conducted in Kenya revealed that most of the mothers/caregivers were using home remedies and only went to health facilities when the case complicated even if they felt that health institution care is the best (Ndungu et al 2018:73).

Signs and symptoms of childhood pneumonia have been perceived differently among different mothers. Report showed that mothers of children with acute respiratory infection perceived cough as ordinary occurrence and felt confident about treating themselves and showed poor ability to differentiate pneumonia, simple cough and colds (UNICEF 2014:3). But in some areas mothers have relatively good attitude towards medical care for children with pneumonia. A study conducted in Pakistan reported that mothers had a positive attitudes towards early consultation with qualified medical practitioner regarding acute respiratory infections (Bham et al 2016:1557).

Different factors contributed the negative attitude of mothers/caregivers towards health care seeking practice about childhood pneumonia. The result of study carriedout in Ethiopia revealed that lack of money, thought that no benefit for such childhood illness and thought that mild or moderate illness were found to be contributing factors for not seeking health care treatment (Awoke 2013:155).

2.5.3 Practice of mothers/care givers towards childhood pneumonia prevention and control and its associated factors

In many countries especially rural communities have a decreased rate of health service utilization including IMNCI (Ndu et al 2015:1). A cross sectional study conducted in Saudi Arabia revealed that practice of mothers regarding childhood pneumonia prevention and control was poor. It was reported that most of them use self-medications at home without consulting doctors about the importance of home remedies rather many of them ask physicians for prescribing antibiotics (Alluqmani et al 2017:1959). Similar result was obtained from the mixed method study conducted in Ghana showed that mothers mostly practice home base care and remedies as well as allopathic care for children with pneumonia (Abbey et al 2016:1). Likewise a result of study carriedout in rural districts of Sierra Leone, showed that majority of children were not given a recommended treatment, rather some of them given traditional treatment. This was justified that most of communities' prefere for the local availed treatments and the barriers seeking to facility care were remained even after free health care initiative launched (FHCI) (Diaz et al 2013:1).

On time health care seeking from the recognized health institutions is very important and for this to happen, mothers should be familiar with signs and symptoms of childhood pneumonia. A research conducted in Alexandria, Egypt revealed that about 57.2% of children were given home remedies prior to seeking medical advice and also nearly 26% of mothers waited ≥ 2 days before seeking medical advice (Fadl et al 2020:1). Likewise, study carriedout in Tanzania showed that the provision of care due to community acquired pneumonia in most children was delayed. About 26.6% of children were learnt to be given leftover antibiotics at home without prescribing by the physicians (Ngocho et al 2020:56).

Different studies described that factors like limited education of parents, gap of understanding the severity of the illness and or insufficient knowledge about the disease were found to be significantly associated with the delays in care seeking. Factors such as rural residence, low income of households, giving home treatment before seeking medical advice and previous episodes of pneumonia were found to push parents to use traditional treatments or self-medications at home (Fadl et al 2020:1 and Bham et al 2016:1557). A result of qualitative research employed in Northern India, revealed that mothers consulted local medical practitioners directly: but decisions to visit doctors outside the village required consultation with husband or mother-in-law for health services of children with pneumonia (Awasthi et al 2015:1-2). Similarly, the findings of qualitative, exploratory design employed in South Africa showed that decision making about the health care utilization for a sick child was influenced by different individuals of households in addition to mothers/caregivers. It is assumed that a sick child would be treated at home or can be taken to different places like clinics, private doctors, traditional healers, faith healers and hospitals (Haskins et al 2017:1).

A qualitative study findings from the rural Oromia region of Ethiopia revealed that care givers or mothers of ill children described the cultural, social and community-based resources availability to reduce the uncertainty of time and strategies for accessing these resources in order to receive treatment for a sick child. (Shaw et al 2017:1).

2.6 THE AVAILED CHILDHOOD PNEUMONIA PREVENTION AND CONTROL STRATEGIES

Childhood pneumonia is the leading killer and with diarrhoea can contribute almost quarter of all deaths among the under five children in the poorest countries, especially in areas where there with poor access to health services, lack of nutritious foods and basic sanitation as well as poor hygiene and housing conditions. To reduce child deaths due to pneumonia and support countries to do so, WHO and UNICEF iniated the strategies called integrated global action plan for pneumonia (GAPP). The GAPP strategies proposed as a global framework for actions that partners can integrate and allign their efforts to reduce childhood morbidity and mortality due to pneumonia. In order to rise the awareness of communities towards pneumonia, that it is a major cause of child death, there are calls for scaling up the use of interventions of proven benefit and provide guidance on implementation and support of countries. The GAPP framework was envisioned as that every child, should be protected from pneumonia through creating a healthy environment and access to preventive and curretive measures (WHO and UNICEF 2009a:2-4).

The high concentration of pneumonia deaths and illinesses among less developed communities are the indicators, of that there is inequality within countries which also needs additional effort to reach the most vulnerable children. UNICEF established the commitment to address the equity gaps like reaching the basic causes of child mortality and to ensure that all children have the same opportunity to survive as well as reach their full potential. Fighting pneumonia could be one of the important areas which can help to reduce child mortality. The GAPP established a progressive goals for the control of childhood pneumonia and provided a roadmap for national governments as well as calls for integration and active engagement amongst all stakeholders through protect, prevent and treat framework to end childhood pneumonia by 2025.

In this regard, some of pneumonia related goals were; to decrease child deaths due to pneumonia to less than 3/1000 live births and reduce incidence of severe pneumonia by 75% compared to 2010 levels. Similarly, GAPP given the attention on universal access to drinking water in health facilities and homes, adequate sanitation in health inistitutions by 2030 and in homes by 2040 respectively, hand washing facilities (water and soap) in health facilities and homes, clean and safe energy technologies where in both health care facilities and homes by 2030 (WHO and UNICEF 2009a:1-5 and WHO/UNICEF 2013:7).

The following figure 2.1 presents the three focus intervention areas (protect, prevent and threat) of GAPP strategies to control childhood pneumonia.



Figure 2. 1: GAPP framework for protecting, preventing and treating childhood pneumonia (WHO and UNICEF 2009a:4).

2.6.1 Protective interventions for childhood pneumonia prevention and control

It would be possible to protect the children from pneumonia infections through ensuring a safe environment, exclusive breastfeeding for first six months, balanced nutrition and preventing low birth weight, reducing air pollution and improving hand washing practices regularly (WHO/UNICEF 2013:6).

2.6.1.1 Exclusive breastfeeding

According to WHO recommendations every infants should get only breast milk and nothing else, except those treatment purpose like oral rehydration solution (ORS), medicines, vitamins and minerals if needed. Breastfeeding is global accepted and preferred method for feeding an infant hence it can benefits for the development of children and keeping physical health of mothers. It is important that breastfeeding has to be started within one hour after birth and continued for the first six months of life (Taren and Lutter 2017:361-2).

There is an evidence that breastfeeding can protect from childhood pneumonia. Proper breastfeeding can help to protect children from all infections those may lead to hospital admissions (series/breastfeeding 2017:2). According to the literature review study conducted, a suboptimal breastfeeding can increase the probability of getting pneumonia and deaths across age groups. Furthermore it was shown that pneumonia deaths were higher among those did not breastfed compared to exclusively breastfed infants and among not breastfed compared to breastfed infants and young children (Lamberti et al 2013:1).2.6.1.2 Adequate complementary feeding and continued

breastfeeding

According to WHO (2009a:4), adequate and safe complementary feeding can be started from the age of 6 months with continued breastfeeding up to 2 years or beyond. The transition from exclusive breastfeeding to predominant breastfeeding should include the introduction of complementary feeding that can be solid, semisolid or soft foods with full of energy and nutrients those availed in the community or can be provided as supplements for vulnerable infants (Taren & Lutter 2017:362).

2.6.1.3 Vitamin A supplementation:

Vitamin A deficiency is one of the contributing factors of child mortalities. Improving Vitamin A status of deficient children can increases their resistance to infections and can reduce deaths from all possible causes by 23%. Vitamin A deficiency can be improved through routinely way or enhanced outreach base supplimetations. Sustaining the improved supplementation of vitamin A is essential, not only to eliminate vitamin A deficiency as a public-health problem, but also as a vital element of the child survival agenda (UNICEF 2007:3). The evidences showed that most of severe childhood infections are related with vitamin A deficiency. (Heidarian & Ansarinezhad 2014:57).

2.6.2 Preventative interventions for childhood pneumonia prevention and control

It is possible to prevent children from becoming ill with pneumonia by implementing actions like proper immunization, cotrimoxazole prophylaxis for HIV-infected and exposed children and zinc supplementation for children with diarrhoeal diseases (Save the Children and UNICEF 2019:5-6).

2.6.2.1 Immunization

Immunizing the children is a golden strategy to reduce illinesses and deaths of under five children. It is a cost-effective intervention of children across ages including infancy. The greatest improvement of child mortality in 2000 to 2015 was due to significant reductions in vaccine-preventable diseases through immunizations (WHO 2018c:1).

Childhood pneummonia can be prevented via pneumoncocoal vaccine which is more effective than the other prevention strategies, like reduced crowding and indoor air pollutants and nutrition. The use of pneumococcal vaccine would be taken as complementary to the use of other pneumonia control strategies, such as appropriate case management, promotion of exclusive breastfeeding for first six months and the reduction of known risk factors, such as indoor pollutants and tobacco smoke (WHO 2015:44). The pneumococcal conjugate vaccine can protects against severe forms of pneumococcal diseases such are pneumonia, meningitis and bacteraemia. There are two pneumococcal conjugate vaccines those available since 2009; 10-valent (PCV10) and 13-valent (PCV13). World health organization recommends 3 primary doses of pneumococcal conjugate vaccine (WHO 2013:8).

2.6.2.2 Sanitation and hygiene

According to WHO (2018d: 4), pure drinking water, proper sanitation and hygiene are essential to the health of human. They are not only a prerequisite to health, but also can contribute to livelihoods school attendance and helps to create resilient communities. WASH interventions include safe water, storage for portable water, latrines for sanitation and soap for hygiene, bath and hand washing practices. It also includes the promotion of key hygiene practices those can contribute in health, economic and social benefits (Gomathi et al 2017:575 and WHO & UNICEF 2013:75).

Most of deaths due to preventable illinesses are subjected to poor WASH conditions throughout the developing world. Globally the burden of disease due to poor WASH falls heavily on under five children. Washing hands with soap can reduce the risk of acute respiratory infections by 20%, thereby addressing one of the major causes of under five deaths (UNICEF 2015:3-4 &7).

2.6.2.3 Reduced household air pollution

According to WHO health fact sheet (2018a:1-4); household air pollution is one of the common global public health problems. The sudden exposure to pollutants in the air can affect the health of communities, across the world. Morely it affects the most vulnerable groups like children. Majority of the populations depend on solid fuels (wood, coal, crop waste and animal dung) to cook and heat their homes that can increase household air pollution which may double the risk of pneumonia. An intervention study conducted in Guatemala revealed that young children who had exposed to household solid fuels were twice more likely develop pneumonia compared to children who were not exposed (Smith 2011:1717).

2.6.2.4 HIV prevention

Pneumonia is one of the most common infectious complications in HIVinfected patients' leads to illiness and deaths (Seda 2016:11). Similar result was obtained from the reviewed literature study conducted and it revealed that pneumonia is the most associated pulmonary illness and remained as a common cause of illiness and deaths in the HIV-infected population (Breanne 2017:1).

Knowing and identifying the burden of the HIV epidemic is vital for planning, implementing and monitoring any health programs that attempts to tackle the impact of HIV/AIDS. This is also right for the under five children HIV epidemic, for which prevention, care, and treatment efforts lag behind those done for adults (WHO and UNAIDS 2013:3).

Cotrimoxazole prophylaxis is safe and cost effective in decreasing illinesses and deaths among HIV-infected infants and children. It can protect against the the severe opportunistic infections and supports to improve the quality of life of HIV-infected infants (WHO and UNICEF 2009b:1).

2.6.3 Treatment interventions for childhood pneumonia prevention and control

Childhood pneumonia can be treated at health centres or hospitals through case management in community (WHO and UNICEF 2009a:4). Possible treatments include improved care seeking and referral, timely and accurate diagnosis, appropriate antibiotics and oxygen therapy provisions (UNICEF 2016a:19).

2.6.3.1 Improved care seeking and referral

On time recognition of childhood pneumonia and its symptoms supports to early care seeking and can ensure that children get to a health provider early.That also helps the care providers to assess and treat children effectively according to WHO and UNICEF standards or based on IMNCI and ICCM guidelines (UNICEF 2016a:20).

2.6.3.2 Timely and accurate diagnosis mechanism to prevent childhood pneumonia deaths

The GAPP framework considered the faster and effective diagnosis mechanisms of childhood pneumonia as part of interventions. It was planned to ncrease the level of care-seeking for childoood pneumonia by 13% among the community and reduce treatment failure by 40%. Community case management brought 32% reduction in childhood pneumonia specific mortality among community (UNICEF 2016a:43 and Bhutta et al 2013:1417).

2.6.3.3 Appropriate use of antibiotics

Amoxcilline is effective treatment for bacterial pneumonia (UNICEF 2016a:20). According to the revised WHO (2014:6), guideline children with only fast breathing pneumonia and without any general danger signs could be treated with oral amoxicillin: at least 40 mg/kg/dose twice daily (80mg/kg/day) for five days. In areas with low HIV burden, amoxicillin can be given for three days and children with fast-breathing pneumonia who fail on first-line treatment with amoxicillin will better to have the option of referral to a facility where there is appropriate second-line treatment.

2.6.3.4 Oxygen therapy

Children with severe childhood pneumonia need oxygen therapy to treat hypoxemia (insufficient oxygen in the blood); which is a fatal complication. Improving the access to oxygen can significantly decrease pneumonia deaths among under five children (UNICEF 2016a:20). It is known that the most common sources of oxygen are cylinders, concentrators and pipelines (WHO 2016a:21).

2.7 CONCLUSSIONS

In this chapter, literature was reviewed and analysed on childhood pneumonia concepts, burden and associated factors from the view of worldwide to local concepts. It also discussed about the knowledge, attitudes and practices of mothers or caregivers towards childhood pneumonia prevention and control strategies. Lastly, the existing childhood pneumonia prevention and control strategies including global action plan for prevention and control of pneumonia (GAPP) framework were discussed.

The following chapter 3 presents the research paradigm.

Chapter 3

THEORETICAL FOUNDATIONS OF THE STUDY

3.1 INTRODUCTION

This chapter presents and discusses the concept of framework, theoretical framework, conceptual framework, Health Belief Model and global action plan of pneumonias' framework for pneumonia control.

3.2 A FRAMEWORK

A framework is the summary or logical structure of meaning that guides development of the study and enables the researcher to link the findings with the core body of knowledge, or it is a combination of concepts that used to explain relationships (Gray et al 2017:1060). A framework for study is a structure that provides guidance for the researcher as study questions or objectives are fine-tuned, ways for measuring variables those selected and analyses are planned (Imenda 2014:188). According to Polit and Beck (2018:550), it is a conceptual underpinning of a study and can be named as theoretical or conceptual framework.

3.3 THEORETICAL FRAMEWORK

A theoretical framework is the way of structuring or networking of different theories that have a relationships on the research topic (Kumar 2011:349). It is the framework with theory-based study or it is a theory that a researcher used to guide in his/her study. Thus, a theoretical framework is the application of theory; or a set of concepts drawn from one and the same theory to give an explanation of an event, or shed some light on a specific phenomena or research problem(Polit and Beck 2018:550, Imenda 2014:189).

3.4 CONCEPTUAL FRAMEWORK

A conceptual framework orginated from the theoretical framework and focuses on specific section that theoretical framework becomes the basis of the study. It is the aspects selected from the theoretical framework to become the basis of the research enquiry and is the basis for research problem (Kumar 2011:332). According to Polit and Beck (2018:550), conceptual framework is a framework with studies based on a specific conceptual model. Conceptual framework focuses on a limited scope, carefully put together in the form of a conceptual model, which would be immediately applicable to a particular study (Imenda 2014:189).

The conceptual framework for the current study was developed based on Health Belief Model (HBM) and global action plan of pneumonia (GAPP) control framework. These were used because of the relative applicability and interpretation of the models throughout the study. The purpose of the The advantage of HBM is that it specifys an incloherent set of common sense beliefs that happen to explain or mediate the effects of different factors on health behaviours and are amenable to change through educational intervention. The model would be applied in different health behaviours and to provide a framework for adjusting behaviour patterns that are relevant to public health (Mark and Paul 2015:31). According to the literature review study conducted by Elvis and Francis (2015:1), HBM bargains that the readiness of people to take an action to prevent disease and promote health depend on, their level of believe about the severity of disease. However, HBM have some limitations, like it does not consider for a one's attitude, beliefs and/or the individual determinants that can challenge a person's acceptance of a health behaviour; it does not consider about behaviours that are habitual which may inform the decision-making process to accept a proposed inteventions; neither does it account for other factors like environmental or economic those may control or encourage the recommended actions and also it assumes that everyone has access to equal amounts of information on the illness or disease (Wayne 2019:2). By considering these limitations, the researcher used HBM jointly with Global Action Plan of Pneumonia (GAPP) control framework to develop the feasible conceptual framework for the current study.

Global Action Plan of Pneumonia control framework is relatively a comprehensive approach to pneumonia control and it includes almost all the interventions that could be implemented as necessary part of child survival that are globally recommended and has been presented with protective, preventive and threating factors for the prevention and control of under-five pneumonia (WHO and UNICEF 2009:4).

Current study tried to address the prevalence of childhood pneumonia and possible determinants with assessing knowledge, attitude and practice of mothers/caregivers that can affect the level of prevention and control of the disease. By using HBM with the GAPP framework makes the following conceptual framework (Figure 3.1), was developed with considering the possible assocated factors of childhood pneumonia, which guides the current study more effectively to develop the awareness and prevention strategies on pneumonia among under-five children.

3.4.1 Health Belief Model (HBM)

The HBM initial developed in United States Public Health Service since 1950s by a group of social psychologists to understand "the widespread failure of people to accept disease preventives or screening tests for the early detection of asymptomatic disease"; and it was later used to patients' responses to symptoms and to compliance with prescribed medical regimen (Janz and Becker 1984:2).

The importance of the HBM is geared towards minmizing or controlling a disease conditions and aims to justify and forecaste health behaviours. The central assumptions of the HBM is that individuals will act if they feel their personal health is threatened and they perceive the benefit that health promoting activity outweighs the detriment of following through with the behaviour. The assumption is congruent with the model's focus on health promotion and disease prevention (Tarkang and Zotor 2015:2).

The basic concepts of the original HBM is that health behaviour is determined by personal beliefs or perception about a disease and the strategies available to decrease its burden; and personal perception is influenced by the whole range of intrapersonal factors affecting health behaviour. Primarily, there were four perceptions that serve as the main constructs of the model; namely perceived seriousness, perceived susceptibility, perceived benefits, and perceived barriers (Turner et al 2004:31). Later the model proposes the cues to action which may activate the behaviour when appropriate beliefs are held. These cues include a diverse range of triggers, including individual perceptions of symptoms, social influence, and health education campaigns (Mark and Paul 2015:31-2).

3.4.1.1 Perceived Susceptibility

Individual risks or degree of susceptibility is one of the common perceptions which initiates people to adopt a healthier behaviours. The more perceived risks, the more likelihood of engaging in behaviours to reduce the risk. A person's perception towards the relevance of health problem can help the individuals to take the required action to prevent the health problem thus there must be activities that improve the individual's perception of one's susceptibility to the health conditions (Tarkang and Zotor 2015:5 and Turner et al 2004:32). In the current study, it is thought that people who perceived their children to be susceptible to pneumonia would take more likely the pneumonia protective, preventive as well as management base actions to protect their children from childhood pneumonia.

3.4.1.2 Perceived Seriousness/Severity

Perceived seriousness refers to individual's belief/trust about the severity of a disease which usually depend on medical information or knowledge that may come from beliefs which a person has about the difficulties of a disease (Turner et al 2004:1).

It assumes that an individual can take the proper actions as him/her understoods the magnitude of the disease and its negative consequences (Tarkang and Zotor 2015:5). In the current study, it was learnt that if mothers/caregivers perceive childhood pneumonia as a serious infection that has severe negative consequences and implications on their physical and social lives, they would then adopt actions to prevent and treat the disease. Thus, knowledge and attitude of mothers or caregivers plays a crucial role in the prevention and control of childhood pneumonia.

3.4.1.3 Perceived Benefits

It is the indviduals' feelings or assumptions of the value or usefulness of a new behaviour that assumed to decrease the probability of getting a disease. People proclivity to accept the new behaviours when they believe that will increase the chances of getting healthier or decrease their chances of getting a disease (Turner et al 2004:32). The person needs to believe or accept that taking a certain action will help to tackle a problem from happening. It is this belief that may give a person's confidence to take the actions because of the expected outcomes (Tarkang and Zotor 2015: 5). For the current study, the perceived benefits are assumed to be beliefs of parents /mothers or care givers about the effective interventions of childhood pneumonia preventing and controlling health actions.

3.4.1.4 Perceived Barriers

The perceived barrier is an individual's own evaluation of the challenges in the process of accepting a healthier behaviour. It is the most important in which plays a significant role indetermining the behaviour change. In order for a new behaviour to be accepted/adopted, one needs to accept the benefits of the new behaviour outweighs the consequences of continuing the old behaviour (Turner et al 2004:33). Perceived barriers can also be referred to one's belief which persons realize that they have the power to challenge with different obstackles and that they would be able to take the interventions (Tarkang and Zotor 2015:5).

The current study identified that decreased level of knowledge, attitude and practices among mothers/caregivers can be a challenge and retards the implementation of pneumonia prevention and control strategies in the communities.

According to Tarkang and Zotor (2015:3-4), there are three different elements of HBM those are individual's perceptions about new behaviour, modifying factors and benefits of taking preventive measures. Each of these are described as follows:

3.4.1.5 Individual's perceptions

Individual's perceptions are opinions towards one's own susceptibility to a disease plus the seriousness with which one views the perceived threat of the illness (Tarkang and Zotor, 2015:3-4). In the current study, knowledge and attitude of mothers or caregivers on the prevention and control of childhood pneumonia were considered as contributing factors of childhood pneumonia.

3.4.1.6 Modifying Factors

Factors like socio-demographic, socio-psychological and structural are among the modifying factors those may affect one's perceptions and thus they can influence health related behaviours (Tarkang and Zotor, 2015:4). From the current study, socio-demographic factors, like educational and economic status of the parents can affect a family's perceptions of susceptibility and severity of suffering of their childrens' ill effects resulting from childhood pneumonia.

3.4.1.7 Benefits of taking the preventive measures

Perceived benefits and practicing variables affect the likelihood of initiating and maintaining action (Tarkang and Zotor, 2015:5). The current study assessed the knowledge and practices by mothers or caregivers towards prevention and control mechanisms like protective, preventive and to evaluate how they are taking care for their children and implementing childhood pneumonia interventions.

3.4.2 Global Action Plan for Prevention and control of Pneumonia Framework

WHO and UNICEF (2009:2&19), have been developed childhood pneumonia preventionand control approach called GAPP that would support the countries in their efforts for effective management of pneumonia. The GAPP framework proposes strategies of interventions that partners can collaborate their efforts. The GAPP was developed to improve awareness of communities towards childhood pneumonia prevention and control, that it is a possible cause of child death and call for scaling up the use of interventions of proven benefit and provide the way of implimentations.

The GAPP framework has been envisioned with the different areas for controlling pneumonia among under-five year's children and categorized into three major areas namely; protecting the children through a healthy environment, preventing them not to becoming ill with pneumonia and properly treating children who became ill with pneumonia.

In current study, different interventions on protection, prevention and treatment of childhood pneumonia were assessed and analysed. The protective factors including birth weight, breastfeeding, nutrition, indoor air pollution and hand washing practices were assessed. In addition, knowledge and attitude of respondents towards childhood pneumonia prevention and control were addressed. Regarding the preventive factors, vaccination status, HIV/AIDS status of family members, zinc and Vit A supplementation and deworming status were also assessed. Similarly, the study also assessed the treatment related factors that the mothers/caregivers practiced towards childhood pneumonia prevention and control.

Based on the theories and assumptions of both Health Belief Model (HBM) and GAPP framework, the researcher has developed the conceptual framework for the present study which has been used as the guiding principle for the study and came up with the potential findings. This conceptual framework has also been used as the roadmap for the development of awareness and prevention strategies for childhood pneumonia control.



Figure 3. 1: Conceptual framework developed by researcher (adapted and modified from HBM (Turner 2004: 34) and GAPP framework (WHO and UNICEF 2009a: 4).

3.5 CONCLUSSIONS

This chapter presented the framework, theoretical framework and conceptual framework. It also described the HBM and GAPP framework for prevention and control of pneumonia among under five children those are used as the base for newly developed conceptual framework of the current study. Furthermore it was also discusses about the four perceptions of HBM like perceived seriousness, perceived susceptibility, perceived benefits and perceived barriers were discussed in detail. In addition, the three intervention areas of GAPP, namely: protection, prevention and treatment were discussed. The newly developed conceptual framework was also showed in this chapter.

The next chapter will present the methods and materails of the current study.

Chapter 4

RESEARCH DESIGN AND METHODS

4.1 INTRODUCTION

The chapter presents and discusses about the research design and the methods used to atain the purpose and objectives of the current study. It also describes the type of research design used, study area conducted, population and sample groups selected. Further sample size calculation and sampling methods, method of data collection and analysis as well as ethical considerations and measures taken to ensure validity and reliability were also presented in this chapter.

4.2 RESEARCH DESIGN

According to Polit and Beck (2010:567), a research design is taken as the overall plan for addressing a research objectives for increasing the study's integrity. Researcher can choose and take the right way to answer a research question considering with different considerations like number of study groups will be included, resources and time of data collection (Gray et al 2017:1054). For the *current study, the researcher used a convergent type of mixed method.* The use of quantitative and qualitative data as a mixed research method is believed to triangulate the data as well as the findings that will help to create a better understanding of a research problem than using individual approaches (Creswell 2014:32).

In the present study, the *descriptive and explorative type of research* approaches were applied. A community based cross sectional study design and focus group discussion were employed for the quantitative part and for the qualitative part respectively. This was made so to conduct the study with the objectives of assess childhood pneumonia prevalence and its contributing factors, likewise it was to explore knowledge, attitudes and practices of mothers/caregivers towards of child hood pneumonia prevention and control. Then after to develop awareness and prevention strategies on pneumonia among under five children in Sidama

Regional State; in line with the strategy to accelerate the goal of ending preventable pneumonia disease by 2025.

4.2.1 Quantitative Research

Polit and Beck (2010:565), argued that it is the study of phenomena which concede themselves to precise measurement and quantification. It is also described as the systematic study or process that counts or measures formally to answer the research question or objective and usually uses numerical data (Gray2017:1082). In the current study it was used to measure or quantify the prevalence of childhood pneumonia and assocaiated factors numerically. Similary it was used to assess KAP of mothers/care givers towards and its determinants towards pneumonia prevention and control among under five children. The findings also generated with numeric values.

4.2.1.1 Cross-sectional study design

According to Polit and Beck (2013:239), a cross-sectional study design would be used to collect data at one point in time and or multiple times in a short time period. The cross sectional study design is right the approach to describe the status of certain events or phenomena and its relationships at a fixed point of time. In the current study all quantitative data were collected at one point of time or data collection period and which aimed to determine the prevalence of childhood pneumonia and its contributing factors, and also to explore KAP of mothers/caregivers towards childhood pneumonia prevention and control.

4.2.2 Qualitative Research

From the books a qualitative research is defined as, it is an indepth study and rigorous approach used to describe ones' cultures, life experiences and social processes from the individual perspectives or it is a method of exploring and understanding the meaning of individuals' or groups' justifications to a social or human problem (Gray 2017:1081-Creswell 2014:32). In the current study, a qualitative research was basical used to answer objective seven (explore KAP of mothers/caregivers towards childhood pneumonia prevention and control). It was applied to complement quantitative data or findings and to explore the factors related to the life experiences of the communities that may contribute to childhood pneumonia- especially knowledge, attitudes and practices of mothers/caregivers towards childhood pneumonia prevention and control. **4.2.3** Mixed Research method

It is the use of two research designs usually quantitative and qualitative methods together, in order to describe the issues in detail and triagulates the data to create better understanding on the issues (Gray 2017:1070). In

the current study quantitative and qualitative approachs (mixed research method) were applied in convergent way of mixed method. Both quantitative and qualitative data were collected concurrently but analysed separately. Finaly the results were merged during interpretation. This helped for better strengthening of data as well as study findings and for vigorous and dependable conclusions. It was used for complementarity, incrementally and enhanced validity. Thus doing so avoids limitations of a single approach and provides an opportunity for testing alternative interpretations of the data. Supported to obtain more complete understanding from two data sets and corroborate results from different methods.

4.3 RESEARCH METHOD

Research is a serious, or systematic assessment or study to re-assure and amend the existing knowledge and effectuate new knowledge. Whereas, research methods are the specific ways that researcher choses to conduct the research, within the sellected design. A research methods could be topic selection, choice of study area, sample size calculation, data collection and analyisis with appropriate statistical tests (Gray 2017:1070 &1085). The current study has used different research methods to achieve the planned goal or proposed objectives and help in the development of the proposed awareness and prevention strategies on childhood pneumonia within the study area. Those methods and approaches are discussed as follows:

4.3.1 Study area/setting

Study area/setting is a place which could be organization, agency or community where a study has planned to conduct or it is the location and conditions in which data collection for the study will take place (Kumar 2011:208 & Polit and Beck 2010:568).

Current study was *conducted in five* districts and one town administration of *Sidama Regional State which has been established newly from SNNPR as of June 2020. They are* Dale, Aleta Chuko, Bensa, Bona Zuria and Loka Abaya districts and Aleta Wondo town. *The Region found in between* 6°14' & 7°18' N and 37°92' & 39°14' E with the altitude ranging from 1,200 metres to 3,211 metres and is one of densely populated regions in Ethiopia. According to 2019 restructuring, there are 30 rural districts, 6 towns, one city administration with 8 sub-cities. The region has 19 functional governmental hospitals, 135 governmental and 3 nongovernmental health centres and 620 health posts (Sidama Zone Health Department 2019:12).



Figure 4. 1: Map of study site, Ethiopia, 2020 4.3.2 Study Population

Study population is the whole aggregation of subjects that a researcher is interested and not restricted to human or respondents can be specified with the reasons that set the study population based on the eligibility criteria (Polit and Beck 2010:306),

Sidama regional state is among the densely populated regions in Ethiopia with having nearly 4.5 million population in 2019. Among these 607,435 are children aged under five years and there are estimated 772,533 women in reproductive age group. Majority of people in the region are rural dwellers and agriculture dependents.

Source population of the current study comprised of all mothers/caregivers with under-five children who living in the six districts of Sidama regional state, namely: Dale, Aleta Chuko, Bensa, Bona Zuria and Loka Abaya districts and Aleta Wondo town in Sidama region, Ethiopia.

4.3.3 Study sample and sampling techniques

Sample is a subset of the accessible group or population that the researcher selects for participation in the study; whereas sampling is the way of taking a portion of the population to represent the entire group. Or it is a way of selecting study participants like groups of people, behaviours or other elements those are appropriate to conduct a study (Polit 2018:243 Gray 2017:1087).

4.3.3.1 Sample size calculation

Sample size is the number of particiapants/respondents to be taken from the universe based sample. A sample size expected to be enough/optimum for the study and needs to fulfil the requirements like efficiency, reliability, representativeness and flexibility (Kothari 2004:56), The sample size was calculated for the quantitative part of current study by considering all necessary estimations of a proportion with specified absolute precision. The prevalence of childhood pneumonia and its contributing factors for the areas covered by the current study site was not known from previous studies. Due to this reason, a 50% prevalence rate was used to calculate the sample size. The sample was calculated by using the formula for estimation of single population proportion.

$$n = \frac{(2 \times F(1-F))}{d^2} x k$$

$$n = \frac{(1.96^2 \times 0.5 \ (0.5))}{0.05^2} x 2$$

$$n = \frac{(1.9208)}{0.0025}$$

$$n = 768 \text{ Respondents}$$

Where:

n = minimum sample size

Z= critical value at 95% confidence interval adding 10% non-response

which is 1.96,

P = the prevalence of childhood

pneumonia at the area and assumed to be

For the qualitative part of this study, six FGDs were conducted to attain data saturation and 4 FGDs each group included seven (07) mothers while 2 FGDs included 6 mothers with child less than 5 years age purposively selected. The opinions and experiences of respondents were solicited simultaneously and the moderator guided the discussion.

Regarding the Delphi group, 10 expertise based on their previous maternal, neonatal and child health experiences and exposures on the area, who have been working in University, regional health Bureaus, Zonal health department, Hospitals and NGOs were participated. Delphi method is a way of getting informations/judgments from an expert panel about specific issues where experts requested individually in number of rounds, by circulating a summary of the panel's views between rounds to foster consensus (Polit and Beck 2018:546).

4.3.3.2 Sampling techniques/procedures

The probability sampling approach was applied for the collection of quantitative data from the current study area and purposive or non probability method was used for sampling of FGD participants.

For the quantitative study a multi-stage sampling techniques were used to collect information from the study site. A multi-stage sampling technique is one of the sampling strategies that move on through a set of stages/levels from larger to smaller or higher to lower sampling units (Polit and Beck 2010:560). The researcher has randomly selected a total of 6 districts (5 rural and 1 urban), namely: Bensa, Bona Zuria, Aleta wondo town, Aleta Chuko, Dale and Loka Abaya districts firstly and 21 *kebele* from these districts. The current study used simple random sampling method through table for random numbers method to select those six districts and 21 *kebele* accordingly where each element in the population is given an equal and fair chance to be selected.

Cluster sampling method was used to select 845 study samples proportionally from 21 *kebele*. Each *kebele* was divided into small villages called "gote" or "limat budin" and randomly selected for data collection. From the selected "gote" or "limat budin" data were collected until the proportional allocated sample size reaches as shown in figure 4.2. According to Kothari (2004:65), cluster sampling can be conveniently taken by grouping the area into smaller non-overlapping areas and then to randomly select some of smaller areas until satuareting the proposed sample size



Figure 4. 2: Schematic presentation of how to select sample population for quantitative part of study in Sidama Regional State: 2020

Non-probability or purposive sampling method is the way of sampling by non-random means to select respondents. It is a judgmental way of sampling which involves conscious selection by the researcher of certain respondents to be included in a study (Polit 2018:291; Gray 2017:1081) For the qualitative part of the current study, the purposive or purposeful sampling was used for selection of respondents for collection of data.

The researcher decided on six focus group discussions (FGD) with each FGDs containing seven respondents/discussants (mothers with their child age less than 5 years). The respondents were chosen with the consideration of their past active participation and knowledge on health issues at their village. All six FGDs were selected from three districts namely: Loka Abaya, Dale and Aleta wondo town; hence it was assumed to be representative of the remained three districts as that leaving conditions of communities are more or less similar in different districts. Two FGDs from each districts and one FGD from each *kebele* were involved in 5 rural districts while 2 FGDs were involved from one *kebele* at Aleta wondo town. The selection of each discussant from the selected

kebele was done with the help of health extension workers who knew households very well and had participated earlier in health issues in their *kebele*.

4.3.3.3 Eligibility criteria for being a respondent for interview

Polit and Beck (2018:547) defined eligibility criteria as it is the criteria that designated to select the specific target population to be included in a study. Households were selected through cluster base sampling and respondents who were availed at home during data collection and conseted were made to be included in the current study.

4.3.3.3.1 Inclusion criteria

The study inclusion criteria are measurements identified by the researcher which would be presented for the respondents to be included in the sample (Gray 2017:1063). In the current study the following were considered as inclusion criterias:

- Mothers/caregivers with under-five children;
- Caregivers aged 18 years and above;
- Caregivers with full information of the child and has stayed with the index child for at least 6 months and above before data collection;
- Children under five years old who were not critically ill

4.3.3.3.2 Exclusion criteria

Exclusion sampling criteria are descriptive criteria that eliminate some respondents from the inclusion of study sample for the reason of eliminating sample characteristics that may introduce error (Gray 2017:1057). The following were the exclusion criteria agreed in the current study:-

- Mothers/caregivers who had critically ill or unconscious during the time of data collection were excluded;
- Caregivers who were not familiar about the detail information of child or who did not leave with the child for atleast last 6 months prior to data collection
- Children who were critically ill also excluded and advised to be taken to health facilities

4.3.3.4 Study Variables

Variable is abstract ideas that have been made measurable. In quantitative study, variables are studied in order to investigate incidence, the assocaitions that may exist among them and or cause-and-effect relationships (Gray 2017:1096). Thus, study variables can be defined as the components of the indicators which can be measured (Walliman 2011:66). In the current study different dependent and independent variables were considered by reviewing different literatures including Ethiopian Demography and Health Survey.

4.3.3.4.1 Dependent Variables

They are called the outcome variables or the variables hypothesized to depend on which can be caused by another variable. It is a response, behaviour or outcome that is predicted and measured in research (Polit and Beck2018:546, Gray 2017:1053). In our current study, dependent variables were listed as follows: -

- Childhood pneumonia;
- Knowledge, attitude and practice of child's mothers or caregivers towards childhood pneumonia prevention and control.

4.3.3.4.2 Independent Variables

According to Polit and Beck (2018:552), an independent variable is the variable which is considered to cause an effect or influence the dependent variable. It is the variable or variables that may predict the occurrence of the dependent variable (Gray 2017:1063).

Independent variables considered in current study were categorized into socio-demographic and economic, environmental, behavioural and health status, nutritional and immunization related variables.

Socio-demographic and economic variables included parental education and occupation, religion, maternal age, family size, economic status, ethnicity group, sex of child and age, marital status and residence of the respondents.

Environmental variables like type of household fuel, tobacco smoking, latrine availability, housing conditions, presence or absence of cattle,

availability of separate kitchen, presence or absence of ventilation and waste management were addressed.

Nutrition and immunization related variables included feeding practice, immunization status of child, Vit A supplementation and deworming status, duration of breastfeeding, and nutritional status of child were included.

Health status variables like family history of acute respiratory infections, history of HIV/AIDS and disease history of child were assessed.

4.3.4Data collection process and tool4.3.4.1Data collection approach

Data collection is a way of precise and systematic collecting of relevant information to the study purpose and objectives (Gray 2017:1052). For the quantitative part of current study data were collected from home to home survey. Individual interview and observational approaches were applied to collect the relevant informations to the specific objectives of the study. An interview is a way of data collection in which data collector asks questions about respondent, either face-to-face, by telephone, or via the Internet. However observational methods can be used to collect information by observing the conditions of individuals (Polit and Beck 2018:553).

For the qualitative part of the study data were collected via face to face participation of FGD participants. The discussions were conducted at selected kebele.

The Delphi survey was conducted through mail communication to collect inputs for the development of awareness and prevention startegies. **4.3.4.2** Research data collection tool

The structured interview schedule were used to collect the quantitative data of current study. An interview schedule was taken as an arranged instrument which specified the wording of all questions used to ask respondents in a structured self-report studies (Polit and Beck 2018:553). The structured interview schedule of the current study was developed in English language by reviewing the standard survey questions including DHS and previous similar studies. It was translated into *Sidaamu Afoo*-the local language spoken in the study area. Then by checking the validity of the questionnaire, the interview schedule was translated back into English to avoid possible difference in the meaning.

The questions of the structured interview schedule were grouped into six parts namely: socio-demographic and economic information, Childs' characteristics, information on environmental factors, nutritional related factors, information on immunization status of the child, and KAP of respondents. There are total of 110 items or questions includeding the questionnaries for wealth index assessement. The major contents included in the current interview schedule were:-

Part one: Socio-economic and demographic information (26 items):

Household socio-economic and demographic data, respondents' sociodemographic, children's demographic data and wealth index.

Part two: Child's characteristics (13 items)

Age, sex, place of delivery, birth interval and order and disease status of child.

Part three: Information on environmental factors (24 items)

Latrine availability, waste management, housing condition, fuel used and charcoal utilization.
Part Four: Nutritional related factors (16 items)

Anthropometric measurements, breastfeeding, additional food initiation, Vit A supplementation and deworming.

Part Five: Information on Immunization status of the child (06 items)

Immunization status, PVC 10 vaccine, measles vaccine.

Part Six: KAP of respondents (25 items)

Knowledge, attitude and practice;

For qualitative part of the current study, data were collected by using FGD guide. FGD guide includes the procedures and guiding questions that could support to explore the KAP of mothers/caregivers towards childhood pneumonia prevention and control. The guiding questionnaire was also translated into local language (*Sidaamu Afoo*) and then back to English by independent translators to optimize the accuracy of the translation.

The FGD guiding questionnaires included the following basic areas:-

Introduction part or aim of FGD, group consent for the participation, ground rule, knowledge about childhood pneumonia and its causes, common symptoms, prevention as well as treatment methods and their practices towards prevention and control of childhood pneumonia and challenges related with it. The questions were open-ended and non-leading. As the discussion progressed, some of the questions were modified to effectively attain the purpose of the interview.

The researcher considered the above employed methods of data collection based on the factors like nature, scope and object of enquiry; availability of funds, time and precision required (Kothari 2004:112):

4.3.4.3 Training provided for data collectors and supervisors

Researcher developed a training manual which includes the introduction part, aim of training, importance of the manual, short definition of research, sampling, data collection methods and tools, research ethics and data quality issues. The manual was prepared in English language. Data collectors and site supervisors were trained by the researcher for two days on the approach and method of data collection, at the training hall of Yirgalem General Hospital. The detailed explanations were provided about the purpose and objectives of the study, sampling method, way of interviewing, content of the data collection tool, data quality and ethical issues. Furthermore they oriented about research methodology and data collection procedures.

Data colectors and supervisors had been familiarised with each questions of the interview schedule especially on the translated one through reading and discussing. The clarifications of each questions regard to its purpose were made clear. An orientation was given on how to fill out the questions of the structure of the interview schedule, both open and closed-ended questions, the skip patterns, observation questions, questions with one and multiple responses, how to measure weight, height and count pulse rate of the index children. Morely emphasis was also given them about how to establish mutual trust with respondents before asking questions.

Regarding the qualitative data collection, orientation was given for the FGD assistances and health extension workers to equip them with the basic aim of research and selecting participants as well as way of guiding the discussion.

4.3.4.4 Pilot study

According Polit and Beck (2018:561), pilot study can be taken as a smallscale study conducted in the way of preparing for a main study, usually to assess feasibility data collection tool. Regarding the quantitative section of current study, 10% of actual sample size or about 85 questionnaires were piloted at Tula sub city where outside the study districts before actual data collection. The aim of piloting was to test the data collection tool as well as equip and familarize the data collectors and supervisors with the data collection instruments. Based on the pilot study findings, some changes those related with culture sensitivity, language error and might missed the expected meaning were corrected and modified with more relavant and understandable way of writing. However the qualitative instrument was not piloted rather edited and re-checked by researcher.

4.3.4.5 Data collection process

Quantitative data collection

Twelve data collectors and three site supervisors attained the collection of quantitative data. Six of the data collectors had a bachelor's degree, six held diploma in nurses and all three site supervisors had a master of public health degree. They were purposeful selected with the consideration of earlier experience in data collection, familiarity with local language and who took IMNCI training.

There were six teams established for those six districts of data collection area, each team with two data collectors in Sidama regional state, to collect data from the households. One site supervisor was assigned for two teams to manage and support them. All data collectors and site supervisors are native for the Sidaamu Afoo language and had earlier experience in research data collection. Furthermore they were also trained on basic IMNCI pneumonia management guideline. Each team was equipped with the interview schedules, weighing scales, MUACs and tape meters. Data were collected from home to home in the selected cluster of the study area, by interviewing mothers/caregivers using interview schedules. The pilot study participants were not included in the main study. During thedata collection measuring the weight, MUAC and height of the index child was carried by data collectors. The immunization status of the child was assessed by referring their immunization cards. Where there was no either of them (mother or caregiver) data collectors made return visits to the households for at least three times for the interview, and enforced to visist and collect data from next household. Respondents' confidentiality was maintained; and kept records/data confined to them confidential. In addition, the reliability of the weighing scales and measuring tapes checked for accuracy and calibrated after every measurement. Each measurement was done twice, and the average was taken.

Site research data collection supervisors were charged with the supervising of the process of data collection and resolve the problems encountered. They were also responsible to organise the logistics needed for data collection and orient data collectors about the field work in the districts and checked the filled instruments for completeness and correctness. Furthermore they have checked the daily filled interview schedules and took actions if there were inconsistent or incompleteness. They guided data collectors to re-check those incomplete schedules and made to re-interview the households based on the type of problems identified and reported to the researcher.

The researcher was able to coordinate the whole process of data collection and checking the sample for data quality. Where the filled interview schedules were found to be inconsistent, incomplete or having missing information, the supervisors and data collectors were communicated to re-checked and filled data or made to re-interview the households based on the type of problems identified.

Qualitative data collection

Six FGDs were conducted at three study districts, two FGDs per district. The researcher and assistants together with health extension workers facilitated the discussion according to the FGD interview guide. The discussants had chosen the safe place and confortable way of sitting by themselves. Most of them chose a semi-circular sitting arrangement and used compound of health posts' where there were shades or sitting facilities for the discussion. FGDs were audiotaped with discussants permission and the researcher and assistants took notes as well. In each FGDs discussants were coded with number; like FGD1 participant 1, 2, 3, 4, 5, 6 and 7 for analysis, while conducting discussion with field note book, they coded and registered with their age. Based on their interest and saturation of needed data 1 hour to 1:30 hours were taken to finalize the discussion within each FGDs.

To ensure the discussants safety from covid-19, researcher provided a facemask for all discussants and ensured social distancing among them while sitting. Discussants also utilized accordingly to keep them selves from Covid-19.

Delphi data collection

In the current study, 10 Delphi panellists were included, who are child and maternal health experts and other relevants from university, RHB, Health departments, NGOs and hospitals. The aim of Delphi method data collection was to gather inputs from experts' and professionals. Data were collected through electronic mail. Firstly, experts were identified, and telephonic interview was done to introduce the researcher and the purpose of the call. Secondly, the study aim was explained to the experts and consent from each individual was obtained. The envisaged framework was then shared to the experts for their input and had one to two weeks to respond with their comments and suggestions through electronic mail.

The Delphi data validating tool was prepared by researcher which contained 08 (eight) Issues to be considered while panellists have commenting the developed awareness and prevention strategies (Annex I).

4.3.5 Ethical considerations

According to Polit and Beck (2018:548), ethics is considered as a system of moral values which is dealt with the degree to which research/study procedures has followed an ethical, legal as well as social obligations to the study respondents. In the journey of study, the application of ethics begins with properly identifying research topic and continues through publication of the study findings (Gray et al 2017:265). This sub-topic presents the principles, permissions needed and granted by relevant bodies of the study.

4.3.5.1 Protecting the Rights of Institutions

Institutional review is a process of examining the design and methods of a proposed study for ethical considerations, and also for overseeing studies in progress. Institutional review is undertaken by an independent committee of peers at an institution to identify the level to which the proposed study protects the rights of respondents. The peers' reviews committee of research are responsible to ensure that the researcher is conducting the research in an ethically manner (Gray 2017: 1064). The final research proposal of the current study was submitted and approved by Research Ethics Committee, Department of Health Studies at UNISA (Annex A). The institutional permissions were obatained from the former Regional Health Bureau of SNNPR and Sidama Zone Health Department to carry out the research (Annex B & C).

4.3.5.2 Protecting the rights of respondents

- A. self-determination, right to: A right which is respect for person according to the ethical principle and also states that humans should be handled as autonomous and have the freedom to decide their lives as they chose, without any influences, because humans are able of making their own decisions (Gray 2017:1088). The current study involved mothers/caregivers from households as human respondents they had interviewed about the childhoods' pneumonia and associated factors, KAP of respondents towards childhood pneumonia prevention and control. Thus the respondents had informed that their participation in this research that would be only voluntary base and there might not be any negative consequences if they decide not to participate (Annex D).
- **B. Confidentiality:** Is the privacy management of the data provided by respondents which the information would not be shared with third bodies without the respondent's permission. Thus the access to data will be guarded carefully, to prevent breaches of confidentiality (Gray 2017:1049). In the current study, the confidentiality were maintained in each steps. The confidentiality agreement was signed between data collectors and principal investigator, similarly it was doen between supervisors and principal investigator by creating the awareness on data collectors' code of conduct and assurance of confidentiality agreement (Annex D).

The privacy code for each sample were used during data collection rather than taking name of individuals. Recruitment of study respondents and interview were done in privately, with only the data collectors, supervisors as well as principal researcher and the potential respondents. The respondents were clearly informed that the collected information or data would be disseminated via seminars, conferences or articles in such a way that no one could identify who was interviewed and it would be only in group or percentage form. Both paper base and audiotaped collected data has been locked at the office of the researcher and the key has only been with him so that no third party can access to ensure the confidentiality. Computer base record has been secured by personal password and the computer is used by the researcher only.

- **C. Anonymity:** Meaning literally "without a name"; in research, the removal of all names and identifiers from data or it is the protection of respondents' confidentiality such that no one including the investigator cannot append individuals with data provided (Polit and Beck 2018: 537, Gray 2017:1043). In the current study, the respondents were informed that the collected data would not be simple disclosed to the third body rather will be kept confidential as no names and addresses of any of the respondents while used in the event of thesis writing, result presentation and publication. Thus, the anonymity of respondents in this study were controlled through assigning individual codes numerically (Annex D & E).
- **D. Informed consent:** is the respondent's engagement to participate voluntarily in a study, as they assimilates essential information about the study (Gray 2017:1064). In current study respondents were informed that involvement/engagement was only voluntary base and written informed consent obtained without any influence and or promise for any special kind of payment or award. Respondents were also clearly informed that they can withdraw at any time if they feel any discomformt. Furthermore they were informed that withdrawal would be without giving reason(s) and there will be no penalty for withdrawal. Based on this they gave a consent with signaning and writing their names (Annex D). Additionally they were informed about the parental consent wether they agree or disagree about the participation of teir child (Annex O). The respondents were informed about data protection that no

one could access the data and investigator would kept data in a computer access by password protection. Benefits and risks due to particiapating in current study were also informed to respondents.

- **E. Autonomy:** In both qualitative and quantitative part of data collection; respondents were well informed about the aim and objectives of research. They were also informed about their right not to participate, not to answer a question that they think is sensitive and or to withdraw from the study at any time during data collection. Additionally, data collectors assured written informed consent for the quantitative study and both supervisors and researcher assured oral informed consent for qualitative study. Thus data collection for current study was conducted safely and voluntary basis only (Annex D & G).
- **F. Beneficence**: According to Grove et al. (2017:1044); beneficence is the principle of the ethical position that compels the researcher to actively strive to do well and confer benefit in respect to the study respondents. Its ethical counterpart is non-maleficence which compels the researcher to actively strive to do no harm to research respondents. In current study, the following beneficence perspectives were considered:

The right to protection from exposure: mean not to exposing respondents to conditions or situations for which they have not been informed well and ready. Data collectors and supervisors had introduced themselves to the study respondents and the fact that their role was only data collection. Respondents were informed about the objectives and scope of the study before the interview has been started. (Annex D & G).

The right to freedom from harm and discomfort: Beneficence is one the basic ethical principle which seeks to maximize advantage for study respondents and minimize harm (Polit and Beck 2010:548). Data were collected at the household level and had no known risks which would expose respondents to harm. However there was low risk considered hence the study included direct human participant mean that under five age children and their anthropometric measurements like height, weight and physical examinations (checking bilateral pitting edema, pulse rate count). Parental consent was obtained and sufficient counselling and care were given to minimize the discomfort which mothers/caregivers may felt due physical examinations done for child. To minimise any discomforts related with mis-understanding during interview; the questionnaire were translated to Sidaamu Afoo local language, training was given for data collectors on interviewing skills and way to establish mutual trust with respondents. Furthermore data colectors advised to take only a reasonable time to complete the interview schedule (Annex F & H). Benefits from current study include improved community awareness about the burden of childhood pneumonia and its risk factors and further development of awareness and prevention strategies to prevent and control childhood pneumonia.

Justice

According to Grove et al. (2017:1066), justice is one of ethical principles that states human respondents should be treated fairly, as groups and as individuals. The justice for the current study has been secured by randomly selecting the clusters for study respondents' enrolment with an equal chance to be included in the study. Moreover the appropriate scientific research methodology were followed in each steps.

4.3.5.3 Scientific integrity

The current study contributed to the improved body of knowledge and study regarding childhood pneumonia and associated factors for its prevention and control through honest finalization, result presentation and publication. The scientific integrity of the study was maintained trough the protecting the intellectuals' property of the author's publications that were consulted in the current study by giving proper recognition and citation in the list of references. The reported data/result was collected, analysed and interpreted based on appropriate scientific procedures and evidence in an ethical manner.

4.3.6 Processing and analysis of data

Collected data would be processed that editing, coding, classification and tabulation should done under data processing making them amenable to analysis. The term analysis used to refer the computation of finate measures along with searching for patterns of relationship that exist among data-groups those are subject to statistical tests of significance to determine what validity data can be said to indicate any conclusions (Kothari 2004:122).

4.3.6.1 Data processing

A. Processing the quantitative data

The collected raw data supposed to be edited for the detection of errors and omissions and correct them when possible. Data edition helps to ensure the accuracy and consistent of gathered facts (Kothari 2004:122-23). In the current study editing was carriedout at field level by supervisors and at central level by principal researcher to correct the evident errors like an entry in the wrong place were corrected. For instance entry recorded in year when it should have been recorded in months for the age of index child were corrected and the like also done accordingly.

According to Kothari (2004:123), coding can be done by giving numerals or other symbols to answers that responses can be limited into number of categories or classes. All the necessary coding was done prior to data analysis in the current study. Data were arranged into similar categories or groups based on their common characteristics. For instance, study variables such as age of respondent, income and age of child were classified according to their common characteristics.

B. Processing Qualitative part of data

According to Ranjit and Kumar (2011:248), findings from the qualitative data can be reported through developing a narrative to describe a situations and identifying the main themes that emerge from the field notes. In the current study data from field note and recorded audio were analysed manual with the support of Atlas ti soft ware. The narrative reported prepared, voice transcription and verbatim has done, three themes and six sub-themes were identified.

4.3.6.2 Data analysis

The analysis is an interpretation of data that include comparison of different outcomes of the various interventions upon the several groups to make a decision as to the achievement of the research goals (Singh 2006:150). For the current study data analysis were done as the following ways:-

4.3.6.2.1 Analysis of quantitative part

Before starting data entry, the questionnaires those filled by data collectors were reviewed for completeness and consistency of quantitative data. The missed values were verified against the sources accordingly. After cleaning, the interview schedules were coded numerically with in each districts as well as *kebele*. Then double data entry was done by using Epi data version 4.2.2 software to minimize the missed values and exported to SPSS version 25 software for analysis. The statisticians and senior researchers were consulted for the proper

applications of both Epi data and SPSS software programs as per their experiences and earlier exposures. After exporting the quantitative data, the missed values were checked through each study variables by running them with frequency and rearranging with ascending order. The missed values were rechecked and addressed from hard copy data and cleaned as per its appropriateness.

The analysis were done by using SPSS version 25 software programme. Descriptive, bivariate and multivariate statistics were carriedout for quantitative data.

Descriptive statistics were used to present information using frequencies, tables, graphs, mean and standard deviations.

Bivariate and multivariate analysis were conducted to determine the risk factors of pneumonia among children aged under-five. Bivariate analysis were employed by using Chi-square tests or Fisher's exact tests. However multivariate analyses were carriedout through using binary logistic regression with backward stepwise regression technique in order to identify the childhood pneumonia contributing factors independently, while controlling for confounders.

Independent variables those with a p-value less than 0.25 in the bivariate analysis were checked for multivariable analysis. The crude odds ratio (COR) and adjusted odds ratio with 95% confidence interval (CI), regression coefficient and p-value of the variables were calculated in the logistic regression analysis. Through the analysis, the two-sided tests were applied and a p-value less than or equal to 0.05 was considered as statistically significant.

4.3.6.2.2 Analysis of qualitative part

Qualitative data were analysed manually supported with Atlas ti software. Data were analysed by arranging a thematic content and framework analysis. The thematic content analysis was used in exploring or describing the way of awareness creation and prevention of childhood pneumonia.

Verbatim was done with caution. Firstly the audio taped data were listened careful to transcribe. The thematic content analysis was done secondly with the identification of pertinent information. Then significance of the information categorized or coded from the transcript based on the objectives of the study. The frequency or degree of uniformity of ideas that discussants stated were helped to form code. Those codes that can come together were merged to form a new code. Then final mutuality exclusive codes that could not be reduced to any form were taken as final themes in the current study.

4.3.6.2.3 Analysis of Delphi method data

The Delphi data were analysed based on the feedbacks collected from the individual Delphi panelists via email. The common feedbacks from Delphi panels included editing for typographic errors, additional inputs, questions and comments on the content highlighted in track changes. Researcher then analysed and corrected the comments by incorporating the important inputs and reverted it back to the Delphi Panels for final checking and editing of the developed strategies. Researcher gave additional justification to Delphi panellists when needed.

4.3.7 Validity, Reliability and Trustworthiness

4.3.7.1 Validity

The degree of measurements by possible instruments to what it is supposed to measure is called validity. Polit and Beck (2010:377), put that content validity, face validity and construct are major types of validities.

Face validity focuses what the instrument looks as though it is measuring the appropriate construct. Who will be completing the instrument or it is a subjective assessment, usually by an expert that verifies a measurement instrument appears to measure the content it is purported to measure(Grove et al 2017:1058; Polit and Beck 2010:377). For the current study, questionnaires were developed by the researcher based on modified standard survey questions from WHO, EDHS and previous similar researches. Moreover, comments and suggestions from statisticians and epidemiologists those thought as important were incorporated to improve the validity of the instrument. Ilk type

Content validity used to assure that the degree of all an appropriate items has been included in an instrument for the construct being measured. It concerns about the adequacy of domained to be covered and it also examines the levels that the measurement covers/addresses relevant elements to the construct being measured (Grove et al 2017:1050 & Polit and Beck 2013:377-78). To address the content validity of the current study, detailed literature was conducted and conceptual framework was done before developing data collection tool. According to the published evidences of related literatures, the data collection tool was developed which contained the common elements those are relevant to the study objectives and conceptual framework. Furthermore the developed instruments were reviewed and checked by supervisors.

4.3.7.2 Reliability of the research instrument

Reliability can be considerd as the consistency of measuremnets that instrument measures the attribute or concerns a measure's accuracy. Quantitative researchers better to consider the three aspects of the reliability interests like stability, internal consistency and equivalence (Polit and Beck 2013:373). Data collectors as well as supervisors were trained on way of data collection and consistensive use of tool, which was to assure the reliability of the current study. The data collection tool was piloted and the necessary refinement was done accordingly before data collection. The questionnaires were translated to local language as both data collectors and supervisors were local language speakers. The similar questions were used for all respondents and the information was clearly given to them. Field supervisors tried to verify the reliability of sample questions respondents during data collection. The investigator also has done close monitoring and daily follow-up through data collection process and has checked the filled data collection tools for completeness and consistency.

The data entry clerk entered the sample data into Epi Data version 4.2.2 software. Another ten randomly selected questionnaires were entered into Epi Data version 4.2.2 software and checked for consistency. Its completeness was also re-checked during analysis through SPSS software.

4.3.7.3 Trustworthiness

According to Polit and Beck (2013:570), trustworthiness can be defined as the level of confidence that investigators have developed in their data qualitative, through credibility, transferability, dependability, conformability, and authenticity criteria. In the current study the general trustworthiness has been ensured as follows and the following general trustworthiness aspects have been assured in this study.

4.3.7.3.1 Credibility

The confidence of the data truthness and proper interpretations of them refers as credibility. Qualitative investigators supposed to develop confidence in the truth of the findings from each respondents and syntax in their research (Polit and Beck 2013:492). Credibility of the current study were ensured by giving enough time during data collection from FGD discussants to establishing amity with them. The investigator's earlier exposure/experience as a public health care practitioner and facilitator with childhood pneumonia prevention and control services in Ethiopia has been of a paramount assistance in conceptualising the study. The discussion was conducted in local language and which helped the respondents to understand and give their experience freely. In addition, the principal investigator and assistances have used FGD guide to probe respondents. The field notes and audiotapes were used to capture the data and helps to enable the principal investigator to understand in-depth. The investigator also tried ensure the authenticity of data through member cross checking mechanism as that transcripts were given to senior researcher and expertise.

4.3.7.3.2 Dependability

Dependability is a criterion to be used for checking the integrity in qualitative studies, through the stability of data over situations as well as analogous to reliability in quantitative research (Polit and Beck 2010:568). In the current study the supervisors and copy editors were involved as hearers of the reviewed documents produced during the study. Supervisors and independent qualitative researcher were participated to review the transcripts for analysis.

4.3.7.3.3 Confirm-ability

Confirm-ability is objectivity, which is the potential for congruence between two or more independent people about the data accuracy, relevance, or meaning (Polit and Beck 2013:492). The audit trial was used to ensure the confirm-ability of current study. The principal Investigator used field notes and audio tape recorder in order to keep the original voice of the discussants and properly interview transcripts from individuals. The independent qualitative researcher and supervisors were involved for confirm-ability cehck. Researcher tried to focus on discovering omissions of comments and to verify placement of phrases into themes with the hlp of member check.

4.3.7.3.4 Transferability

According to Polit and Beck (2013:492), transferability is can be taken as a similarity of the generazebility or the extent to which qualitative findings can be transferred to or have an applicability in other settings. In current study transferability has been assured by its methodology that researcher used a mixed type of study which can be applicable in other areas too. Further the study provided adequate information in each study processes like study purpose, objectives, sampling strategies, data collection procedures and data analysis. The researcher also used member check.

4.3.7.3.5 Authenticity

According to Polit and Beck (2013:493), authenticity is the level to which investigator symmetry and faithfully show a range of different realities. It can be emerged in a report when it shows the feeling tone of respondents' lives as they are lived. The FGDs of the current study were guided by local language and participants can reflect their opinion clearly. The transcription was also done by the most experienced individuals who knew the both languages very well, which helped to capture the multiple perspectives of participants and foster change across them to ensure the authenticity of the study. Additionally the investigator cross checked the transcribed data with original audio and field notes and corrected any differences found.

4.4 CONCLUSSION

This chapter presented the research design and methods used in current study which was -the mixed research approach. It also discussed the study setting, population, sample and sampling procedures, and inclusion and exclusion criteria, determination of sample size. Besides, study variables, data collection tool and process were also described accordingly. The chapter also presented the ethical considerations followed under this study, method of data processing and analysis, mechanism of validity and reliability assurance and trustworthiness of the study.

The the next chapter will present the analysis and research findidngs of the study.

Chapter 5

ANALYSIS AND PRESENTATION OF THE RESEARCH FINDINGS

5.1 INTRODUCTION

This chapter presents the result of the study that alines with the study objectives. The study findings include the prevalence of pneumonia among under-five years children and its different associated factors like; socio-demographic and economic, behavioural and health related, nutritional and immunization related and environmental risk factors in Sidama regional state of Ethiopia. The study result also includes the level of knowledge, attitude and practice of mothers/caregivers of under-five children towards childhood pneumonia prevention and control and its associated factors in the current study area. Furthermore the qualitative findings also presented in this chapter.

5.2 DATA COLLECTION, MANAGEMENT AND ANALYSIS

Current study includes both quantitative and qualitative data. Quantitative data were collected from 15th March to 20th May 2020 from 13 *kebele* of six randomly selected study districts in newly established Sidama regional state of Ethiopia.

For the qualitative part of study six (06) FGDs with total of 40 women participants who had under-5 children; meaning that two FGDs with six participants and four FGDs with seven participants were involved and data were collected based on the prepared discussion guideline.

It is true that quality issues of data can affect the analysis and interpretation of research findings. Different quality measures were taken in each steps of quantitative and qualitative data process to ensure its quality.

5.3 RESEARCH RESULTS

5.3.1 Response rate and residence of respondents

Overall 845 study respondents randomly selected from households for the current study, while 843 were involved in the interview and giving a response rate of 99.8%.

Out of the 843 respondents, 217(25.7%) were from Dale, Loka Abaya 128(15.2%), Aleta Wondo town 59(7%), Bona Zuria 108(12.8%), Bensa 197(23.4%) and Aleta Chuko districts 134(15.8%) respectively. Regarding their kebele; 73(8.7%) from Wara, Soyama 73(8.7%), Gane

71(8.4%), Arada Gale 56(6.6%), Chalbessa 72(8.5%), Mesalemia 59(7.0%), Olonso Hore 56(6.6%), Bona Kike 52(6.2%), Chalba 62(7.4%), Hatessa 66(7.8%), Shanta Wene 69(8.2%), Chuko Lamala 64(7.6%) and Rufo Wacho 70(8.3%). Thus majority of respondents 784 (93.0%) were rural dwellers whereas 59 (7.0%) were urban habitats. **5.3.2** Socio-demographic and economic characteristics of

respondents

The study result revealed that 707(83.9%) of the respondents were mothers of the children and 136(16.1%) were care givers. Majority of the respondents 801(95.0%) were formally married and living together, 19(2.3%) were single and 23(2.7%) widowed. Regarding the sex of respondents 738(87.5%) were female and 105(12.5%) male.

Majority of the respondents 747(88.6%) were Protestants followed by Apostolic 57(6.8%), Orthodox 32(3.8%) and Muslim 7(0.8%) respectively. Out of 843 respondents 803(95.3%) belongs to Sidama ethnicity, 26(3.1%) Amhara and the others namely Wolaita, Oromo and Silte constitutes the rest 14(1.6%).

The analysis showed that the literacy rate was 657(77.9%) while 186(22.1%) did not have formal education. Most of the participants

522(61.9%) had elementary school level (grade 1-8) and 41(4.9%) had tertiary level of education or graduate certificate and above. Regarding respondents' occupation, 605(71.8%) responded that they were housewife followed by merchants 92(10.9%), government employees 74(8.8%), farmers 61(7.2%) and 11(1.2%) were daily laborers respectively.

The analysis showed that the mean family size of study respondents at the study area were 5.46 (standard deviation ± 1.87). The household family size ranged from three to twelve persons. The result of recoded categorical data of family size showed that 456(54.1%) households had small family size, 251(29.8%) of them had medium family size and 136(16.1%) had large family size. Half of mothers from the surveyed households had only one or two children, followed by 179(21.2%) of mothers who had three, 132(15.7%) had five and above and 120 (14.2%) had four children respectively.

Findings of analysis showed that the age of respondents reached from 16 to 65 years with a mean age of 29.6 years (standard deviation \pm 7.09). The most frequent age in the current study was 30 years while the median age was 29. The majority of respondents 505 (59.9%) were 25-35 years age followed by 16-24 years 193(22.9%) and 145(17.2%) equal or more than 36 years.

Table 5. 1: Socio-demographic and economic characteristics of respondents in Sidama Region	onal State: 2020 (N=84	3)
	D	

Characteristics		Frequency	Percentage
Relation of respondent to child	Mother	707	83.9
(N=843)	Care giver* (father, sister, brother of child or others)	136	16.1
	Total	843	100
Marital status of respondents	Single	19	2.3
(N=843)	Married	801	95.0
	Widowed	23	2.7
	Total	843	100
Sex of respondent (N=843)	Male	105	12.5
	Female	738	87.5
	Total	843	100
Religion of respondents (N=843)	Protestant	747	88.6
	Apostolic	57	6.8
	Orthodox	32	3.8
	Muslim	7	0.8
	Total	843	100
Ethnicity of respondents	Sidama	803	95.3
(N=843)	Amhara	26	3.1
	Others** (oromo, silte and wolaita)	14	1.6
	Total	843	100
Occupation of respondents	Housewife	605	71.8
(N=843)	Merchant	92	10.9
	Government employer	74	8.8
	Farmer	61	7.2
	Daily labourer	11	1.3
	Total	843	100
	*No formal education/Illiterate	186	22.1
Educational level of respondents	Elementary school (1-8 grade)	522	61.9
(IN=845)	Secondary school (9-12 grade)	94	11.2

	Above secondary school (certificate and above)	41	4.9
	Total	843	100
Age of respondents (N=843)	16-24 years	193	22.9
	25-35 years	505	59.9
	>=36 years	145	17.2
	Total	843	100
Family size of households	Small family size (<=5 members)	456	54.1
(N=843)	Medium family size (6-7 members)	251	29.8
	Large family size (>=8 members)	136	16.1
	Total	843	100
Parity of mothers (N=843)	One	197	23.4
	Two	215	25.5
	Three	179	21.2
	Four	120	14.2
	Five and above	132	15.7
	Total	843	100

*Care givers refers fathers, sisters, brothers of child or others who were participated under the study where the mother was absent. *Formal education= respondents did not attend any education and who cannot read and write also

5.3.2.1 Wealth index of households

Total of 13 questions were used to assess and evaluate the information of wealth index of households in the current study area. Assets of each households' that information collected were given a weight or factor score made over principal components analysis. The resulting asset scores were uniform in relation to a standard normal distribution with a mean value of zero and a value of standard deviation one. These standardized scores were also used to generate the break points that define wealth quintiles of

population which remained presented in five groups with the similar number of individuals in each group as; poorest (lowest), poor (second), middle, rich (fourth), and richest (highest).

Findings of the current study revealed that 156(18.5%) of households were in the poorest/lowest/ and 177(21%) in poor/second/ categories while 173(20.5%) in middle level, whereas 168(19.9%) and 169(20.1%) of them belonged to the rich/fourth/ and richest/highest/ categories respectively. (See Figure 5.1).



Wealth Index



5.3.3 Socio-demographic characteristics of surveyed child

The current study analysis revealed that half of the children 422(50.1%) were male and 421(49.9%) were female children. Majority of surveyed children 626(74.3%) were delivered at health facility and the rest

217(25.7%) were at home. In addition 664(78.8%) of sample children had two years and above birth interval with their older child and the rest 179(21.2%) were delivered with in less than two years' interval.

The study result showed that age range of children involved in the study were between 1-59 months. The average age of children was 28.9 months (standard deviation ± 15.8).

Furthermore, half of the children 429(50.9%) were between the age group of 13-36 months followed by 37-59 months 239(28.4%), 7-12 months 122(14.5%) and 53(6.3%) were less than or equal to6 months (see table 5.2).

Table 5. 2: Socio-demographic	characteristics of surveye	ed children in Sidama re	gional state: 2020 (N=843)
Tuble 5. 2. Docio demographic	characteristics of surveye		

Characteristics		Frequency	Percentage
Sex of child (N=843)	Male	422	50.1
	Female	421	49.9
	Total	843	100
Age of child (N=843)	<=6months	53	6.3
	7-12 months	122	14.5
	13-36 months	429	50.9
	37-59 months	239	28.4
	Total	843	100
Where was the surveyed child born (N=843)	Health facility	626	74.3
	Home	217	25.7
	Total	843	100
What is the birth order of sample child (N=843)	1 st	196	23.3
	2 nd	214	25.4
	3 rd	179	21.2
	4 th and above	254	30.1
	Total	843	100
What was the birth interval of surveyed child within its older	< 2 Years	179	21.2
child(N=843)	>= Years	664	78.8
	Total	843	100

5.3.4 Environmental and housing characteristics of study respondents

5.3.4.1 Availability of latrine and solid waste management

The availability of latrine and solid waste management analysis revealed that 740(87.8%) of households had latrine. Of those households with

latrine, majority 628(84.9%) had simple pit latrine and 112(15.1%) ventilated improved latrine (VIP). Regarding solid waste management, 525(62.2%) of respondents had their households managed solid waste backyard simple system followed with 197(23.4%) blow a fuse and 121 (14.4%) burying system (see table 5.3).

Tuble of of it fundbing of number in bond manufement in brauna Regional Barropia, 2000 (11-0 h	Table 5. 3	3: Availability	of latrine and solid	l waste managem	ent in Sidama Re	gional State,	Ethiopia	: 2020 (N=843
--	------------	-----------------	----------------------	-----------------	------------------	---------------	----------	----------	-------

Household characteristics		Frequency	Percentage
	Yes	740	87.8
Presence of HH latrine (N=843)	No	103	12.2
	Total	843	100
Type of latrine utilized by HHs (N=740)	Simple pit latrine	628	84.9
	VIP	112	15.1
	Total	843	100
	Backyard simple	525	62.2
Ways of solid waste management at HH level (N=843)	Blow a fuse	197	23.4
	Bury	197	14.4
	Total	843	100

5.3.4.2 Housing characteristics of the respondents'

Housing conditions were also assessed and half of the houses were roofed with corrugated iron 430(51%) followed with 404(47.9%) roofed with grass and 9(1.1%) were roofed with bamboo. Regarding floor materials

458(54.3%) of houses were with mud and 338(40.1%) and 47(5.6%) floored with cement and wood respectively.

Further analysis revealed that majority of houses from the study area, that means 529(62.8%) had windows for the ventilation purposes whereas

314(37.2%) did not. Three hundred and eighty-four (45.6%) of houses had 2 rooms and 236(28%) had three rooms. In addition, 181(21.5%) had one room and 42(5%) lived in a house with four rooms.

Half of the total respondents' 432(51.2%) of households shared the living houses with different kind of animals but 411(48.8%) households did not. Three hundred and twelve respondents lived with a cow/bull and 76(22.8%) with sheep, goats, donkey and hens within different proportions (see table 5.4).

Characteristics		Frequency	Percentage
Roof materials of living house (N=843)	Grass	404	47.9
	Corrugated	430	51.0
	Bamboo	9	1.1
	Total	843	100
Floor material of living house (N=843)	Mud	458	54.3
	Cement	338	40.1
	Wood	47	5.6
	Total	843	100
Presence of window in the living house (N=843)	Yes	529	62.8
	No	314	37.2
	Total	843	100
Number of rooms in the living (843)	One	181	21.5
	Two	384	45.5
	Three	236	28.0
	Four	42	5.0
	Total	843	100
Animal living with the same house (N=843)	Yes	411	48.8
	No	432	51.2
	Total	843	100
Kind of animal which lived with the same	Cow/Bull	311	75.7
house (N=411)	Sheep	58	14.1
	Goat	33	8.0
	Donkey	5	1.2
	Hen	4	1.0
	Total	411	100

Table 5. 4: Housing characteristics of respondents in Sidama Regional State, Ethiopia	: 2020 (N=843)

5.3.4.3 Households energy source

Household energy sources were assessed and the analysis revealed that majority of households 686(81.4%) used firewood while 147(17.4%) used charcoal as major energy sources. More than quarter 263(31.2%) of households used charcoal for cooking and other related purposes.

Of 263 households that used charcoal; 107(40.7%) used it twice per day followed with 102(38.8%) once a day, 43(16.3%) three times and above per day and 11(1.3%) used it only once per every two days.

More than half 489(58%) of households used kitchen for cooking and 336(39.9%) cooking in the main house and 18(2.1%) cooked outdoor.

Three hundred twenty-two (38.2%) of households had chimney during data collection.

The current study analysis revealed that 12(1.4%) used electricity as source of energy for lighting in their households while 428(50.8%) of households used kerosene lamp (locally known as *kuraz*), 364(43.1%) used hurricane lamp or lantern (locally known as *fanos/masho*) and 39(4.6%) wood.

The place of children where they usually stay when their mothers are preparing food was also assessed and 486(57.7%) of respondents responded that the child stays outside of the cooking place while 357(42.3%) carried the child on their back or by the side of mothers (see table 5.5).

Table 5. 5: Households' of	energy sources and	1 related characteristics i	n Sidama regio	nal state, Ethior	oia: 2020 (I	N=843)	
----------------------------	--------------------	-----------------------------	----------------	-------------------	--------------	--------	--

Characteristics of HHs energy sources		Frequency	Percentage
	Charcoal	147	17.4
Type of firewood (N=843)	Wood	686	81.4
	Electricity	10	1.2
	Total	843	100
Charcoal use (N=843)	Yes	263	31.2
	No	580	68.8

	Total	843	100
Frequency of charcoal use(N=263)*	Once per day	102	38.8
	Twice per day	107	40.7
	Three times and above per day	43	16.3
	Once per every two days	11	4.2
	Total	263	100
	Main house	336	39.9
Usual place for cooking (N=843)	Kitchen	489	58.0
	Outdoor	18	2.1
	Total	843	100
Presence of chimney during data collection (N=843)	Yes	322	38.2
	No	521	61.8
	Total	843	100
	Hurricane lamp/Masho/Fanos	364	43.2
Source of light at home (N=843)	Kerosene lamp/Kuraz	428	50.8
	Wood	39	4.6
	Electricity	12	1.4
	Total	843	100
Usual location of child while mothers cook (N=843)	On back or beside of mothers	357	42.3
	Outside the cooking area	486	57.7
	Total	843	100

*There were only 263 households those used charcoal

5.3.5 Nutrition related characteristics of surveyed child

Respondents were asked about the status of breastfeeding and complementary feeding practices of child. Majority of mothers 578(68.6%) had initiated their children on breast milk immediately after birth while 159(18.9%) within one hour after delivery, 85(10.1%) after an hour but within 24 hours and 21(2.5%) of them initiated after a day.

Out of 843 mothers' children aged under 5 years included in this study, 328(38.8%) had breastfed their child for more than two years; followed by 154(18.3%) who breastfed their child up to 2 years, and 63(7.5%) of them did it up to 12 months and 57(6.8%) for less than 6 months. In addition, 242(28.7%) of the respondents include both mothers who did not know for how long they breastfed their children and those who were still breastfeeding during data collection.

Regarding frequency of breastfeeding of children per day, 239(28.4%) had breastfed their children for up to 10 times, 115(13.5%) of them breast fed five times or less and 91(10.8%) for more than 10 times. The remaining 398(47.2%) could not remember how many times they breast-fed their children and were not breast-feeding during the data collection period.

Majority of mothers 763(90.5%) had commence complementary feeding during data collection and only 80(9.5%) did not. For children on complementary feeding, 620(81.3%) started exactly at recommended 6 months while 103(13.5%) had initiated their children for complimentary feeding early (less than six months) and 40(5.2%) did after 6 months. Furthermore, 344(38.4%) of mothers or caregivers used spoons for feeding the children, 210(24.9%) used their hands, 135(16%) used bottles and 94(11.2%) used cups (refer table 5.6)

Ta <u>b</u> l	le 5. 6:	Child	breastf	eeding	and con	iplementa	y feed	ling p	oractices a	mong	mothers	in Sidan	ia reg	gional stat	e, Ethioj	pia: 202	0, (N=	843)

Characteristics of child feeding practices of househo	lds'	Frequency	Percentage
	Immediately after birth	578	68.6
Initiation of breast-feeding (N=843)	Within an hour	159	18.8
	Between 1 and 24 hours	85	10.1
	After a day	21	2.5
	Total	843	100
	< 6 months	57	6.7
Duration for breast-feeding (N=843)	7-12 months	63	7.5
	Up to 2 years	154	18.3
	>2 Years	327	38.8
	Others*	242	28.7
	Total	843	100
	<= 5 times	115	13.6
Frequency of breast-feeding per day (N=843)	Up to 10 times	239	28.4
	>10 times	91	10.8
	Others**	398	47.2
	Total	843	100
	Yes	763	90.5
Had started complementary feeding (N=843)	No	80	9.5
	Total	843	100

When had started complementary feeding (N=763),	< 6 months	103	13.5
	At 6 months	620	81.3
	>6 months	40	5.2
	Total	763	100
Materials used for feeding the child (N=763)	Bottle	135	17.7
	Cup	96	12.6
	Spoon	321	42.0
	Hand	211	27.7
	Total	763	100

Others^{*} = the respondents could not remember the frequency that they breastfed their children Others^{**} = the respondents did not remember the specific number of times (frequency) they breastfed their children

5.3.6 Vitamin A supplementation, deworming and immunization status of child

while 478(56.7%) were either not dewormed or had exceeded the age for deworming (see table 5.7)

The status of vitamin A supplementation and deworming for the children were assessed and verified from the children's immunization card. The results showed that 545(64.7%) of children received vitamin A, 245(29.1%) did not and 53 (6.3\%) were not eligible for vitamin A supplementation. In addition, 365(43.3%) of the children were dewormed

The current study analysis showed that 765(90.7%) of children had received at least one vaccine while 78 (9.3%) had not get vaccinated against any disease. Of those children who received vaccination, 802(85.1%) received three doses of both penta 3 and PCV10, 82(9.7%) received two doses and 18(2.1%) received one dose. About 656 (77.8%) of the children were vaccinated for measles as depicted in table 5.7.

Table 5. 7: Vitamin A supplementation, deworming and immunization status of surveyed children in Sidama regional state, Ethiopia: 2020(N=843)

Characteristics of surveyed child		Frequency	Percentage
	Yes	545	64.7
Supplemented with vitamin A (N=843)	No	245	29.1
	Not eligible*	53	6.3
	Total	843	100
	Yes	365	43.3
Deworming received (N=843)	No	169	20.0
	Not eligible**	309	36.7
	Total	843	100
Received any of vaccine and dose (N=843)	Yes	765	90.7
	No	78	9.3
	Total	843	100
	One	18	2.1
Received dose of Penta3 & PCV10 (N=843)	Two	82	9.7
	Three	743	88.2
	Total	843	100
	Yes	656	77.8
Received measles vaccine (N=843)	No	98	11.6
	Not eligible***	89	10.6
	Total	843	100

Not Eligible* = Children who were not eligible for Vitamin A (whose age was < 6 months) Not Eligible** = Children whose age were not eligible for deworming (<23 months) Not Eligible*** = Children who were not eligible for measles (whose age was < 9 months)

5.3.7 Behavioural and disease status of child's family

Data on families' history of smoking, presence or absence of any respiratory disease and HIV/AIDS status were assessed and results in this regard revealed that 19(2.3%) families had history of smoking, 38(4.5%) respiratory disease and 4(0.5%) HIV positive status.

Of those who had the history of smoking, 13(68.4%) were father of the child, 3(15.8%) were mothers and others 3(15.8%) were brothers for the sample child. Similarly, of the 38 families who had reported the history of respiratory diseases, 17(44.7%) were child's father, 13(34.2%) were mothers and 8(21.1%) were other family members (see table 5.8 below).

Table 5. 8: Behavioural and disease history of child's family member in Sidama Regional State, Ethiopia: 2020, (N=843)

Behavioural and disease status characteristics of H	Frequency	Percentage	
Any smoker among family members (N-843)	Yes	19	23
The shoker allong failing members (11–043)	105	17	2.5
	No	824	97.7
	Total	843	100
If yes who was the smoker (N=19)	Father	13	68.4
	Mother	3	15.8
	Brother	3	15.8
	Total	19	100
The presence of a family member living with HIV	Yes (positive for HIV)	4	0.5
(N=843)	No (negative for HIV)	814	96.5
	Unknown status (did not tested for HIV)	25	3.0
	Total	843	100
Who was positive for HIV (N=843)	Father	2	50.0
	Mother	2	50.0
	Total	4	100
A family members had history of respiratory	Yes	38	4.5
disease (N=843)	No	805	95.5
	Total	843	100
Who had history of respiratory disease (N=38)	Father	17	44.7
	Mother	13	34.2
	Others*(brothers & sisters)	8	21.1
	Total	38	100

5.3.8 Prevalence of childhood pneumonia and associated factors in Sidama Regional State

5.3.8.1 Prevalence of pneumonia among under-five children

According to the respondents responses, 203(24.1%) of children had the



history of cough and cold with or without fever for less than two weeks duration. However detailed analysis revealed that 187(22.2%) of the children were further assessed and were confirmed to suffer from pneumonia by counting their respiration rate (RR), (see figure 5.2) which shows the prevalence of childhood pneumonia.

Figure 5. 2: The prevalence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843)

5.3.8.2 Bivariate analysis of factors associated with pneumonia among children under- five years old

The bivariate analysis of the current study was done using Chi-square test to investigate the presence of significant association between childhood pneumonia and possible risk factors. The result of the bivariate analysis of the study was presented following the proposed objectives. The variables with p-value less than or equal to 0.25 at the bivariate analysis were included in the multivariate analysis.

- 5.3.8.2.1 The association between the socio-demographic and economic factors and the occurrence of pneumonia among under-5 year's children
- 5.3.8.2.1.1 Households' socio-demographic and economic characteristics and association with the occurrence of pneumonia among under-5 year's children

Table 5.12: below presents the results of bivariate analysis amongst the households' socio-demographic and economic characteristics and

childhood pneumonia disease occurrence in the study area. The result showed that the children living in household with medium family size (COR=2.30; 95%CI: 1.58, 3.36) and large family size (COR=4.00, 95%CI: 2.54, 6.30) were more likely to develop pneumonia. Similarly, mothers/caregivers with no formal education (COR= 3.25, 95%CI: 1.42, 7.42) were more likely to develop childhood pneumonia.

Being rural residence (COR=0.54, 95%CI: 0.30, 0.95) was found as the protective factor for the children not to develop pneumonia. However, the rest of socio-demographic and economic characteristics like relation of respondent, sex of the mother/caregiver, marital status of the mother, religion, ethnicity, and residence, age of mother/caregiver and education level of child's father were found not to influence the occurrence of pneumonia among children aged under 5.

Table 5. 9: Bivariate analysis of mother's/caregivers' socio-demographic and economic characteristics of mothers/care givers in Sidama regional state: 2020

No	Variables	Pneumonia sta	tus	COR of 95% CI	p-value	
		Yes (n %)	No (n %)			
1	Relation of respondent to child					
	Mother	154(21.8)	553(78.2)	1		
	Father	26(24.8)	79(75.2)	1.18(0.73, 1.91)	0.50	
	Care giver	7(22.6)	24(77.4)	1.05(0.44, 2.48)	0.92	
2	Sex of mothers/care givers					
	Female	161(21.8)	577(78.2)	0.85(0.53, 1.36)	0.50	
	Male	26(24.8)	79(75.2)	1		
3	Marital status of mothers/care givers					
	Single	3(15.8)	16(84.2)	1.25(0.22, 7.05)	0.80	
	Married	181(22.6)	620(77.4)	1.95(0.57, 6.62)	0.29	
	Widowed	3(13.0)	20(87.0)	1		
4	Religion of mothers/care givers					
	Protestant	164(22.0)	583(78.0)	1.06(0.54, 2.04)	0.87	
	Orthodox	10(31.3)	22(68.7)	1.71(0.64, 4.55)	0.29	
	Muslim	1(14.3)	6(85.7)	0.63(0.70, 5.70)	0.68	
	Others	12(21.1)	45(78.9)	1		
5	Ethnicity of mothers/care givers					
	Sidama	175(21.8)	628(78.2)	0.65(0.32, 1.31)	0.26	
	Others	12(30.0)	28(70.0)	1		
	Residence of mothers/care givers					
6	Urban	20(33.3)	40 (66.7)	1		
	Rural	167(21.3)	616 (78.7)	0.54 (0.30, 0.95)	0.03*	
	Family size of household					
7	Small	57(12.5)	399(87.5)	1		
	Medium	71(28.3)	180(71.7)	2.30(1.58, 3.36)*	0.00*	
	Large	59(43.4)	77(56.6)	4.00(2.54, 6.30)*	0.00*	
8	Educational status of mothers/care givers					
	No formal education	82(44.1)	104(55.9)	3.25(1.42, 7.42)*	0.05*	
	Primary education (1-8)	77(14.8)	445(85.2)	0.71(0.32, 1.60)	0.41	
	Secondary education (9-12)	20(21.3)	74(78.7)	1.11(0.45, 2.79)	0.82	
	Certificate and above**	8(19.5)	33(80.5)	1		
9	Age category of mothers/care givers					
	16-24 years	46(23.8)	147(76.2)	1		
	25-35 years	104(20.6)	401(79.4)	0.87(0.59, 1.21)	0.50	
	>=36 years	37(25.5)	108(74.5)	1.31(0.79, 2.16)	0.30	

*Statistical significant at p ≤0.25

5.3.8.2.1.2 Socio-demographic characteristics of child and association with the occurrence of pneumonia among under-5 year's old children

The result of bivariate analysis showed that being male child (COR=1.23, 95% CI: 0.89, 1.70, p=0.22), children with birth interval of less than two years (COR=1.99, 95% CI: 1.38, 2.87, p<0.001) and being a child between

the age of 7-12 months and 13-37 months (COR=2.11, 95% CI: 1.24, 3.58; p=0.006 and COR: 1.84; 95% CI: 1.22, 2.78; p=0.004) were found to be the contributing factors for the occurrence of childhood pneumonia in Sidama Regional State. But birth order of child, number of children mothers had and place of birth did not show any significant difference for the occurrence of pneumonia.

Table 5. 10: Socio-demographic characteristics of child and the occurrence of pneumonia among under five children in Si	idama Regional
State: 2020	

No	Variables	Pneumonia status		COR of 95% CI	p-value
		Yes (n %)	No (n %)		
1	Age of child				
	≤ 6 months	8(15.1)	45(84.9)	0.97(0.42, 2.23)	0.94
	7-12 months	34(27.9)	88(72.1)	2.11(1.24, 3.58)	0.006*
	13-36 months	108(25.2)	321(74.8)	1.84(1.22, 2.78)	0.004*
	37-59 months	37(15.5)	202(84.5)	1	
2	Place of birth				
	Health institutions	137(21.9)	489(78.1)	1	
	Home	50(23.0)	167(77.0)	1.07(.74, 1.56)	0.72
3	Sex of Child				
	Male	101(23.9)	321(76.1)	1.23(0.89, 1.70)*	0.22*
	Female	86(23.4)	335(79.6)	1	
	Number of children the mothers had				
4	One	45(23.1)	150(76.9)	1.16(0.68, 2.01)	0.58
	Two	48(22.3)	167(77.7)	1.12(0.65, 1.91)	0.69
	Three	45(25.1)	134(74.9)	1.31(0.75, 2.26)	0.34
	Four	23(18.1)	104(81.9)	0.86(0.46, 1.60)	0.63
	Five and above	26(20.5)	101(79.5)	1	
	Birth interval with in the older child				
5	< 2 years	58(32.4)	121(67.6)	1.99(1.38, 2.87)*	0.00*
	≥2 years	129(19.4)	535(80.6)	1	
	Birth order of the child				
6	First	45(23.0)	151(77.0)	1	
	Second	48(22.4)	166(77.6)	0.97(0.61, 1.54)	0.90
	Third	45(25.1)	134(74.9)	1.13(0.70, 1.81)	0.62
	Fourth and above	49(19.3)	1205(80.7)	0.80(0.51, 1.27)	0.34

*Statistical significant at p ≤0.25

5.3.8.2.2 Bivariate analysis of behavioural and disease history of household members and occurrence of pneumonia among under-five year's children

Findings of bivariate analysis showed that children from households where family member had smoking history (COR=0.41, 95%CI: 0.09, 1.78, p= 0.23) and also history of respiratory diseases (COR=11.38, 95%CI: 5.41, 23.91, p=0.00) were found to more likely suffer from pneumonia than those who were not from such family.

Children who had history of disease during data collection period (COR=1.91, 95% CI: 1.28, 2.86, p=0.00) and those children who had past history of measles within last one year during data collection (COR=8.70, 95% CI: 1.04, 72.77) were more likely to suffer from pneumonia than those who did not have same history. But children who had other chronic disease history were not identified as the factors for pneumonia in current study (see table 5.11) below.

Table 5. 11: Behavioural and disease history and the occurrence of pneumonia among under-five year's children in Sidama Regional S	state:
2020	

No	Variables	Pneumonia status		COR of 95% CI	p-value	
		Yes (n %)	No (n %)			
1	A smoker among family member					
	Yes	2(10.5)	17(89.5)	0.41(0.09, 1.78)*	0.23*	
	No	185(22.5)	639(77.5)	1		

2	The presence of a family member living with HIV				
	Yes	1(25.0)	3(75.0)	1.06(0.09, 12.14)	0.97
	No	180(22.1)	634(77.9)	0.90(0.35, 2.29)	0.82
	Unknown status	6(24.0)	19(76.0)	1	
3	A family members had history of respiratory disease				
	Yes	28(73.7)	10(26.3)	11.38(5.41,23.91)*	0.00*
	No	159(19.8)	646(80.2)	1	
4	Had history of current disease during data collection				
	Yes	44 (32.6)	91 (67.4)	1.91 (1.28, 2.86)*	0.00*
	No	143(20.2)	565 (79.8)	1	
5	Child had known disease historical				
	Yes	77(23.5)	251(76.5)	1.07(0.22, 5.28)	0.93
	No	108(21.3)	398(78.7)	0.95(0.19, 4.64)	0.95
	Not known	2(22.2)	7(77.8)	1	
6	Have child had measles disease last year				
	Yes	67(55.4)	54(44.6)	8.70(1.04, 72.77)*	0.05*
	No	119(16.7)	595(83.3)	1	

*Statistical significant at p ≤0.25

5.3.8.2.3 Bivariate analysis of environmental characteristics of households and the occurrence of pneumonia among under-five year's old children

According to the result of bivariate analysis households without latrine (COR=5.81, 95% CI: 3.77, 8.96, p=0.00) were more likely to contribute for the occurrence of childhood pneumonia. Furthermore the absence of a separate house for animals and humans (COR=1.73, 95% CI: 1.25, 2.41, p<0.001) and houses without windows (COR=3.68, 95% CI: 2.62, 5.17, p<0.001) were more likely to contribute to the occurrence of childhood pneumonia.

The other environmental variables that influenced the occurrence of pneumonia among under-five children were household with only one room (COR=2.02, 95% CI: 0.84, 4.83, p=0.12), families using charcoal as main source of energy (COR=3.69, 95% CI: 2.63, 5.18, p<0.001) and the occurrence of chimney during the period of data collection (COR=2.12, 95% CI: 1.52, 2.95, p<0.001) respectively.

The other environmental variables like the usual place of cooking and solid waste management were found to be the protective factors of the childhood pneumonia while roofing materials did not show any significance contribution for the occurrence of pneumonia in the current study area.

Table 5. 12: Bivariate analysis of environmental characteristics of households in Sidama Regional State: 2020

No	Variables	Pneumonia sta	atus	COR of 95% CI	P- value	
		Yes (n %)	No (n %)			
1	Households had latrine					
	Yes	130(17.6)	610(82.4)	1		
	No	57(55.3)	46(44.7)	5.81(3.77, 8.96)*	<0.001*	
2	Solid waste management system of the HHs					
	Backyard simple	116(22.1)	409(77.9)	0.80(0.50, 1.24)	0.31	
	Blow a fuse at anywhere	39(19.8)	158(80.2)	0.69(0.40, 1.47)	0.17*	
	Bury	32(26.4)	89(73.6)	1		
3	Separate living house for animals and humans					
	Yes	76(17.6)	356(82.4)	1		
	No	111(27.0)	300(73.0)	1.73(1.25, 2.41)*	< 0.001*	
4	Roofing material of living house					
	Grass	112(27.7)	292(72.3)	1.34(0.27,6.56)	0.72	
	Corrugated iron	73(17.0)	357(83.0)	0.72(0.15, 3.51)	0.68	

	Bamboo	2(22.2)	7(77.8)	1	
	Presence of window in the living house				
5	Yes	69(13.3)	448(86.7)	1	
	No	118(36.2)	208(63.8)	3.68(2.62, 5.17)*	< 0.001*
	Number of rooms of the living house				
6	One	52(28.7)	129(71.3)	2.02(0.84, 4.83)*	0.12*
	Two	83(21.6)	301(78.4)	1.38(0.59, 3.22)	0.46
	Three	45(19.1)	191(80.9)	1.18(0.49, 2.82)	0.71
	Four and above	7(16.7)	350(83.3)	1	
	Usually place for cooking				
7	Main house	63(18.7)	273(81.3)	0.36(0.14, 0.97)	0.04*
	Kitchen	117(23.9)	372(76.1)	0.16(0.19, 1.30)	0.16*
	Outdoor	7(38.9)	11(61.1)	1	
8	Use of charcoal				
	Yes	102(38.8)	161(61.2)	3.69(2.63, 5.18)*	< 0.001*
	No	85(14.7)	495(85.3)	1	
9	Presence of chimney				
	Yes	98(30.4)	224(69.6)	2.12(1.52, 2.95)*	0.00*
	No	89(17.1)	432(82.9)	1	

*Statistical significant at p ≤0.25

5.3.8.2.4 Bivariate analysis of nutritional and immunization characteristics of child and the occurrence of pneumonia among under-five year's old children

Table 5.13 presents the result of bivariate analysis of nutritional and immunization status of child and the association with the occurrence of pneumonia among under-five children. The result showed that severely malnourished children (COR=2.33, 95%CI; 1.43, 3.81, p=0.001) and initiation of breastfeeding after a day of delivery (COR=2.82, 95%CI; 1.00, 7.97, p=0.03), were found to be contributing factors of childhood pneumonia occurrence in Sidama Regional State. Similarly lack of receiving vitamin A and deworming within last six months preceding to

data collection, and not receiving any dose of vaccine and measles were also obtained as significant contributing factors of childhood pneumonia in study area with the [(COR=5.81, 95%CI: 4.06, 8.31, p<0.001); (COR=1.50, 95%CI: 0.98, 2.31, p=0.06); (COR=8.74, 95%CI: 5.29, 14.43, p<0.001) & (COR=4.30, 95%CI: 2.76, 6.70, p<0.001)], correspondingly.

However, type of materials used to feed the child were found to be the protective factors while receiving PCV10 vaccine did not show any significant association with the occurrence of pneumonia among children in the study area (see table 5.13).

No	Variables	Pneumonia statu	S	COR of 95% CI	P-value	
		Yes (n %)	No (n %)			
1	Nutritional status of children (weight for age)					
	Normal	130(20.5)	505(79.5)	1		
	Under weight	27(21.1)	101(78.9)	1.04(0.65, 1.66)	0.87	
	Severe under weight	30(37.5)	50(62.5)	2.33(1.43, 3.81)	0.001*	
2	Initiation of breast feeding					
	Immediately at birth	124(21.5)	454(78.5)	1		
	Within one hour after birth	37(23.3)	122(76.7)	1.11(0.69, 1.76)	0.68	
	Between first 1 and 24 hour	17(20.0)	68(80.0)	0.86(0.46, 1.62)	0.65	
	After a day	9(42.9)	12(57.1)	2.82(1.00, 7.97)	0.03*	
3	Materials used to fed the child					
	Bottle	51(37.5)	85(62.5)	1		
	Cup	25(26.0)	71(74.0)	0.59(0.33, 1.04)	< 0.001*	
	Spoon	66(20.6)	254(79.4)	0.43(0.28, 0.67)	< 0.001*	
	Hand	31(14.6)	181(85.4)	0.29(0.17, 0.49)	< 0.001*	
	Not eligible	14(17.7)	65(82.3)	0.36(0.18, 0.70)	0.003*	
4	Received vitamin A last 6 month					
	Yes	68(12.5)	477(87.5)	1		
	No	111(45.3)	134(54.7)	5.81(4.06, 8.31)*	< 0.001*	
	Not eligible	8(15.1)	45(84.9)	1.25(0.56, 2.76)	0.59	

Table 5. 13: Bivariate analysis of nutritional and immunization related status of children and the occurrence of pneumonia among underfive year's children in Sidama Regional State: 2020

5	Dewormed with in last 6 months				
	Yes	71(19.5)	294(80.5)	1	
	No	45(26.6)	124(73.4)	1.50(0.98, 2.31)	0.06*
	Not eligible	71(23.0)	238(77.0)	1.23(0.85, 1.79)	0.26
6	Any vaccine and dose given				
	Yes	136(17.8)	629(82.2)	1	
	No	51(65.4)	27(34.6)	8.74(5.29, 14.43)*	< 0.001*
7	Received measles vaccine				
	Yes	122(18.6)	535(81.4)	1	
	No	48(49.5)	49(50.5)	4.30(2.76, 6.70)*	< 0.001*
	Not eligible	17(19.1)	72(80.9)	1.04(0.59, 1.82)	0.90
8	PCV10 given				
	Yes	173(21.6)	628(78.4)	1	
	No	2(40.0)	3(60.0)	1.34(0.47, 3.80)	0.59
	Not eligible	12(31.6)	25(68.4)	0.84(0.48, 1.49)	0.56

*Statistical significant at $p \le 0.25$

5.3.8.4 Multivariate analysis of factors and association with the occurrence of pneumonia among under-five year's old children

The multivariate analysis were employed by using the binary logistic regression with conditional backward stepwise method. Likwise Hosmer and Lemeshow Test was performed in order to recognise the risk factors that are self-sufficiently associated with childhood pneumonia though controlling for confounders. The same controlling variables for binary logistic regression were kept for multivariate analysis also. Independent variables those were significantly associated with childhood pneumonia at a p-value less than 0.25 in the bivariate analysis while using Chi-square tests were included in the multivariable analysis. The association of risk factors with the occurrence of pneumonia were recognised based on p-value less than 0.05 and considering 95% confidence interval also.

5.3.8.4.1 Multivariate analysis of socio-demographic and economic factors and the occurrence of pneumonia among under-five children

From binary logistic regression there were five different sociodemographic and economic factors found to influence the occurrence of childhood pneumonia with p-value less than 0.25. Those potential risk factors were checked under multivariate analysis for proper identification as risk factors of pneumonia with controlling the likely confounders. From the backward stepwise method used for the analysis, the Omnibus Tests of Method coefficients analysed for the socio-demographic factors and childhood pneumonia occurrence, to check that the model significantly elucidated the relationship (see table 5.15).

Table 5. 14: Omnibus Test of model coefficients for the socio-demographic and economic variables and the occurrence of childhood pneumonia, in Sidama Regional State, Ethiopia: 2020

Block 1: Method = Backward Stepwise (Conditional)

		Chi-square	df	Sig.
Step 1	Step	125.269	10	.000
	Block	125.269	10	.000
	Model	125.269	10	.000
Step 2 ^a	Step	-1.122	1	.290
	Block	124.147	9	.000
	Model	124.147	9	.000

Omnibus Tests of Model Coefficients

 A negative Chi-squares value indicates that the Chi-squares value has decreased from the previous step.

As the result from multivariate analysis indicate: that family size, educational level of respondents, age of child and birth interval of sample child with its older child were found to be associated factors for the occurrence of pneumonia among under-five children.

Both medium and large family size were significantly associated with the occurrence of childhood pneumonia with p-value less than 0.001 for both

Auctores Publishing LLC – Volume 8(1)-255 www.auctoresonline.org ISSN: 2637-8892 categories. Children from large family size were 4.34 times and children from medium family size were 2.39 times more possible to suffer from pneumonia with (AOR: 4.34; 95%CI 2.65, 7.10; p<0.001 and AOR: 2.39; 95%CI; 1.60, 3.59; p<0.001) respectively.

The result of multivariate analysis revealed that mothers or caregivers who never had formal education were found to be one of the causal factors for the occurrence of pneumonia among under five children. Children from mothers or caregivers who did not have formal education were 4.10 times more likely to develop pneumonia as compared to their counterparts of educated mothers/caregivers (AOR: 4.10; 95%CI; 1.65, 10.00; p=0.002).

Birth interval of the surveyed child with its older child was also significantly associated with the occurrence of pneumonia in children, p=0.01. Detailed analysis showed that children who had birth intervals of less than 2 years were 1.65 times more likely to develop pneumonia than those that had birth intervals of more than 2 years with their respective older siblings (AOR:1.65: 95% CI: 1.10, 2.48; p=0.02).

There was a momentous association among the age category of children and pneumonia occurrence. The finding of multivariate analysis revealed that the middle age group (7-36 months) children were affected more than children in the rest age group. Children from age group 7-12 months were 2.63 times affected and those 13-36 months were 2.20 times more likely to develop pneumonia than the others, with (AOR: 2.63; 95%CI: 1.46, 4.74; p=0.001 and AOR:2.20; 95%CI: 1.40, 3.46; p=0.001) respectively.

However, the sex of the child and their residence did not show any significant association for the occurrence of pneumonia in the current study area as the result of multivariate analysis revealed (see table 5.15).

 Table 5. 15: Multivariate analysis of household's socio-demographic and economic factors and occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020

No	Variables	Variables Pneumonia status		COR of 95% CI	AOR of 95% CI	P-value
		Yes (n %)	No (n %)			
	Family size					
1	Small	57(12.5)	399(87.5)	1	1	
	Medium	81(26.8)	221(73.2)	2.30(1.58, 3.36)*	2.39(1.60, 3.59)	< 0.001**
	Large	49(38.9)	77(61.1)	4.00(2.54, 6.30)*	4.34(2.65, 7.10)	< 0.001**
2	Educational status of respondents					
	No formal education	82(44.1)	104(55.9)	3.25(1.42, 7.42)*	4.10(1.65, 10.00)	0.002*
	Primary education (1-8)	77(14.8)	445(85.2)	0.71(0.32, 1.60)	0.89(0.37, 2.18)	0.81
	Secondary education (9-12)	20(21.3)	74(78.7)	1.11(0.45, 2.79)	1.33(0.50, 3.56)	0.57
	Certificate and above	8(19.5)	33(80.5)	1	1	
	Age of child					
3	≤ 6 months	8(15.1)	45(84.9)	0.97(0.42, 2.23)	1.15(0.47, 2.80)	0.76
	7-12 months	34(27.9)	88(72.1)	2.11(1.24, 3.58)*	2.63(1.46, 4.74)	0.001*
	13-36 months	108(25.1)	321(74.8)	1.84(1.22, 2.78)*	2.20(1.40, 3.46)	0.001*
	37-59 months	37(15.5)	202(84.5)	1	1	
	Sex of Child					
4	Male	101(23.9)	321(76.1)	1.23(0.89, 1.70)*	1.23(0.84, 1.73)	0.31
	Female	86(23.4)	335(79.6)	1	1	
5	Birth interval within older					
	< 2 years	58(32.4)	121(67.6)	1.99(1.38, 2.87)*	1.65(1.10, 2.48)	0.02*
	≥ 2 years	129(19.4)	535(80.6)	1	1	
	Residence of respondents					
	Urban	20(33.3)	40 (66.7)	1	1	
	Rural	167(21.3)	616 (78.7)	0.54 (0.30, 0.95)	0.75(0.38, 1.45)	0.39

1 indicates reference category, * indicates, Significant association at p <= 0.05 and ** indicates highly significant association at p < 0.001

5.3.8.4.2 Multivariate analysis of behavioural and disease history of households and the occurrence of pneumonia among under-five children

Four different behavioural and disease related variables those presented a significant association in binary logistic regression were performed in multivariate analysis to check the association with childhood pneumonia occurrence. From these variables like children who had family member/s with the history of respiratory disease and children had suffered from

measles disease in the previous one year prior to data collection showed high significant association (p<0.001), while history of any disease among children during data collection showed the significant association with (p=0.03).

The value of chi-square with degree of freedom and p-value of Omnibus Test of model coefficients were checked to identify the association (see table 5.16).

Table 5. 16: Omnibus Test of model coefficients for the behavioral and disease history of households' family member and the occurrence of childhood pneumonia, in Sidama Regional State, Ethiopia: 2020

Omnibus Tests of Model Coefficients

		Chi-square	df	Sig.
Step 1	Step	131.183	5	.000
	Block	131.183	5	.000
	Model	131.183	5	.000

Similarly the p-value and X^2 with degree of freedom of Hosmer and Lemeshow Test was also assured. See table 5.17:

 Table 5. 17: Hosmer and Lemeshow Test for the socio-demographic variables and the occurrence of childhood pneumonia, in Sidama Regional State, Ethiopia: 2020

Hosmer and Lemeshow Test

Step	Chi-square	df	Sig.
1	7.856	3	.049

History of respiratory disease among family member was found as risk factor for the occurrence of childhood pneumonia with p- value less than 0.001. Children who had family member/s with the history of respiratory disease were 5.63 times more likely to suffer from pneumonia than their counterparts who lived with family members without the history of respiratory disease (AOR: 5.63; 95%CI 2.74-11.57; p<0.001).

The history of any disease among children during data collection was found to significantly influence the occurrence of pneumonia p=0.03. Children with any of disease history during data collection were 1.63

times more likely to develop pneumonia (AOR: 1.63; 95%CI: 1.04, 2.53; p=0.03). Detailed analysis revealed that children had suffered from measles disease in the previous one year prior to data collection were 5.96 more likely to develop pneumonia with (AOR: 5.96; 95%CI: 3.91, 9.07; p<0.001).

However, having the history of HIV virus among family members did not show significant association after controlling the confounders (see table 5.18).

No	Variables	Pneumonia status		COR of 95% CI	AOR of 95%CI	P- value
		Yes (n %)	No (n %)	-		
1	A smoker among family member					
	Yes	2(10.5)	17(89.5)	0.41(0.09, 1.78)*	0.32(0.07, 1.47)	0.14
	No	185(22.5)	639(77.5)	1	1	
2	A family members had history of respiratory disease					
	Yes	22(57.9)	16(42.1)		5.63(2.74, 11.57)**	<0.001**
	No	165(20.5)	640(79.5)	1	1	
3	Had history of current disease during data collection					
	Yes	44 (32.6)	91 (67.4)	1.91 (1.28, 2.86)*	1.63(1.04, 2.53)	0.03*
	No	143(20.2)	565 (79.8)	1		
4	Have child had measles disease last year					
	Yes	67(55.4)	54(44.6)	6.20(1.12, 9.34)*	5.96(3.91, 9.07)	<0.001**
	Unknown	1(12.5)	7(87.5)	0.71(0.09, 5.86)	0.41(0.04, 3.78)	0.43
	No	119(16.7)	595(83.3)	1	1	

Table 5. 18: Multivariate analysis of behavioural and disease history of households and the occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843)

1 indicates reference category, * indicates, Significant association at $p \le 0.05$ and ** indicates highly significant association at p < 0.001

5.3.8.4.3 Multivariate analysis of environmental factors of households and the occurrence of pneumonia among under-five children

Table 5.19 below presents environmental factors that influence occurrence of childhood pneumonia at the study area at multivariate analysis. Five factors namely presence or absence of latrine in the household, living in the same household with domestic animals, the use of charcoal as the main source of energy, presence of chimney during the time of data collection and living in house without window were found to influence the occurrence of pneumonia.

There significant association were found between the absence of household latrine and the occurrence of childhood pneumonia at the study area. Children from households that did not avail latrine were 4.32 times more likely to develop pneumonia than those who were accessed for latrine (AOR: 4.32; 95% CI: 2.61, 7.15; p<0.001).

Children whose family did not have separate house for the animal and human being were 1.51 times more likely to develop pneumonia than those whose family had separate shelter for animals and human beings (AOR: 1.98; 95% CI: 1.30, 3.01; p=0.001). Also, children living in household that used charcoal as the main source of energy were 3.55 times more likely to suffer from pneumonia compared to their counterparts living in households who did not use charcoal (AOR=3.55; 95% CI 2.36, 5.35; p<0.001).

Analysis also revealed that having house with windows was found as the protective factor from pneumonia occurrence; as children who lived in houses without windows were found to be 4.60 times more likely to develop pneumonia than those who lived in houses with windows (AOR: 4.60; 95% CI: 2.93, 7.15; p<0.001). Similarly the presence of chimney inside the houses during the time of data collection was found to be 1.91 times more likely to contribute the occurrence of pneumonia than the counterpart (AOR: 1.91; 95% CI: 1.25, 2.91; p=0.003). (See table 5.19).

Table 5. 19: Multivariate analysis of environmental factors and the occurrent	nce of pneumonia among under-five children in Sidama Regional
State, Ethiopia: 2020 (N=843)	

No	Variables	Pneumonia status		COR of 95% CI	AOR of 95%CI	p- value
		Yes (n %)	No (n %)			
1	Households had latrine					
	Yes	130(17.6)	610(82.4)	1	1	
	No	57(55.3)	46(44.7)	5.81(3.77, 8.96)*	4.32(2.61, 7.15)	< 0.001**
2	Separate living house for animal and human being					
	Yes	76(17.6)	356(82.4)	1	1	
	No	111(27.0)	300(73.0)	1.73(1.25, 2.41)*	1.98(1.30, 3.01)	0.001*
3	Solid waste management system of the HHs					
	Backyard simple	116(22.1)	409(77.9)	0.80(0.50, 1.24)	1.91(1.25, 2.91)	0.90
	Blow a fuse at anywhere	39(19.8)	158(80.2)	0.69(0.40, 1.47)	0.76(0.40, 1.48)	0.42
	Bury	32(26.4)	89(73.6)	1	1	
4	Presence of window in the house					
	Yes	69(13.3)	448(86.7)	1	1	
	No	118(36.2)	208(63.8)	3.68(2.62, 5.17)*	4.60(2.93, 7.15)	< 0.001**
5	No of rooms of the living house					
	One	52(28.7)	129(71.3)	2.02(0.84, 4.83)*	2.20(0.80, 6.04)	0.13
	Two	83(21.6)	301(78.4)	1.38(0.59, 3.22)	1.20(0.46, 3.13)	0.71
	Three	45(19.1)	191(80.9)	1.18(0.49, 2.82)	1.25(0.47, 3.36)	0.66
	Four and above	7(16.7)	350(83.3)	1	1	
6	Usually place for cooking					
	Main house	63(18.7)	273(81.3)	0.36(0.14, 0.97)	0.39(0.12, 1.22)	0.11
	Kitchen	117(23.9)	372(76.1)	0.16(0.19, 1.30)	1.58(0.51, 4.89)	0.42
	Outdoor	7(38.9)	11(61.1)	1	1	
	Use of charcoal					
7	Yes	102(38.8)	161(61.2)	3.69(2.63, 5.18)*	3.55(2.36, 5.35)	< 0.001**
	No	85(14.7)	495(85.3)	1	1	
8	Presence of chimney					
	Yes	98(30.4)	224(69.6)	2.12(1.52, 2.95)*	1.91(1.25, 2.91)	0.003*
	No	89(17.1)	432(82.9)	1	1	

1 indicates reference category, * indicates, Significant association at p<=0.05 and ** indicates highly significant association at p< 0.001

5.3.8.4.4 Multivariate analysis of nutritional and immunization related factors and the occurrence of pneumonia among under-five children

The result of the multivariate analysis revealed that being severely underweight child for its age, a child who did not take vitamin A in the last six month during data collection, a child who did not take any dose of vaccine for their age and children who did not receive measles vaccines were found to be the contributing factors for the occurrence of pneumonia in the study area.

Detailed analysis revealed that children who were severely underweight for their age were 2.43 times more likely to develop pneumonia than those with proportional weighed children (AOR: 2.43, 95%CI; 1.35, 4.37 p=0.003).

Children who did not receive vitamin A were 5.27 times more likely to develop pneumonia than those who received (AOR: 5.27; 95% CI: 3.41, 8.15; p<0.001). Similarly, children who did not receive any vaccine (AOR: 8.03; 95% CI: 4.24, 15.22; p<0.001) and specifically who did not receive measles vaccine (AOR: 2.27; 95% CI: 1.34, 3.86; p=0.002); were found to more likely develop pneumonia 8.03 times and 2.27 times respectively than those children who had received vaccines for their age.

However using hand to fed the children was less likely to develop childhood pneumonia (AOR=0.22, 95%CI: 0.09, 0.55; p<0.001) as shown in table 5.20.

Table 5. 20: Multivariate analysis of nutritional and immunization related factors and the occurrence of pneumonia among under-five children in Sidama Regional State, Ethiopia: 2020 (N=843)

No	Variables	Pneumonia status		COR of 95% CI	AOR of 95% CI	P-value
		Yes (n %)	No (n %)	_		
1	Nutritional status of children (weight for age)					
	Normal	130(20.5)	505(79.5)	1	1	
	Under weight	27(21.1)	101(78.9)	1.04(0.65, 1.66)	1.06(0.62, 1.83)	0.82
	Severe under weight	30(37.5)	50(62.5)	2.33(1.43, 3.81)*	2.43(1.35, 4.37)	< 0.003*
2	Initiation of breast feeding					
	Immediately at birth	124(21.5)	454(78.5)	1	1	
	Within one hour after birth	37(23.3)	122(76.7)	1.11(0.69, 1.76)	1.30(0.82, 2.11)	0.29
	Between first 1 and 24 hour	17(20.0)	68(80.0)	0.86(0.46, 1.62)	0.90(0.46, 1.75)	0.75
	After a day	9(42.9)	12(57.1)	2.82(1.00, 7.97)	2.35(0.82, 6.71)	0.11
3	Received vitamin A last 6 month					
	Yes	68(12.5)	477(87.5)	1	1	
	No	111(45.3)	134(54.7)	5.81(4.06, 8.31)*	5.27(3.41, 8.15)	< 0.001**
	Not eligible	8(15.1)	45(84.9)	1.25(0.56, 2.76)	1.30(0.40, 4.22)	0.67
4	Dewormed with in last 6 months					
	Yes	71(19.5)	294(80.5)	1	1	
	No	45(26.6)	124(73.4)	1.50(0.98, 2.31)	0.67(0.39, 1.15)	0.15
	Not eligible	71(23.0)	238(77.0)	1.23(0.85, 1.79)	0.74(0.45, 1.21)	0.23
5	Any vaccine and dose given					
	Yes	136(17.8)	629(82.2)	1	1	
	No	51(65.4)	27(34.6)	8.74(5.29, 14.43)*	8.03(4.24, 15.22)	<0.001**
6	Received measles vaccine					
	Yes	122(18.6)	535(81.4)	1	1	
	No	48(49.5)	49(50.5)	4.30(2.76, 6.70)*	2.27(1.34, 3.86)	0.002*
	Not eligible	17(19.1)	72(80.9)	1.04(0.59, 1.82)	0.70(0.30, 1.64)	0.41
7	Materials used to fed the child					
	Bottle	51(37.5)	85(62.5)	1		
	Cup	25(26.0)	71(74.0)	0.59(0.33, 1.04)*	0.52(0.27, 1.02)	0.06
	Spoon	66(20.6)	254(79.4)	0.43(0.28, 0.67)*	0.61(0.37, 1.02)	0.06
	Hand	31(14.6)	181(85.4)	0.29(0.17, 0.49)*	0.28(0.15, 0.55)	< 0.001
	Not eligible	14(17.7)	65(82.3)	$0.36(0.18, 0.70)^*$	0.22(0.09, 0.55)	

1 indicates reference category, * indicates, Significant association at p <= 0.05 and ** indicates highly significant association at p < 0.001

5.3.9 Knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control and associated factors with integration of qualitative finding

Knowledge, attitude and practice of mothers/ caregivers were assessed based on prepared questionnaires and analysed separately for each of the three variables for quantitative part of this study. The qualitative part of the data were analysed manually with the support of Atlas ti software. A total of 40 mothers with under five children were participated in 6 different FGDs. Their age ranged from 24 to 42 years, with mean age 28.7 and the mode 30. The detail biblograph of FGDs preseted with the following table.

Table 5.21A: Biblograph of FGD participants in Sidama regional state, Ethiopia 2020

List of FGDs	Age of participants (P)	Their district	Kebele	Remark
FGD 1	P1=26, P2 36, P3 40, P4 34, P5 35, P6 30 & P7 31	Dale	Soyama	
FGD2	P1 38, P2 26, P3 28, P4 26, P5 31 P6 26 & P7 33	Dale	Gane	
FGD3	P1 32, P2 40, P3 29, P4 37, P5 30, P6 38 & P7 27	Aleta wondo town	Mesalemia	
FGD4	P1 40, P2 32, P3 39, P4 28, P5 34 & P6 30	Aleta wondo town	Mesalemia cluster 2	
FGD5	P1 24, P2 41, P3 42, P4 38, P5 27, P6, 32 & P7 30	Loka Abaya	Arada gale	
FGD6	P1 40, P2 25, P3 37, P4 35, P5 29 & P6 30	Loka Abaya	Chalbesa	

FGD: Focus group discussion, P: Participants

Three themes and six sub-themes were obtained from the analysis of qualitative data that was aligned with the proposed objectives (See table 21B below).

Table 5.21B: Themes and sub-themes identified from qualitative data analysis in Sidama regional state, Ethiopia 2020

Themes	Sub-themes
1. Knowledge related concerns of mothers	1.1 General opinion and knowledge of mothers towards causes of childhood
	pneumonia
	1.2 knowledge of mothers related with transimission of childhood pneumonia
	1.3 knowledge of mothers towards prevention of childhood pneumonia
2. Attitude related opinions of mothers	2.1 Attitude towards severity and dangereneous of childhood pneumonia
	2.2 Attitude towards prevention and control mechanisms of childhood
	pneumonia
3. Practice related issues	3.1 Weather mothers practicing or not those childhood pneumonia prevention
	and control mechanism

The results of each sub-themes were used to support the findings of quantitative part of study, as discussed below.

5.3.9.1 Heard about pneumonia and source of information

The knowledge of mothers or caregivers related to identification of sign and symptoms, causes, mode of transmission and prevention of pneumonia was assessed. Results of descriptive analysis in this regard revealed that 634(75.2%) of mother/caregivers had heard of childhood pneumonia while about quarter 209(24.8%) of them did not know. Of those who had heard, almost half of respondents got the information from health extension workers and 146(17.3%) from older people or community leaders while 25(3%) and 13(1.5%) directly from health workers and TV/Radio respectively.

Table 5. 21C: Knowledge and source of information of mothers/caregivers about	t childhood pneumonia in Sidama Regional State, Ethiopia
2020	

No	Characteristics		Frequency (n)	Percent (%)
	Heard childhood pneumonia (N=843)	Yes	634	75.2
1		No	209	24.8
		Total	843	100
		Health extension workers	450	53.4
2	Source of information (N=634)	Health workers	25	3.0
		TV/Radio	13	1.5
		Older people/community leaders	146	17.3
		Total	834	100

The results from focus group discussions analysis also confirmed that most participants agreed that they had heard about pneumonia as common childhood disease and got the information from health extension workers and health professionals when they took their children to the health facilities for postnatal consultation.

A 26-year-old mother from Soyama *kebele* said that "Health professionals have informed me that my child had get pneumonia and I have to properly handle him. They also gave me a syrup to treat... it" (FGD1 participant).

5.3.9.2 Descriptive analysis of knowledge of mothers/caregivers on sign and symptom of pneumonia with integration of qualitative finding

Nine different questions about signs and symptoms of childhood pneumonia were asked and tried to assess knowledge of mothers/care givers. Majority of them 497(59.0%) knew that fever is one of the signs and symptoms of childhood pneumonia. Almost half of the respondents 412(48.9%), 443(52.6%), 444(52.6%), 432(51.2%) and 443(52.6%)

responded that they are knowledgeable on chills, chest in-drawing, fast breathing and stridor for being signs and symptoms of pneumonia respectively.

Most of the FGD discussants mentioned that pneumonia may cause fast breathing, chills and sign of stridor among under-five children while they get pneumonia.

One 26-year-old mother from Gane *kebele* said "My child had fast breathing and sign of stridor [*Afale udulinosi, sheengarsannosi*], while he was diseased with pneumonia..." (FGD2 participant).

Another 32-year-old mother from A/wondo town reflected "Childhood pneumonia caused shortness of breathing and that my child became very weak with difficulty of breathing..." (FGD3 participant).

As the result revealed two third of mothers/caregivers 622(73.8%); knew cough as sign and symptom of pneumonia while 374(44.4%) and

323(38.2%) of them knew that weakness and loss of appetite as signs and symptoms of pneumonia respectively.

Majority of FGD participants argued that cough is typical sign and symptom of pneumonia among under-five children.

A 40-year-old woman from Aleta wondo town showed her feeling saying, "Cough is the most typical sign and symptom of pneumonia that children can show when they had diseased with it. Once up on time my child got pneumonia and was challenged with cough critically [*mudde mudde buusiisinosi*], till I took him to health facility..." (FGD4 participant.

Another 24-year-old mother from Arada Gale *kebele* believed "Children became weak [*dalaganno*] and unable to eat and drink when they got pneumonia..." (FGD5 participant).

However, except cough, almost half of the respondents did know that the other signs and symptoms of pneumonia properly as shown in table 5.22.

Fable	5.2	22:1	Knowle	edge o	f mot	hers/	careg	ivers	on sig	n and	svm	ptom (of r	oneumonia	in	Sidama	Re	gional	State.	Ethio	oia: 20	20
I GOIC	~ -						carch		U II 01,		., .	prom.		meanonia		Juantin		5101141	Dunce	Louis	Jun 1 0	

No	Variables for assessing knowledge of respondents on sign and symptom of pneumonia	Right answer		Wrong answer	
		Frequency (n)	(%)	Frequency (n)	(%)
1	Knew that fever as sign and symptom of pneumonia	497	59.0	346	41.0
2	Knew that chills as sign and symptom of pneumonia	412	48.9	431	51.1
3	Knew that cough as sign and symptom of pneumonia	622	73.8	221	26.2
4	Knew that chest in drawing as sign and symptom of pneumonia	443	52.6	400	47.4
5	Knew that difficulty in breathing as sign and symptom of pneumonia	444	52.7	399	47.3
6	Knew that Stridor as sign and symptom of pneumonia	443	52.6	400	47.4
7	Knew that fast breathing as sign and symptom of pneumonia	432	51.2	411	48.8
8	Knew that weakness as sign and symptom of pneumonia	374	44.4	469	55.6
9	Knew that appetite loss as sign and symptom of pneumonia	323	38.3	520	61.7

5.3.9.3 Knowledge of mothers/caregivers on causes of pneumonia with integration of qualitative findings

Table 5.23 presents the knowledge of mothers/caregivers on causes of childhood pneumonia. Three hundred eighty-eight of mothers/caregivers (46.0%) knew that bacteria are a cause of childhood pneumonia; while, 357(42.3%) and 223(26.5%) of them knew that dust or germs and virus respectively might be the causes of pneumonia.

However, from almost six (06) FGDs, none of the discussants showed that pneumonia can be caused by bacteria, virus or germs. Rather they perceived that childhood pneumonia can be caused by poor handling and the sudden clash of cold and hot temperature or due to weather change. This can be inferred from a 40-year-old mother from Chalbesa *kebele* who was quoted for saying: "Childhood pneumonia can be caused by poor handling practices of mothers, starting from the very beginning of birth; that means poor wearing practice, exposing the child to cold weather and poor hygienic practice..." (FGD6 participant).

Another 36-year-old mother from Soyama *kebele* perceived, "The childhood pneumonia can occur within the sudden interaction of hot and cold weather condition. When mothers did not dress their children and

probably those exposing the chest of child to cold weather, the child may get pneumonia..." (FGD1 participant).

Surprisingly, 550(65.2%) and 137(16.3%) of the mothers/caregivers responded that childhood pneumonia might be caused by weather change and witchcraft respectively which was wrongly perceived.

One 38-year-old mother reacted, "Health professionals told me that I did not have to expose my child to cold weather and rather I have to keep doors properly closed. While I did not properly close the door and expose my child to cold weather, he could be diseased with pneumonia. I understood that the cause for pneumonia was interaction of hot and cold temperature..." (FGD2 participant).

But regarding witchcraft as cause of childhood pneumonia none of FGD participants confirmed it confidently.

Two different FGD participants (one 25-year-old and the other 34-year-old mothers) said, "The cause for childhood pneumonia cannot be witchcraft, rather it could be poor handling practice that community showed..." (FGD6 and FGD1 participants respectively).

Table 5. 23: Knowledge of mothers/caregivers on causes of pneumonia in Sidama Regional State, Ethiopia: 2020

	<u> </u>			
No	Variables for assessing knowledge of resp	ondents on causes of 1	Right answer	Wrong answer
	pneumonia			

		Frequency (n)	(%)	Frequency (n)	(%)
1	Knew that dust/germs may cause pneumonia	357	42.3	486	57.7
2	Knew that bacteria may cause pneumonia	388	46.0	455	54.0
3	Knew that virus may cause pneumonia	235	27.9	608	72.1
4	Knew that fungi may cause pneumonia	156	18.5	687	81.5
5	Thought that weather change may cause pneumonia	293	34.8	550	65.2
6	Thought that witchcraft may cause pneumonia	706	83.7	137	16.3

5.3.9.4 Knowledge of mothers/caregivers towards transmission method of pneumonia with integration of qualitative findings

From the total of 843 interviewed mothers/caregivers 544(64.5%) knew that pneumonia can be transmitted through air droplet; while 234(27.8%) responded that pneumonia can be transmitted by personal contact.

The findings from qualitative part of study also revealed that childhood pneumonia can be transmitted through air droplet. A 40-year-old mother said, "Pneumonia can be transmitted via air droplet; hence it is the disease of interaction between cold and hot temperature..." (FGD1 participant). But another discussant aged 41 reflected, "Pneumonia can be transmitted from the older child to the smaller one; if there was the child or children who had experienced with history of being diseased with childhood pneumonia, it can be transmitted during coughing..." (FGD5 participant).

There was also an idea raised from discussants that pneumonia can be transmitted through poor personal hygiene of children which their mothers practiced. One 28-year-old mother said, "Pneumonia can be transmitted when mothers had not washed their child properly including their clothes.

Some mothers did not change the clothes for their children in timely manner and that dirty clothes may aggravate the transmission of childhood pneumonia..." (FGD2 participant).

Some discussants told the researcher that pneumonia can be transmitted by poor sanitation and poor feeding that parents practiced. One 32-yearold mother said that "Open defecation can contribute or the transmission of pneumonia among under-five children. If community defecate in open space, there might be flies those can facilitate the transmission of pneumonia. Additional pneumonia can be transmitted if children drink milk without boiling..." (FGD4 participant).

Table	e 5. 24: H	Knowledge o	of mothers/careg	ivers on transm	ission method o	f pneumonia i	in Sidama reg	gional state,	Ethiop	ia: 2020
									•	

No	Variables for assessing knowledge of respondents on transmission method of pneumonia	Right answer		Wrong answer	
		Frequency (n)	(%)	Frequency (n)	(%)
1	Knew that pneumonia can be transmitted through air droplet	544	64.5	299	35.5
2	Knew that pneumonia can be transmitted through personal contact	234	27.8	609	72.2
3	Knew that pneumonia can be transmitted through dirty hands	229	27.2	614	72.8

5.3.9.5 Knowledge of mothers/caregivers on prevention method of pneumonia with integration of qualitative findings

From the total of respondents only quarter of mothers/caregivers of this study were knowledgeable on most of the prevention methods of pneumonia. However, 573(68.0%) of them were able to identify that pneumonia can be prevented by vaccine.

As the findings obtained from FGDs discussion indicated, some of the discussants witnessed that childhood pneumonia can be prevented via immunization, even if they did not know that there is pneumococcal vaccine, most of them agreed that they have had their children immunized until nine months.

One 35-year-old mother said, "We have to immunize our children properly to prevent them from different diseases including childhood pneumonia. Some mothers may delay immunizing their kids and those children can be challenged with diseases..." (FGD1 participant).

Other discussants argued that breastfeeding can prevent pneumonia. Two different mothers (one 26 and another 31 both from FGD2) said, "Properly breastfeeding of the children may help families to protect their

children from being pneumonia diseased and other diseases. So, we have to breastfeed our children continuously even if they are diseased..."

Majority of discussants from different FGDs raised that pneumonia can be prevented with proper clothing and they argued, "Pneumonia can be prevented with proper clothing of the child [*Du'mu uduunichinni* gagasosi seekkine funge amada hasiisanno/qiidu afirannosiki gede/] since from the time of delivery. If the children are exposed to the cold temperature pneumonia cab be occurred, but if we can wear them properly, we can prevent them from the suffering of the disease. Some mothers may expose their children with cold and hot temperature or vice versa, but this might harm their children. So, mothers should take care of their children to prevent them from pneumonia..." (FGD1, FGD2, FGD4, FGD5 & FGD6 participants).

There were also perceptions like childhood pneumonia can be prevented only through treatment after children getting it. Some of the discussants assumed that pneumonia can be continued and it can be controlled with on time treatment only.

One 42-year mother argued, "Pneumonia may happen suddenly within children and the parents should took them to health facility or purchase

the syrup to treat them. Once my daughter got sick and I took her to health centre, and health professional told me that she had pneumonia and gave

me a syrup for seven days with advising me that I have to administer it properly..." (FGD5 participant).

Tabl	le 5. 25: Knowledge of mothers/caregivers or	prevention methods of childho	ood pneumonia in Sidama re	gional state, Ethiopia: 202	0

No	Variables for assessing knowledge of respondents on prevention method of pneumonia	Right answer		Wrong answer	
		Frequency (n)	Percentage (%)	Frequency (n)	Percentage (%)
1	Knew that pneumonia can be prevented through vaccine	573	68.0	270	32.0
2	Knew that pneumonia can be prevented through proper nutrition	252	29.9	591	70.1
3	Knew that pneumonia can be prevented through Exclusive breast feeding	276	32.7	567	67.3
4	Knew that pneumonia can be prevented through proper clothing within weather conditions	303	35.9	540	64.1
5	Knew that pneumonia can be prevented through proper house ventilation	253	30.0	590	70.0
6	Knew that pneumonia can be prevented through washing hands before feeding child	225	26.7	618	73.3

5.3.9.6 Overall level of knowledge of mothers/caregivers towards childhood pneumonia prevention and control

A total of 25 questions were used to assess knowledge of mothers/caregivers in the current study. These include one (01) question about general information, nine (09) questions about signs and symptoms of pneumonia six (06) questions about causes of pneumonia and three (03) questions were about ways of transmission and the remaining six (06) questions were about prevention methods of pneumonia. Every correct response received a score of 1 and wrong response received a score of 0.

The knowledge score ranged from 0 to 25 and rank as knowledge percentile into three levels namely poor knowledge <50%, moderate/fair knowledge 50-74% and good knowledge >=75%.

The study results revealed that the respondents' average knowledge score was 11.44 with standard deviation \pm 4.27 and a range between 4 and 23. Majority of mothers/ caregivers 536(63.6%) had poor knowledge, 237(28.1%) had moderate/fair knowledge and 17(8.3%) of mothers/caregivers had good knowledge towards childhood pneumonia prevention and control as indicated in figure 5.3.



Figure 5. 3: Knowledge of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020

5.3.9.7 Attitude of mothers/caregivers towards pneumonia prevention and control with integration of qualitative findings

Table 5.26 presented the attitude of mothers/care givers towards pneumonia prevention and control in Sidama Regional State. About 717(85.1%) of mothers/care givers agreed, 105(12.4%) disagreed and 21(2.5%) were indifferent that pneumonia can harm children. Similarly, 783 (92.8%) of respondents agreed that a child suffering from pneumonia should be taken immediately to a health facility, 42(5.0) disagreed and 18(2.1%) were indifferent. Regarding the mothers/care givers' attitude towards pneumonia preventive methods, 726(86.1%) agreed reducing

indoors smoke and dust may help to prevent pneumonia, 85(10.0%) of them disagreed and 32(5.3%) were indifferent.

The majority of FGD discussants reported that they have thought pneumonia as risky on the health of children. One 35-year-old mother said, "Pneumonia is highly dangerous which can disturb the health of our children badly. I never forget the history when my child had diseased with pneumonia while his father was far away from us and I had been through a lot of challenges. The child had become critical, he could not eat and even he could not breathe properly and become lethargic until I took him to health facility. So, I believe that pneumonia is the worst disease for children..." (FGD1 participant).

Majority of respondents showed negative attitude towards buying antibiotics from the shop without visiting health facilities. About 602(71.4%) of respondents reported that they did not like to buy drug from shop for their child without consulting health professionals or visiting health facility when they feel that their child showed sign and symptom of pneumonia.

The findings from the qualitative part of study also supports the above evidences. One 30-year-old mother said, "We have to take our children to health professionals or health facility if they showed any sign and symptom of disease. In the past, most of communities purchase the drug or medication from anywhere even from shop when they thought that their children seem sick of pneumonia as well as other health problems. But nowadays there is a good progress as community members no longer buy road side drugs or use drugs without consulting health professionals or visiting health facilities which is a good practice that I am also following..." (FGD3 participant).

Table 5. 26: Attitude of mothers/caregivers towards p	ieumonia prevent	ion and control i	n Sidama Region	al State, Ethiopia	a: 2020
					(

Characteristics	Strongly disagree	Disagree	Indifferent	Agree	Strongly agree
Attitude towards risk of pneumonia	50(5.9)	55(6.5)	21(2.5)	380(45.1)	337(40.0)
Attitude towards taking pneumonia infected child to					
health facility immediately	30(3.6)	12(1.4)	18(2.1)	303(35.9)	480(56.9)
Reducing indoor smoke and dust may help to prevent					
pneumonia	40(4.7)	45(5.3)	32(3.8)	436(51.7)	290(34.4)
EBF and EPI may help to prevent pneumonia					
	34(4.0)	78(9.3)	42(5.0)	360(42.7)	329(39.0)
Mothers can treat cough or common cold at home					
	79(9.4)	192(22.8)	54(6.4)	416(49.3)	102(12.1)
Attitude towards getting antibiotics for child without					
visiting health facility	238(28.2)	364(43.2)	36(4.3)	130(15.4)	75(8.9)
Traditional healers sometimes can treat pneumonia					
	62(7.4)	117(13.9)	94(11.2)	331(39.3)	239(28.4)

5.3.9.8 Overall attitude level of mothers/caregivers towards prevention and control of childhood pneumonia

Another 7 questions were used to assess and evaluate the attitude of mothers/care givers towards pneumonia prevention and control in the current study area. The responses of those all 7 questions were labelled into 5 categories (strongly disagree, disagree, indifferent, agree and strongly agree). The questions were computed with SPSS version 25 and checked for its mean, standard deviation, minimum and maximum values.

Then the attitude score was recoded into different variables and categorized into negative and positive attitude. Those whose attitude with mean value and below were categorized into negative attitude while whose attitude with more than mean value were grouped into positive attitude.

The mean attitude score value was obtained as 25.9% with SD 3.237 and 11 minimum and 35 maximum value scores. As the study finding showed the majority of respondents 521(61.8%) were with positive attitude and 322(38.2%) were with negative attitude.



Negative attitude Positive attitude

Figure 5. 4: Attitude of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020

5.3.9.9 Practices of mothers/caregivers towards pneumonia prevention and control with integration of qualitative findings

The following table 5.26 presents respondents' practices which were exercised to prevent and control children from pneumonia in Sidama Regional State. Majority of mothers/caregivers 683(81%) responded that they took their child to health facility when the child seems sick of

pneumonia; while 160(19.0%) did not. Similarly, 656(77.8%) showed that they did not purchase the drug without consulting health professionals.

The discussion result showed that most of the FGD participants also supported the above findings. One 40 year old mother said, "I personally took my children to health facility while they seemed sick. I can witness that most of my neighbours also take their sick children to health facility early as much as possible..." (FGD3 participant).

Regarding breastfeeding practices, current findings revealed that only half of, (47.6%) mothers/caregivers agreed that they breastfed continuously even if their child was sick of pneumonia as compared to 442(52.4%) who did not do so.

From the qualitative findings of the current study, discussants raised their feeling differently of the above result. A 25-year-old mother reflected, "I personally do not agree with the idea of giving food or breastfeeding to a child sick of pneumonia. This is because I feel that the breast milk may

suffocate the sick child. Due to this I do not support the idea that giving food for the children while they are ill..." (FGD2 participant).

About 523(62%) of mothers/caregivers responded that they have been practicing proper clothing for their children to manage childhood pneumonia during data collection. This was taken as that they have been practicing proper dressing based on weather conditions.

Most of the FGDs discussants' findings also supported the above quantitative result positively. They argued that properly clothing of the child may help to reduce/manage/ childhood pneumonia. One 36-year-old mother said, "Childhood pneumonia can be managed by proper clothing of the children when they became sick. I believe that most of the pneumonia cases among under-five children could happen when their mothers did not follow the proper clothing approach, especially when there might be the interaction of hot and cold weather. So, to minimize or manage challenges such like this, it is better to dress children with right clothes according to the weather conditions..." (FGD1 participant).

 Table 5. 27: Mothers'/caregivers' practices towards prevention and control of childhood pneumonia in Sidama Regional State, Ethiopia:

 2020, (N=843)

No	Variables for assessing practice of respondents towards pneumonia prevention and control	Yes		No	
		Frequency (n)	(%)	Frequency (n)	(%)
1	Took child to the nearest health facility immediately when child seems sick of pneumonia	683	81.0	160	19.0
2	Took child to the traditional healers when child seems sick of pneumonia	155	18.4	688	81.6
3	Purchase drug from pharmacist without consulting health professional when child seems sick of pneumonia	187	22.2	656	77.8
4	Took child to the religious leaders when child seems sick of pneumonia	214	25.4	629	74.6
5	Purchase drug from any shop when child seems sick of pneumonia	196	23.3	674	76.7
6	Taking child to Health facility within 24 hours when child seems sick of pneumonia	524	62.2	319	37.8
7	Breast fed continuously when child seems sick of pneumonia	401	47.6	442	52.4
8	Vaccinating the children to prevent pneumonia	732	86.8	111	13.2
9	Using ventilation to prevent pneumonia	494	58.6	349	41.4
10	*Proper clothing the child to manage pneumonia when seems sick of pneumonia	523	62.0	320	38.0

*Proper clothing= wearing children as per their body temperature and weather conditions

5.3.9.10 Overall level practice of mothers/caregivers towards childhood pneumonia prevention and control

There were ten different questions used to evaluate and check the practice of mothers/care givers towards pneumonia prevention and control whether it was proper or improper practice. The response of each individual was labelled as "0" and "1" where "0" for improper practice and "1" for proper practice.

The responses were computed with SPSS version 25 and checked for its mean. The value of respondents practice level below mean practice score

value were categorized as improper practice and those practice level was equal to mean score value and those above were categorized as proper practice.

The findings indicated that mean practice score value was 7.1 with minimum and maximum value of 0 and 10 respectively. About 486(57.5%) of mothers/caregivers had practiced the childhood pneumonia prevention and control ways properly while 358(42.5%) showed improper practice.



Figure 5. 5: Practice of mothers/caregivers towards childhood pneumonia prevention and control in Sidama Regional State, Ethiopia: 2020 (N=843)

5.3.9.11 Factors associated with KAP of mothers/care givers towards childhood pneumonia prevention and control

To assess the associated factors with the level of KAP of mothers/caregivers, the three-level category of knowledge was recoded into dichotomous namely adequate and inadequate knowledge from which below mean score value was inadequate while equal and above mean value labelled as adequate knowledge. For attitude and practice the variables already dichotomized (positive and negative attitude; and proper and improper practice) towards childhood pneumonia prevention and control.

Both bivariate and multivariate analysis were conducted with the sociodemographic and economic characteristics and level of KAP of mothers to identify the contributing factors.

5.3.9.11.1 Factors associated with knowledge of mothers/caregivers towards childhood pneumonia prevention and control

Table 5.28 presents factors associated with level of knowledge of mother/caregivers on pneumonia prevention and control at the current study area. At the bivariate analysis, 12 variables were investigated for association and variables with their p-value <0.25 were taken as significant factors and six variables were found as candidate variables for multivariate analysis. The six selected variables were included in the multivariate analysis and conditional stepwise back-ward method was performed. Three variables (the presence of a functional radio, educational status of the child's father and parity of the mother) were found as significantly associated with level of knowledge.

The presence of a functional radio was found as one of the positively associated factors for having adequate knowledge among mothers/caregivers at the current study area (AOR: 1.28; CI, 0.94, 1.75; p=0.047). In addition, the increased level of educational status of child's father was found to be significant contributing factor for having adequate knowledge among mothers/care givers (AOR=1.88; CI: 0.86, 4.08; p=0.005). Mothers who had two children also found to be positively associated factor with having adequate knowledge among mothers/caregivers (AOR=1.86; CI: 1.20, 2.89; p=0.005).

No	Characteristics	Status of knowledge		COR of 95%CI	AOR of 95%CI	P-value
		Adequate	Inadequate			
1	Number of children mothers had					
	One	74(37.9)	121(62.1)	1	1	
	Two	95(44.2)	120(55.8)	1.29(0.87, 1.92)	1.41(0.93, 2.14)	0.10
	Three	92(51.4)	87(48.6)	1.73(1.14, 2.61)*	1.86(1.20, 2.89)	0.005**
	Four	49(38.6)	78(61.4)	1.03(0.65, 1.63)	1.13(0.70, 1.84)	0.64
	Five and above	49(38.6)	78(61.4)	1.03(0.65, 1.63)	1.20(0.74, 1.95)	0.45
2	Place of delivery					
	Health Institution	281(44.9)	345(55.1)	1.45(1.05, 1.99)*	1.34(0.96, 1.88)	0.90
	Home	78(35.9)	139(64.1)	1	1	
3	Presence of TV					

 Table 5. 28: Socio-demographic and economic factors associated with knowledge of mothers/caregivers towards pneumonia prevention and control in Sidama Regional State: 2020, (N=843)

	Yes	79(52.7)	71(47.3)	1.64(1.15, 2.34)*	1.11(0.74, 1.66)	0.61
	No	280(40.4)	413(59.6)	1	1	
4	Presence of radio					
	Yes	157(49.2)	162(50.8)	1.54(1.17, 2.05)*	1.28(0.94, 1.75)	0.047**
	No	202(38.5)	322(61.5)	1	1	
5	Educational level of child's father					
	Illiterate/no formal education	37(34.6)	70(65.4)	1	1	
	1-8 grade	189(38.7)	300(61.3)	1.19(0.77, 1.85)	1.08(0.69, 1.71)	0.73
	9-12 grade	88(50.0)	88(50.0)	1.89(1.15, 3.11)*	1.64(0.97, 2.79)	0.07
	Certificate and above	37(62.7)	22(37.3)	3.18(1.64, 6.16)*	1.88(0.86, 4.08)	0.005**
6	Educational level of respondents					
	Illiterate/no formal education	74(39.8)	112(60.2)	1	1	
	1-8 grade	211(40.4)	311(59.6)	1.03(0.73, 1.44)	0.93(0.65, 1.34)	0.71
	9-12 grade	44(46.8)	50(53.2)	1.33(0.81, 2.20)	0.99(0.57, 1.75)	0.99
	Certificate and above	30(73.2)	11(26.8)	4.13(1.95, 8.74)	2.11(0.86, 5.18)	0.10

1 indicates reference category, * indicates, Significant association at p<=0.25 on bivariate analysis and ** indicates significant association at p< 0.05 on multivariate analysis

5.3.9.11.2 Factors associated with attitude of mothers/caregivers towards childhood pneumonia prevention and control

Different factors were considered and checked through both bivariate and multivariate analysis to identify weather there was significant association within the positive or negative attitude of mothers/care givers towards childhood pneumonia prevention and control at the study area.

Ten socio-demographic and economic variables were included in the bivariate analysis to investigate the influence on the attitude of mothers/caregivers towards childhood pneumonia prevention and control at the study area. Six variables were found as candidate variables for multivariate analysis with a p-value of <0.25. After multivariate analysis,

three factors (knowledge of mothers/care-givers, age of respondents and number of children the mothers had) were found to be significantly associated with the attitude of mothers/caregivers.

Mothers/caregivers who had adequate knowledge towards childhood pneumonia prevention and control were significantly associated with positive attitude of mothers/caregivers towards the prevention and control of pneumonia (AOR: 0.64; 95%CI: 0.48, 0.86; p=0.003). Furthermore, mothers/caregivers age older than or equal to 36 years had positive attitude than the other age groups (AOR=0.64; CI: 0.48, 0.86; p=0.003). Lastly, having five and more children was positively associated with mothers/care-givers' positive attitude towards pneumonia prevention and control with p-value 0.01 and (AOR=1.96; CI: 1.15, 3.35) as indicated in table 5.29.

Table 5. 29	9: Factors associated	with attitude o	of mothers/caregive	rs towards pneu	monia preventio	on and control in Sid	lama Regional S	State:
2020, (N=8	843)							

No	Characteristics	Attitude status		COR of 95%CI	AOR of 95%CI	P-value
		Positive	Negative			
1	Knowledge status of respondents					
	Adequate	279(57.6)	205(42.4)	0.66(0.49, 0.87)*	0.64(0.48, 0.86)	0.003**
	Inadequate	242(67.4)	117(32.6)	1	1	
2	Age of respondents					
	16-24	125(61.9)	77(38.1)	1	1	
	25-35	325(64.0)	183(36.0)	1.09(0.78, 1.53)	1.07(0.74, 1.52)	0.71
	>=36	71(53.4)	62(46.6)	0.71(0.45, 1.10)*	0.51(0.30, 0.87)	0.01**
3	Number of children mothers had					
	One	119(61.0)	76(39.0)	1	1	
	Two	129(60.0)	86(40.0)	0.96(0.64, 1.42)	0.94(0.62, 1.42)	0.78

Three	106(59.2)	73(40.8)	0.93(0.61, 1.40)	0.94(0.60, 1.46)	0.97
Four	80(63.0)	47(37.0)	1.10(0.69, 1.72)	1.27(0.77, 2.08)	0.34
Five and above	87(68.5)	40(31.5)	1.40(0.87, 2.23)*	1.96(1.15, 3.35)	0.01**
Job of respondent					
Housewife	364(60.2)	241(39.8)	1	1	
Government employer	44(59.5)	30(40.5)	0.97(0.5, 1.59)	0.94(0.53, 1.66)	0.83
Merchant	66(71.7)	26(28.3)	1.68(1.04, 2.72)*	1.62(0.98, 2.66)	0.06
Daily labourer	7(63.6)	4(36.4)	1.16(0.34, 4.00)	1.32(0.36,4.81)	0.68
Farmer	40(65.6)	21(34.4)	1.26(0.73, 2.19)	1.39(0.78, 2.47)	0.26
Educational level of respondent					
Illiterate	113(60.8)	73(39.2)	1	1	
1-8 grade	313(60.0)	209(40.0)	0.97(0.69, 1.36)	0.86(0.60, 1.23)	0.42
9-12 grade	69(73.4)	25(26.6)	1.78(1.04, 3.07)*	1.50(0.84, 2.66)	0.17
Certificate and above	26(63.4)	15(36.6)	1.12(0.56, 2.26)*	0.95(0.42, 2.13)	0.90
	Three Four Five and above Job of respondent Housewife Government employer Merchant Daily labourer Farmer Educational level of respondent Illiterate 1-8 grade 9-12 grade Certificate and above	Three 106(59.2) Four 80(63.0) Five and above 87(68.5) Job of respondent 106(59.2) Housewife 364(60.2) Government employer 44(59.5) Merchant 66(71.7) Daily labourer 7(63.6) Farmer 40(65.6) Educational level of respondent 113(60.8) 1-8 grade 313(60.0) 9-12 grade 69(73.4) Certificate and above 26(63.4)	Three 106(59.2) 73(40.8) Four 80(63.0) 47(37.0) Five and above 87(68.5) 40(31.5) Job of respondent Housewife 364(60.2) 241(39.8) Government employer 44(59.5) 30(40.5) Merchant 66(71.7) 26(28.3) Daily labourer 7(63.6) 4(36.4) Farmer 40(65.6) 21(34.4) Educational level of respondent 113(60.8) 73(39.2) 1-8 grade 313(60.0) 209(40.0) 9-12 grade 69(73.4) 25(26.6) Certificate and above 26(63.4) 15(36.6)	Three106(59.2)73(40.8)0.93(0.61, 1.40)Four80(63.0)47(37.0)1.10(0.69, 1.72)Five and above87(68.5)40(31.5)1.40(0.87, 2.23)*Job of respondentHousewife364(60.2)241(39.8)1Government employer44(59.5)30(40.5)0.97(0.5, 1.59)Merchant66(71.7)26(28.3)1.68(1.04, 2.72)*Daily labourer7(63.6)4(36.4)1.16(0.34, 4.00)Farmer40(65.6)21(34.4)1.26(0.73, 2.19)Educational level of respondent113(60.8)73(39.2)11-8 grade313(60.0)209(40.0)0.97(0.69, 1.36)9-12 grade69(73.4)25(26.6)1.78(1.04, 3.07)*Certificate and above26(63.4)15(36.6)1.12(0.56, 2.26)*	Three106(59.2)73(40.8)0.93(0.61, 1.40)0.94(0.60, 1.46)Four80(63.0)47(37.0)1.10(0.69, 1.72)1.27(0.77, 2.08)Five and above87(68.5)40(31.5)1.40(0.87, 2.23)*1.96(1.15, 3.35)Job of respondent </td

1 indicates reference category, * indicates, Significant association at $p \le 0.25$ on bivariate analysis and ** indicates significant association at p < 0.05 on multivariate analysis

5.3.9.11.3 Factors associated with the practice of mothers/caregivers towards childhood pneumonia prevention and control

From multivariate analysis four variables (knowledge of mothers/care givers, age of respondents, and presence of radio and educational level of respondents) were obtained as significantly associated factors with practice of mothers/caregivers towards pneumonia prevention and control in study area.

Mothers/caregivers with fair knowledge showed proper practice towards pneumonia prevention and control (AOR=1.56; 95%CI: 1.12, 2.17;

p=0.008). Also the presence of functional radio (AOR=1.53; 95%CI: 1.11, 2.11; p=0.009) and mothers/caregivers between the age group of 25-35 years (AOR=1.52; 95%CI: 1.08, 2.16; p=0.02) were positively associated factors with the proper practice of mothers/caregivers to prevent and control pneumonia among under-five children respectively. Similarly, mothers/caregivers with both primary (AOR=1.72; 95%CI: 1.20, 2.48; p=0.004) and secondary educational level (AOR=3.01; 95%CI: 1.70, 5.34; p=0.001) were found to show more proper practice towards childhood pneumonia prevention and control than the rest with different level of education (see table 5.30).

Table 5.	30: Factors	s associated with practic	e of mothers/caregivers towards j	oneumonia prevention a	nd control in Sidama	Regional State:
2020, (N	(=843)					

No	Characteristics	Practice status		COR of 95%CI	AOR of 95%CI	P-value
		Proper	Improper			≤0.05
1	Wealth index					
	Poorest	75(48.1)	81(51.9)	1	1	
	Poor	102(57.6)	75(42.4)	1.47(0.95, 2.26)*	0.52(0.32, 0.82)	0.06
	Middle	107(61.8)	66(38.2)	1.75(1.23, 2.72)*	0.91(0.58, 1.43)	0.68
	Rich	104(61.9)	64(38.1)	1.75(1.23, 2.73)	1.03(0.65, 1.62)	0.91
	Richest	97(57.4)	72(42.6)	1.45(0.94, 2.25)	0.96(0.60, 1.51)	0.85
2	Knowledge status of respondents					
	Poor	286(54.4)	250(46.6)	1	1	
	Fair	155(65.4)	82(34.6)	1.65(1.20, 2.27)*	1.56(1.12, 2.17)	0.008**
	Good	44(62.9)	26(37.1)	1.48, 0.88, 2.47)	1.45(0.84, 2.50)	0.18
3	Age of respondents					
	16-24	105(52.0)	97(48.0)	1	1	
	25-35	310(61.0)	198(39.0)	1.45(1.04, 2.01)*	1.52(1.08, 2.16)	0.02**
	>=36	70(52.6)	63(47.4)	1.03(0.66, 1.59)	1.53(0.84, 2.18)	0.21
4	Presence of radio					

Yes	210(65.8)	109(34.2)	1.74(1.31, 2.35)*	1.53(1.11, 2.11)	0.009**
No	275(52.5)	249(47.5)	1		
Marital status of respondents					
Single	8(42.1)	11(57.9)	1	1	
Married	467(58.3)	334(41.7)	1.92(0.76, 4.83)*	2.00(0.77, 5.22)	0.15
Widowed	10(43.5)	13(56.5)	1.06(0.31, 3.61)	1.16(0.32, 4.26)	0.82
Educational level of respondents					
Illiterate/no formal education	84(45.2)	102(54.8)	1	1	
1-8 grade	306(58.6)	216(41.4)	1.72(1.23, 2.41)*	1.72(1.20, 2.48)	0.004**
9-12 grade	67(71.3)	27(28.7)	3.01(1.77, 5.13)*	3.01(1.70, 5.34)	0.001**
Certificate and above	28(68.3)	13(31.7)	2.61(1.27, 5.36)	1.93(0.88, 4.22)	0.10
	Yes No Marital status of respondents Single Married Widowed Educational level of respondents Illiterate/no formal education 1-8 grade 9-12 grade Certificate and above	Yes210(65.8)No275(52.5)Marital status of respondents	Yes 210(65.8) 109(34.2) No 275(52.5) 249(47.5) Marital status of respondents 2 1000000000000000000000000000000000000	Yes 210(65.8) 109(34.2) 1.74(1.31, 2.35)* No 275(52.5) 249(47.5) 1 Marital status of respondents 2 2 2 1 Single 8(42.1) 11(57.9) 1 Married 467(58.3) 334(41.7) 1.92(0.76, 4.83)* Widowed 10(43.5) 13(56.5) 1.06(0.31, 3.61) Educational level of respondents 2 102(54.8) 1 Illiterate/no formal education 84(45.2) 102(54.8) 1 1-8 grade 306(58.6) 216(41.4) 1.72(1.23, 2.41)* 9-12 grade 67(71.3) 27(28.7) 3.01(1.77, 5.13)* Certificate and above 28(68.3) 13(31.7) 2.61(1.27, 5.36)	Yes210(65.8)109(34.2)1.74(1.31, 2.35)*1.53(1.11, 2.11)No275(52.5)249(47.5)1Marital status of respondents249(47.5)11Single8(42.1)11(57.9)11Married467(58.3)334(41.7)1.92(0.76, 4.83)*2.00(0.77, 5.22)Widowed10(43.5)13(56.5)1.06(0.31, 3.61)1.16(0.32, 4.26)Educational level of respondents11Illiterate/no formal education84(45.2)102(54.8)111-8 grade306(58.6)216(41.4)1.72(1.23, 2.41)*1.72(1.20, 2.48)9-12 grade67(71.3)27(28.7)3.01(1.77, 5.13)*3.01(1.70, 5.34)Certificate and above28(68.3)13(31.7)2.61(1.27, 5.36)1.93(0.88, 4.22)

1 indicates reference category, * indicates, Significant association at $p \le 0.25$ on bivariate analysis and ** indicates significant association at p < 0.05 on multivariate analysis

5.4 CONCLUSSIONS

This chapter presented the result of the study following the research objectives. The prevalence of childhood pneumonia in the current study was reported and the associated factors namely: socio-demographic and economic, behavioural and disease related, environmental and nutritional and immunization related were also clearly justified and presented. This chapter also presented with level of knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control and also its associated factors.

The next chapter will present with discussion of result.

Chapter 6

DISCUSSION OF THE RESEARCH FINDINGS

6.1 INTRODUCTION

This chapter presents the discussion of the study results linked with the stated study objectives. The research findings are also discussed in relation to different previous studies conducted on similar topics or concepts.

Findings were discussed under sub topics of prevalence of childhood pneumonia, different categories of associated factors of pneumonia, level of knowledge, attitude and practice of mothers with aged under five children towards childhood pneumonia and factors associated with KAP of mothers/caregivers towards childhood pneumonia prevention and control were discussed here.

```
6.2 PREVALENCE OF CHILDHOOD PNEUMONIA AND
ASSOCIATED FACTORS
```

6.2.1 Prevalence of pneumonia among under five children in Sidama Regional State

Understanding the burden and associated factors of childhood pneumonia is important for the management and development of proper prevention strategies. The result of current study indicates that the prevalence of pneumonia among children aged under-five years in study area is 22.2%. The prevalence of the current study is higher than that of conveyed from the studies carriedout in northwest Ethiopia and Ghana which were 16.1% and 18.40% respectively (Osei 2018:1 & Fekadu et al 2014:150). However, the result of current study is lower than the results obtained

from the studies conducted in southern and southwest Ethiopia which were 33.5% and 28.1% respectively (Lema 2018:25 and Abuka 2017:19). This difference might be due to study setting, periodic variations and the difference of implementation of health extension program especially on child health management. Most of the above listed studies whose values are higher than the value of current study were conducted at facility base and this might also makes the difference.

6.2.2 The association between socio-demographic and economic factors and the occurrence of childhood pneumonia

Different socio-demographic and economic factors like medium and large family size, respondents with no formal education, the middle age group of children (7-36 months) and less than or 2 years of birth interval of children were found as associated factors for the occurrence of childhood pneumonia in the current study area. The educational level of mothers/caregivers significantly influenced the occurrence of pneumonia (p=0.002). This implies that children who had mothers/caregivers with no level of education were likely to suffer from Pneumonia. The same results were reported from studies done in Vietnam and India that showed the low literacy level of mothers were significantly associated with the occurrence of pneumonia (Hoang 2019:277 & Gupta 2019:951).

Current findings revealed that having medium and large family size were found to be significantly associated with the occurrence of under-five pneumonia with p-values less than 0.001. This implies children who lived in households with large family sizes were more likely to suffer from pneumonia than their counterpart in small family sizes. These findings are in line with findings obtained from a study carriedout in southeast Nigeria which revealed that children with large family size can develop pneumonia than the others (Ujunwa and Ezeonu 2014:95). These might be due to the overcrowding; hence overcrowding is suitable means for the transmission of communicable diseases. For instance, a study conducted in Bellary and India showed that overcrowding was found to significantly influence the occurrences of childhood pneumonia (Srivastava et al 2015:1 & Hemagiri et al 2014:47).

Findings showed that age of children can also contribute to the occurrence of childhood pneumonia. Children with age group 7-12 and 13-36 months were found for being significantly associated with childhood pneumonia occurrence (p=0.001). That implies children who were in the age group of 7-36 months were more likely to suffer from pneumonia than the most lower and upper age groups of children. Similar findings were obtained

from the study conducted in Ethiopia which showed that children in the age group of 2-12 months were found to be determinant factor for the occurrence of pneumonia. Also, the same findings were obtained from a study conducted in Rwanda which revealed that children in the age of 0-11 months and 24-59 months being found as negatively associated with the occurrence of pneumonia. On the contrary, the risk of suffering from pneumonia reduced among children aged 2-11 and 12-23 months in urban areas of Ethiopia (Abuka 2017:19, Harerimana et al 2016:1 and Dadi et al 2014:1). These differences might be due to the environmental exposure of children with age and the degree of the care that mother/caregiver offer to their children. Younger children have the opportunity of always being carried by the mothers/caregivers. Hence, they have not started walking. Being constantly with/on mothers/caregivers' carriage may prevent them from the exposure to move on their own, and that might contribute to the occurrence of any illness including childhood pneumonia.

The study findings revealed that birth interval of children which is less than two years was found as contributing factor of childhood pneumonia (p=0.02). This implies that children whose birth interval is less than 2 years from their older ones were likely to suffer from pneumonia. This might be due to that children who had less than 2 years birth interval may lack proper care from their mothers/caregivers for they could be busy.

6.2.3 The association between behavioural and history of disease related factors and the occurrence of childhood pneumonia

From multivariate analysis, different behavioural and history of disease related factors were found to be meaningfully associated with the occurrence of childhood pneumonia in the current study area. Presence of family members with history of respiratory disease was found as one of the associated factors for the occurrence of pneumonia with (p<0.001); and this implies children who are from family member where at least one of the members with history of respiratory disease were more likely to suffer from pneumonia than their counterparts. Similar findings were obtained from studies carriedout in different parts of Ethiopia which reported that children who lived in households' member with history of acute respiratory infections were at higher likelihoods of developing pneumonia compared to their counterparts (Lema et al 2019:1 and Dadi et al 2014:1). Furthermore, results obtained from a study done in Gambia showed that bed-sharing with someone who had cough was significant risk factor for severe pneumonia. In addition, a hospital-based case control study carriedout in India showed that family history of URTI emerged as risk factor for the occurrence of acute severe pneumonia in under five children (Gupta and Bhadrala 2019:949 and Howie et al 2016:1405). Most of the respiratory diseases are infectious and can be easily transmitted from one individual to another. Children are susceptible for any diseases and especially can easily be infected from household family members.

Children having any disease history during data collection period were also obtained to be significantly associated with the prevalence of pneumonia among under-five children (p=0.03). That is to mean children who were with any other illness during the time of data collection were more likely to suffer from pneumonia than those to the contrary. These results are supported with previous results obtained from research done in different parts of Ethiopia, which showed that children with the history of diarrhea were at higher likelihoods of developing pneumonia compared to those who did not (Markos et al 2019:1 and Dadi et al 2014:1). This might be due to that children suffering from any disease can be immune compromised and susceptible for any other infections including pneumonia.

The study findings reported that children with the history of measles disease in the previous year preceding to data collection were found to be contributing factors of childhood pneumonia (p<0.001). Which implies that children who had history of measles disease with in the earlier one year prior to data collection were more likely to develop pneumonia than those who did not. Similar findings were obtained from studies conducted in Vietnam revealed that children with pneumonia were 25 times more likely to presented with associated measles (Hoang 2019:277). Children pre-exposed for measles might be immune compromised and can be easily affected by other infections.

6.2.4 The association between environmental factors and the occurrence of childhood pneumonia

Five different environmental factors namely; households with no latrine, not having separate house for animal and human, absence of window in the living house, charcoal use and presence of chimney during the time of data collection were found to be positively associated factors with the occurrence of childhood pneumonia in the current study area. Children who lived in a household that did not have separate shelter for animal and human being were at a higher degree of suffering from pneumonia (p=0.001). Current findings concur with results obtained from a study carriedout in northwest Ethiopia which shown that keeping cattle inside the main house was associated with childhood pneumonia (Fekadu et al 2014:150). This might be due to poor sanitation which then can create suitable transmission opportunity for the pneumonia causative agents and children could be easily affected with pneumonias a result.

Results from multivariate analysis revealed that the use of charcoal as the main source of energy in the household significantly influenced the prevalence of childhood pneumonia in the current study area (p<0.001). Burning of charcoal can cause household air pollution and might increase the risk of pneumonia. Alike results were obtained from a study done in Northwest Ethiopia, and it showed that using charcoal was one of the determinants of childhood pneumonia (Fekadu et al 2014:1).

Absence of window in the living house was another positively associated factor with the occurrence of childhood pneumonia in the current study (p<0.001). Children who from absence of window in the living house were more likely to develop pneumonia than their complements. Poor ventilation and indoor air pollution may create opportunity for the transmission of pneumonia causative agents (Yeimo et al 2018:307). These results were supported by a study conducted by Lema et al (2019:7), who noted that children who lived in the house where there were three or above windows were 2.3 times less susceptible to develop pneumonia compared to those who lived in the house that has no window in Ethiopia.

Household without latrine was also found as significantly associated factor of childhood pneumonia (p<0.001) in the current study area. Lack of latrine may lead to poor sanitation and that may also contribute for the existence of pneumonia causative agents and children can be easily infected with the pneumonia. This was corroborated by Yeimo et al (2018:307) in the conclusion that poor sanitation was one of the factors associated with pneumonia in under-five children and also by Lema et al, (2019:5) who also corroborated that children from household without latrines were about 2.5 times more likely to develop pneumonia compared to their complements in household with latrine.

Presence of chimney inside the house during data collection was also found as the significantly associated factors of pneumonia among underfive children (p=0.003) in Sidama Regional State. Children who lived with the house where there was chimney were more likely to develop pneumonia than those who did not reside in a household where there was no chimney. Similar result was obtained from the study conducted in Ethiopia, which revealed that children leaving in a household with a chimney were found to increase the odds of childhood pneumonia (Mesafint et al. 2020:9).

6.2.5 The association between nutritional and immunization related factors and the occurrence of childhood pneumonia

Findings in the current study showed that nutritional as well as immunization related factors were also found to be major contributing factors of childhood pneumonia in the current study area. Severely underweight was one of the positively associated factors with pneumonia in under-five children (p=0.003). Malnutrition can weaken the body's disease fighting ability by affecting the immunity. Hence the severely malnourished children are the vulnerable group for any diseases. These results, are aligned with the results obtained from India, Gambia and Ethiopia, and showed that malnutrition was one of the determinants of childhood pneumonia (Lema et al 2018:25, Howie et al 2016:1405 and Srivastava et al 2015:1). In addition, a literature review study carriedout in Pakistan also showed that malnutrition was a major factor of childhood pneumonia (Aftab et al 2016:1, Ramezani 2015:1173 and Hemagiri et al 2014:47).

Lack of immunization was also found to positively influence the occurrence of childhood pneumonia. Children who, did not get measles vaccine were more likely to develop pneumonia than those who were vaccinated, according to the current findings (p=0.002). Immunization is a way of preventing the infectious disease thus children who are not immunized are at risk of being infected. It is known that immunization can prevent different diseases in children including pneumonia. These findings align with results from studies conducted both in Ethiopia and Pakistan which revealed that lack of vaccination was one of the major risk factors of pneumonia (Lema et al 2018:25 and Aftab et al 2016:1). Similarly, incomplete immunization per one's age was also significant risk factor for pneumonia among under-five children in India (Gupta and Bhadrala 2019:949, Srivastava et al 2015:1 and Hemagiri et al 2014:47).

A statistically significant association exists between lack of vitamin A supplementation and the occurrence of childhood pneumonia (p<0.001). Vitamin A supplementation is administered to strengthen immune system of children and may protect them from diseases. Improving vitamin A status of underprovided children through supplementation increase their opposition to disease and can decrease death from all causes about 23% (UNICEF 2007:3). Likewise, findings from earlier local study by Lema et al (2018:28) noted that children who supplemented with vitamin A were 83% less probable to have pneumonia when compared to children who did not supplimented with Vitamin A.

However, feeding child with hand was obtained as one of the negatively associated factors with the occurrence of pneumonia (p<0.001). This implies that children who were hand-fed by their mothers/caregivers were 72% less likely to develop pneumonia than those fed with bottle. Poor sanitation of bottle feeding due to improper washing might facilitate bacteria growth which can cause pneumonia. Similar finding obtained from case control study conducted in Pakistan revealed that bottle feeding was one of the contributing factors for complicated pneumonia (Aftab et al 2016:1).

6.2.6 Level of knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control

Current findings indicated that majority of respondents (63.6%) had overall poor knowledge while 28.1% had fair and only 8.3% had good knowledge towards pneumonia prevention and control. This indicates that the majority of mother/caregivers lack basic knowledge about pneumonia

prevention and control where having such knowledge is one of the fundamentals that must be presented to build a positive health. These findings coincide with previous findings conducted in Nigeria and Bangladesh which revealed mothers for having poor knowledge on the severity of childhood pneumonia (Pradhan et al 2016:35 & Ferdous et al 2014:1). On the contrary, findings from studies conducted in India reported that 41.3% of mothers had fair knowledge (Yahaya et al 2018:35). This difference might be due to study setting, way of implementing health education program and it might be also socio-economic status variations within each study area.

In the current study, the majority of mothers/caregivers (61.8%) had positive attitude compared to 38.2% of mother/caregivers who had negative attitude towards childhood pneumonia prevention and control as such they feel better to prevent and control pneumonia among under five children. The current findings are higher than that of research findings obtained from the studies conducted in Saudi Arabia and India which reported that more than half of the mothers had poor attitude towards childhood pneumonia prevention and control (Alluqmani et al 2017:1959 & Pradhan et al 2016:35). The difference might be attributed to the level of understanding of the severity of childhood pneumonia and rate of exposure to pneumonia among the mothers/caregivers.

The current study revealed that more than half of the mothers/caregivers (57.5%) reported that they practice measures of prevention and control of pneumonia. Contrarily, research findings obtained from a study conducted in Saudi Arabia revealed that practice of mothers towards ARI including pneumonia was poor (Alluqmani et al 2017:1959). In addition, results from a cross sectional survey conducted in Pakistan reported that 58% of mothers practiced self-medication (Bham et al 2016:1557), which is an improper practice. This might be due to the difference among mothers/caregivers' perception towards health seeking behaviour and understanding of benefits of visiting health facility or consulting health professionals.

6.2.7 The association of factors with knowledge of mothers/caregivers towards pneumonia prevention and control

The current study analysis indicated the determinants of adequate knowledge towards pneumonia prevention and control in under five children. Mothers/caregivers who owned a functional radio had better knowledge than those who did not possess one (p=0.05). This implies households with functional radio could easily access information related to child health including pneumonia. The presence of a radio in households might also indicate a higher economic status or wealth index of the family and positively correlate with the knowledge of mothers/caregivers. These findings coincide with results obtained from India which reported that maternal socioeconomic status have an important role in knowledge towards respiratory disease (Mutalik and Raje 2017:29).

Respondents' level of education was significantly associated with maternal/caregivers' adequate knowledge of childhood pneumonia prevention and control measures. Increased level of education was positively associated with having knowledge on pneumonia prevention and control among under five children (p=0.005). This might be due to the fact that mothers with better literacy can understand issues than those who did not have formal education. Similar findings were found from a cross-sectional study done in India and showed the increased level of mothers' education; and this was significantly associated with the increased knowledge of childhood pneumonia among mothers (Pradhan et al 2016:36). Furthermore, studies conducted in Pakistan and Saudi Arabia revealed that having better literacy or high educational level to be

positively associated with good knowledge towards pneumonia in under five children (Alluqmani et al 2017:1959 & Bhamet al 2016:1557).

Number of children mothers had also was found to be significantly associated with adequate knowledge in current study area. Mothers who had two children (p=0.005) were obtained to be associated with adequate knowledge. Mothers/caregivers who had two children were likely to have adequate knowledge than the others. This might be due to the fact that as number of children increases, there might be increasing experience for mothers also; and this may support them to understand or acquire knowledge about childhood pneumonia. The current findings are consistent with those of Keter (2015:34) that reported mothers in Kenya with more than three children were more likely to have knowledge of pneumonia.

6.2.8 The association of factors with attitude of mothers/caregivers towards pneumonia prevention and control

From the current study, majority of mothers/caregivers showed positive attitude towards pneumonia prevention and control among under five children. Adequate knowledge of respondents about pneumonia was found to have a significant association with having positive attitude towards childhood pneumonia prevention and control (p=0.003). Mothers/caregivers who had adequate knowledge were more likely to have positive attitude than their counterparts. These findings are also aligned with the results of a cross sectional study carriedout in India that revealed the existence of substantial association between the level of knowledge and insight of mothers on childhood pneumonia (Pradhan et al 2016:35). The reason might be the difference in understanding of childhood pneumonia. Mothers who are knowledgeable about childhood pneumonia may think more positively than those who did not recognise the severity of pneumonia among under five children of theirs.

The older age of respondents was found to be negatively associated with the attitude towards pneumonia prevention and control in under five children. Mothers/caregivers of aged >=36 years were less likely to have positive attitude towards pneumonia prevention and control than the younger ones (p=0.01). Similarly, findings obtained from a study conducted in Saudi Arabia showed that better thoughtful and adequate knowledge was significantly associated with young ages (Alluqmani et al 2017:1959). The reason might be that younger mothers/caregivers may have better literacy and understanding than the older age groups of respondents and this may lead them to have positive attitude.

Having five and above children was also found to be another factor for the contribution of mothers/caregivers to have positive attitude towards pneumonia prevention and control. Mothers/caregivers with \geq 5 children have positive attitude (p=0.01). The proper justification for this might be that mothers/caregivers may have positive attitude as their experience increases as the number of their children increases.

6.2.9 The association of factors with practice of mothers/caregivers towards pneumonia prevention and control

The current study results indicate that majority of mothers/caregivers had proper practice towards childhood pneumonia prevention and control at the study area. Mothers/caregivers with fair knowledge had proper practice than those who did not (p=0.01). The result of a cross sectional study carriedout in Kenya showed that mothers who had better knowledge showed good practice (Keter 2015:51). The reason might be that pre information or being knowledgeable about childhood pneumonia may support to do the right thing regarding pneumonia prevention and control.

The study result showed that mothers/caregivers living in households with a functional radio had positive practice towards the prevention and control of pneumonia (p=0.01). This implies that respondents who had functional radio had proper practices than those who did not have the devise. The reason might be that household with radio may get information timely and that may support them to practice positively.

Current findings revealed that mothers/caregivers with 1-8 grade (p=0.004) and 9-12 grade level of education (p=0.001) had better practice towards pneumonia prevention and control. Mothers/caregivers who grew on their education further had good practice than those with lower education. Similar studies conducted in India, Pakistan and Nigeria showed that better education of mothers/caregivers can influence them positively to practice pneumonia prevention and control measures effectively (Mutalik and Raje 2017:29, Bham et al 2016:1557 and INdu et al 2015:1). This might be due to the fact that educated mothers/caregivers can understand the severity of childhood pneumonia in detail and may take positive actions towards pneumonia prevention and control measures.

6.3 CONCLUSSION

This chapter presented the discussion of results of the current study. The chapter commenced through the introduction and study findings in the current study area which included the prevalence of childhood pneumonia and its socio-demographic, economic, environmental, behavioural, disease, nutritional and immunization related factors. Additionally, it presented the level of knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control and its association with the socio-demographic factors. The chapter also included the utilization of different available and valued research findings that are used for the comparison of the current study findings following with the scientific reference using approach.

The next chapter will present the conclusions, recommendations and limitations.

Chapter 7

CONCLUSIONS, RECOMMENDATIONS AND LIMITATIONS

7.1 INTRODUCTION

This chapter presents key findings, conclusions based on the study findings, major contributions and its limitations. Furthermore the chapter presents the recommendations, issues for further research and concluding remarks.

7.2 SUMMARY AND INTERPRETATION OF THE RESEARCH FINDINGS

The purpose of the study was to develop awareness and prevention strategies on pneumonia among under-five years' children in Sidama regional state, Ethiopia in line with the strategy to accelerate the goal of ending preventable pneumonia disease by 2025, through assessing the prevalence of childhood pneumonia and associated factors, and also assessing knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control and its associated factors. The study addressed issues to develop awareness and prevention strategies aligned with the integrated Global Action Plan for Pneumonia (GAPP).

The data were collected from 843 respondents at household level for the quantitative part of the study making the response rate of 99.8%; and six

(06) FGDs were conducted with a total of 40 mothers of under-five children for qualitative part as well.

The quantitative data were analysed using descriptive and analytic statistics; while the qualitative data analysis followed a process of data reduction, presentation, and conclusion drawing. The original records in *Sidaamu Afoo* were transcribed verbatim into English language by the researcher from the tape-recorded interviews, field-notes and then the researcher organised the data thematically. Both qualitative and quantitative data were analysed concurrently.

7.2.1 Socio-demographic and economic characteristics of respondents

Majority of respondents were mothers of the children, formally married and living together with their husbands or wives. The vast majority of respondents belonged to Sidama ethnicity and were Protestants religion followers.

Almost a quarter of respondents did not have formal education or could not read and write. Whereas, the rest of respondents had different level of education. The majority of respondents were between 25-35 years of age and had an average small family size (\leq 5 members).

7.2.2 Socio- demographic characteristics of child

The age of surveyed children ranged from 1 to 59 months with the mean age of 28.85 months (Standard deviation ± 15.83). Most children were delivered at health facilities, more than one-quarter of them were the first child of the family and most children had two years and above birth interval.

7.2.3 Prevalence of pneumonia among under-five children

The current findings reported that the prevalence of pneumonia among under-five children in Sidama Regional State is 22.2%.

7.2.4 Key factors associated with the occurrence of pneumonia among under-five children

7.2.4.1 Findings related to socio-demographic and economic factors and association with the occurrence of childhood pneumonia

According to these findings increased family size, lower level of maternal education, middle age of children (7-36 months) and below two years of birth interval were the socio-demographic and economic factors that influenced the occurrence of childhood pneumonia at the current study area.

7.2.4.2 Findings related to behavioural and disease related history of households and the occurrence of pneumonia among under-five children

Findings from multivariate analysis of the current study indicated that children who had a history of any disease during data collection were more likely to develop pneumonia than those who did not. Similarly, those children who had history of measles disease for the past one year before data collection were more likely to develop pneumonia than those without the history of measles.

Additionally, children whose family members had a history of respiratory disease were more likely to develop pneumonia than those who did not.

7.2.4.3 Findings related to environmental factors and occurrence of pneumonia among under-five children

Multivariate analysis shows that absence of household latrine, window, and separate house for animal and human being, presence of chimney in the house during data collection period and use of charcoal were environmental factors that contributed for the occurrence of childhood pneumonia in the current study area.

The study result showed that children from family without latrine were more likely to develop pneumonia than those who lived in a house with latrine. Also, absence of separate house for animals and human being; and living in households without window, presence of chimney in house and use of charcoal as the main source of fuel for cooking were significantly associated with the occurrence of pneumonia among under-five children than the cases where the reverse is true.

7.2.4.4 Findings related to nutritional and immunization related factors and occurrence of pneumonia among under-five children

The current result showed that children who were severely underweight for their age were more likely develop pneumonia than their counterparts who weigh normal considering their age. Similarly, children who were with lack of vitamin A and did not receive the required vaccines including measles preceding to data collection were likely to developed pneumonia than their counterparts who received vitamin A supplementation and vaccines required for their age. Contrary of this children who were fed with hands were less likely to develop pneumonia than those bottles were used to feeding.

7.2.5 Knowledge, attitude and practice of mothers/caregivers towards childhood pneumonia prevention and control

The current study findings reported that majority of mothers/caregivers had poor knowledge but had positive attitude and showed proper practice regarding childhood pneumonia prevention and control.

7.2.6 Factors associated with knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control

7.2.6.1 Factors associated with knowledge of mothers/caregivers towards pneumonia prevention and control among under five children

The finding showed that mothers/caregivers from households with a functional radio were more knowledgeable about childhood pneumonia prevention and control than those who did not have. Mothers/caregivers from households who had two children were knowledgeable than the others. Similarly, mothers/caregivers with better level of education had more adequate knowledge than their counterparts towards pneumonia prevention and control in under five children.

7.2.6.2 Factors associated with attitude of mothers/caregivers towards pneumonia prevention and control among under five children

The current study findings showed that mothers/caregivers with adequate knowledge and who had five and above children had positive attitude towards childhood pneumonia prevention and control than the other groups. However older age group of respondents were fund to show less positive attitude than younger age groups.

7.2.6.3 Factors associated with practice of mothers/caregivers towards pneumonia prevention and control among under five children

From the present study mothers/caregivers with fair knowledge towards pneumonia prevention and control among under five children showed proper practice than the others. Similarly, respondents' age and educational level were found to be positively associated with proper practice towards pneumonia prevention and control in under five children. Likewise, mothers/caregivers who had a functional radio in the household were positively associated with proper practice on childhood pneumonia prevention and control measures.

7.3 CONCLUSIONS

The present study was designed to develop evidence based awareness and prevention strategies on pneumonia among under five children in Sidama Regional State in line with the goal of ending pneumonia by 2025. The conceptual framework for this study was developed based on health belief model and global pneumonia prevention and control framework. A high prevalence of childhood pneumonia was obtained in a current study that pointed the need for the implementation of good strategy to curb further increase in the prevalence. The study also identified relevant sociodemographic and economic factors, behavioural and disease history related factors, environmental factors, nutritional and immunization related factors associated with the occurrence of childhood pneumonia. Gaps related to the knowledge, attitude and practice of mothers/caregivers towards pneumonia prevention and control and its associated factors were reported.

7.4 RECOMMENDATIONS

Based on the study findings obtained, the following recommendations are proposed to each of the responsible bodies in different levels.

7.4.1 Recommendations for woreda health offices, health centers and health posts

As part of the governmental administrative structures, these levels (administrative Institutes) are taken as the implementers of any health programs because they are closer to the community. Besides, these institutions, integrated with concerned partners/stakeholders, are expected to identify and address gaps seen in the community regarding study objectives.

Due to this the following recommendations are made: -

- Health extension workers should effectively implement the health extension programs and address the child health package with the participation of community to create awareness and prevent childhood pneumonia;
- Woreda health offices, health centers and health posts should effectively integrate family health packages especially birth limitation and spacing in their maternal and child health programs;
- Health centre and health posts should work on model household training to improve households with the availability of household latrine, separate house for animal and human being and general improvement of housing conditions, availing ventilation in the house and the use of improved fire wood other than charcoal;
- Health extension workers and health centers better strengthen the routine as well as outreach immunization and supplementation programs to ensure that they reach all children and create community awareness on the timely vaccination and supplementation of children;
- Under-nutrition is one of the most important factors that should be addressed to improve the health of children in the current study area. Woreda health offices, integrated with responsible bodies (multi-sectorial), should work on nutrition improvement programs and projects;

 Governmental structure should encourage and equip the community with proper health education program and monitor the implementing progress.

7.4.2 Recommendations for Sidama Regional Health Bureau

As health bureau is principal owner and monitoring institution of health system implementation at mid-level; the following recommendations are proposed: -

- Health bureau should work with woreda health offices to strengthen the implementation of health extension program effectively with addressing child health packages;
- Health bureau should integrate media and communication institutions to establish and strengthen advocacy system on pneumonia prevention and control strategies;
- Health bureau should work on strengthening health facility to provide quality health services for the community with timely monitoring and controlling approaches;
- Health bureau should integrate partners to work strongly and strategically on health worker's capacity building in order to address the identified determinants of childhood pneumonia through increased community awareness and participation.

7.4.3 Recommendations for health sector partners and professional associations

The work of health sector can be effective if partners work collaboratively and in supportive fashion with each other.

Considering this, the following recommendations are proposed for all responsible government as well as non-government health organizations:

- Government as well as non-government health sector partners and health professional associations should work in an integrated manner and harmoniously with health sectors to improve child health through capacity building and providing necessary logistics;
- All health sector partners should work collaboratively and advocate health information and promotion activities on child health to increase community awareness on pneumonia prevention and control;

7.4.4 Recommendations to the Federal Ministry of Health

Different policies and programs including implementing strategies and guidelines of health programs have been designed and developed under the Ministry of Health Ethiopia. Considering this the following recommendations are proposed: -

- Ministry of Health should monitor and evaluate the implementations of health extension programs especially those associated on the child health packages;
- Ministry of Health should coordinate and harmonize the programs and activities of child health with concerned partners to provide quality health services for the community;
- Ministry of Health should advocate the health information and promote the preventive and controlling mechanisms of childhood pneumonia.

7.4.5 **Recommendations for further researchers**

The succeeding recommendations are proposed for future researchers: -

Further researchers better consider follow up base study to recognise the seasonal variations of childhood pneumonia and associated factors, as the data for the current study were collected in point of time;
The researchers better assess facility readiness and childhood management status to strengthen the treatment part of childhood pneumonia.

7.5 CONTRIBUTIONS OF THE STUDY

The current study contributes additional supportive evidences for the body of knowledge (BOK) in the area of study. The study establishes the prevalence of childhood pneumonia and its associated factors in Sidama Regional State. The study also contributes the findings on level of knowledge, attitude and practice of mothers/caregivers and associated factors towards pneumonia prevention and control among under-five children in the current study area.

Based on findings and reviewed literature, an awareness and prevention strategies on pneumonia among under five children was developed. The developed awareness and prevention strategies will thus be used by policy developers, program designers, health program implementers and scientific researchers in the field as references and guiding principles.

As the current study was conducted at community level, the findings may help to develop and follow the community base implementing strategy approach which will also help to bring the required change within a limited time; and it will also help to strengthen community participation and involvement in core decision making process on the issue of their own health.

7.6 LIMITATIONS OF THE STUDY

Basically the current research used community based cross-sectional study design for quantitative part integrated with FGD for qualitative part. Furthermore, Delphi method was applied to collect comments and opinions of experts on development of proposed awareness and prevention strategies on pneumonia among under five children. Hence it may share the limitations of any cross-sectional and Delphi methods studies.

There might be calibration error while anthropometric measurements are taken to assess nutritional status of children.

7.7 CONCLUDING REMARKS

The chapter presents a summary of the major findings of the study, conclusions and recommendations drawn as well as the limitations of the study. The subsequent chapter will present the proposed awareness and prevention strategies on childhood pneumonia in Sidama Regional State.

Chapter 8

DEVELOPMENT OF AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER-FIVE CHILDREN IN SIDAMA REGIONAL STATE 8.1 INTRODUCTION

This chapter presents the proposed awareness and prevention strategies on childhood pneumonia in Sidama regional state, Ethiopia in line with the strategy to end preventable pneumonia disease by 2025. The chapter further details the major components of the strategies those are guiding principles of the strategies, validation mechanism, purpose and objectives of the strategies, scope of the strategies and basis for the the strategies development. The chapter also discusses the integration of the findings with the theoretical foundation, methodology, implementation strategies, monitoring and evaluation and strategies dissemination plan.

8.2 GUIDING PRINCIPLES OF THE STRATEGIES

The following guiding principles were used for the development of the proposed awareness and prevention strategies.

- Aligned with the existing policies and strategies;
- Community centered and participation;
- Inter and intra-sectoral collaboration;
- Communication;
- Monitoring and evaluation.

8.3 VALIDATION MECHANISM

The developed awareness and prevention strategies were sent to different potential experts on the area of child health and to experienced university lecturers; as well as to health leaders who have been working in different governmental or non-governmental institutions. All panelists gave their comments and feedbacks based on prepared form. This was made to ensure the feasibility, acceptability and practicability of the proposed strategies in their implementation and to check their scientific integrity.

8.4 PURPOSE AND OBJECTIVES OF THE STRATEGIES

8.4.1 Purpose of the strategies

These awareness and prevention strategies aimed to help policy and programme planners and health care leaders as well as implementers to formulate intervention measures to tackle illnesses and deaths of children due to pneumonia in Sidama Regional State.

8.4.2 Objective of the strategies

The objective of the strategies is to develop and validate awareness and prevention strategies on childhood pneumonia in order to support the implementation of ending preventable pneumonia deaths and illness at local as well as national levels.

8.5 SCOPE OF THE STRATEGIES

These strategies are intended to be used at the study area. However, they can provide some common policy issues of child health programs to fight childhood pneumonia with implications for both national and other regional states. The Federal Ministry of Health (FMOH), Regional Health Bureaus, different level health facilities, governmental and non-governmental organizations can also use these strategies for they have been proposed for implementation based on relevant and available evidences and expertise' opinions.

8.6 BASIS FOR THE DEVELOPMENT OFTHE STRATEGIES

The current study findings and reviewed relevant literatures were used used as the initial basis for the development of these awareness and prevention strategies.

Additional the conceptual framework and Delphi method were also used to develop the current awareness and prevention strategies on pneumonia among under-five children in the study area. Thus the proposed strategies were formulated based on the literatures reviewed to find the existing baseline evidence as discussed in chapter 2 of the current study, and linked to the current research findings. Furthermore, the integrated Health Belief Model (HBM) and global action plan of pneumonia prevention and control framework (GAPP) as discussed in chapter 3 of this study provided the structure for the strategies; and the Delphi method was used to strengthen the proposed strategies.

8.6.1 Delphi method

The Delphi method was also used to develop and strengthen the awareness and prevention strategies. Delphi survey is a technique for obtaining inputs(comments or suggestions) from a panel of experts about specific issue of concern; experts were consulted individually in two rounds via e-mail with a summary of the panelists' views distributed to attain agreement, (Polit 2018:546). Experts included and consulted for the development of these proposed strategies were university lecturers, health leaders and professionals. The panelists were selected from different working institutions and diverse disciplines; and were believed for

acquiring a wide range of experience and exposures in the area of study. The participants in the panel were knowledgeable on the area and helped for the decision reached and conclusions made thereafter.

Two rounds of Delphi panelist's comments and suggestions were collected from experts and the collected data were analyzed based on their importance. The following diagrammatic presentation shows the two rounds of communications and activities conducted with the Delphi panelists.

Delphi panellists.



Figure 8. 1: Diagrammatic presentation of Delphi group data collection and analysis

8.7 INTEGRATION OF THE FINDINGS WITH THE THEORETICAL FOUNDATION

The conceptual framework of the current study was developed based on the integration of HBM and global action plan of pneumonia prevention and control (GAPP) framework. The common components of Health Belief Model like, individual's perceptions, modifying factors and benefits of taking the preventive measures were considered. Under each elements of HBM; the three components of GAPP framework (protective, preventive and treatment) were also addressed to develop the new conceptual framework which is more integrative and helpful for the development of awareness and prevention strategies on childhood pneumonia in Sidama Regional State, Ethiopia.

8.7.1 Individual's perceptions

From the current study, knowledge and attitude of mothers/caregivers towards pneumonia prevention and control among under-five children were obtained as the individual's perceptions that can influence level of disease prevention and control in the community. Most of the current study respondents had poor knowledge towards pneumonia prevention and control measures. Regarding their attitude, majority of them believed that childhood pneumonia is a dangerous disease even though they had poor knowledge on the how of the prevention and control of it.

It is known that if mothers/caregivers are knowledgeable about the causes, mode of transmission, preventive measures and management of pneumonia among under-five children, they can keep their children safe from the disease or may seek medical assistances on time. **8.7.2** Modifying factors

The modifying factors under the current study include socio-demographic and economic factors; environmental and nutritional factors of childhood pneumonia; and also factors like behavioral and knowledge of mothers/caregivers towards childhood pneumonia prevention and control. From the present study, different findings were obtained as the No

contributing factors of under-five pneumonia that can be categorized under modifying factors.

Having medium and large family size, respondents without formal education, age of surveyed children, two and less than two years' of birth interval between children and number of children mothers have, absence of household latrine, living with animal(s) in the household, absence of window in the living house, using charcoal as source of fuel, children with severe under-weight, lack of any immunization, feeding with a bottle and lack of vitamin A supplementation were among the contributing factors of childhood pneumonia under this category.

8.7.3 Benefits of taking the preventive measures Perceived benefits (attitude) and taking the proper measures towards childhood pneumonia prevention and control were considered under the category of the benefits of taking the preventive measures. From the current study majority of respondents showed positive attitude towards pneumonia prevention and control; but the gap was seen on properly practicing those prevention and control mechanisms. Behavioral and health status of household members were addressed; and findings were also obtained. Having history of respiratory disease among family members, the child's past measles disease history and the child's current disease history during data collection period were found to be the associated factors for the occurrence of childhood pneumonia.

8.8 AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER-FIVE CHILDREN IN SIDAMA REGIONAL STATE

Table 8.1: Summary	of major	findings of	the research in	Sidama Regiona	al State, Ethiopia: 2020

No	Areas of gap addressed	Major findings obtained
1	Pneumonia prevalence among under-five children	Childhood pneumonia found to be 22.2% in Sidama Regional State
2	Socio-demographic and economic determinants of childhood pneumonia	 Households with medium and large family size were found as contributing factors childhood pneumonia Caregivers with no formal education were found to be associated factors for the occurrence of pneumonia Being age of 7-12months and 13-36 months were another associated factors with pneumonia prevalence Children who had birth interval less than 2 years developed pneumonia than those counterparts
3	Behavioral and health status related factors associated with childhood pneumonia	 Family members who had a history of respiratory disease found to be contributing factor for childhood pneumonia Children who had history of any disease during data collection had a higher probability of developing pneumonia than their counterpart without a history of any disease. A children who suffered from measles the preceding year before data collection likely to suffer from pneumonia than children without history of measles.
4	Environmental factors contributing the occurrence of childhood pneumonia	 The absence of household latrine was found as contributing factors of pneumonia Absence of separate house for animal and human being was associated with the occurrence of pneumonia Households using charcoal as the main source of cooking fuel was another contributing factor for the occurrence of childhood pneumonia Absence of windows in the living house found to be the contributing factors for the occurrence of pneumonia Presence of chimney inside the living house during data collection was obtained as contributing factors of pneumonia
5	Nutritional and immunization related contributing factors of childhood pneumonia	 Severe under-weight among children was found to be a contributing factors of childhood pneumonia Feeding children by hand was found to be protective factors of pneumonia Children who were not immunized with vaccines including measles were more likely to suffer from childhood pneumonia than their immunized counterparts. Lack of vitamin A supplementation within last 6 months prior to data collection was found to be the contributing factors of pneumonia
6	Knowledge of mothers/caregivers towards pneumonia prevention and control	 About 2/3 of mothers/caregivers had poor knowledge on pneumonia sign and symptom, mode of transmission, causative agents and way of prevention A quarter of mothers/caregivers had moderate/fair knowledge on pneumonia sign and symptom, mode of transmission, causative agents and way of prevention Less than 10% of mothers/caregivers had good knowledge on pneumonia sign and symptom, mode of transmission, causative agents and way of prevention

7	Factors associated with knowledge of mothers/caregivers towards pneumonia prevention and control	 Having three children was found to be factors for having adequate knowledge Mothers/caregivers who had functional radio were more knowledgeable than those did not Mothers/caregivers who had educated with certificate and above were more knowledgeable than those did not
8	Attitude of mothers/caregivers towards pneumonia prevention and control	 Less than two-thirds had positive attitude towards prevention and control of childhood pneumonia About 38.2% had negative attitude
9	Factors associated with attitude of mothers/caregivers towards pneumonia prevention and control	 Respondents with adequate knowledge showed positive attitude than those with in adequate knowledge Mothers/caregivers with age group greater than or equal 36 years had positive attitude than the others Those who had five children and more showed positive attitude than others
10	Practice of mothers/caregivers towards pneumonia prevention and control	• Above half (57.5%) of the respondents had good or proper practice towards prevention and control of childhood pneumonia
11	Factors associated with practice of mothers/caregivers towards pneumonia prevention and control	 Mothers/caregivers with good knowledge were showed proper practice than the others Mothers/caregivers with age group 25-35 years showed proper practice than the other age groups Households who had functional radios had proper practice than those who did not have Mothers/caregivers with educated level of grades 1-8 and 9-12 showed proper practice than the others

8.8.1 Development of strategies

These awareness and prevention strategies were developed primarily from the current study findings and the reviewed existing national policies, guidelines and strategies. Then strengthened by inputs from Delphi group analysis and aligning with ending preventable pneumonia death by 2025. The researcher believed that the strategies will help as baseline reference for related policy makers and programme planners, governmental and non-governmental implementers with due consideration of other available options to improve child health, preventing morbidity and mortality of children caused due to childhood pneumonia. These strategies were basically organized into seven categories and each of them further includes different focus areas as listed here below:

- 1. Health communication and promotion;
 - Community mobilization on pneumonia prevention and control;
 - Health education;
 - Advocacy;
 - Social marketing.
- 2. Protecting strategies;
 - Strengthening vitamin A supplementation;
 - Preventing malnutrition.
- 3. Preventing strategies;
 - Promoting immunization including measles;
 - Promoting good housing condition;
 - Promoting safe fuel utilization;
 - Latrine availability and utilization;
 - Managing the respiratory diseases.
- 4. Treating the sick child.

5.

- Improving health seeking behavior;
- Improving the access and utilization of case management;
- Improving the continuous of feeding.
- Improving maternal and child health;
- 6. Engaging multi-sector partnership;
- 7. Health system strengthening;

8.8.1.1 Strategy one: Health communication and promotion

Increasing knowledge, attitudes and practices of mothers or caregivers towards pneumonia prevention and control in Sidama Regional State should be taken as vital and that needs continuous and effective health communication and promotion work. Communication is the method of distributing the information, ideas, knowledge and experience among individuals or groups by using different communication channels; whereas health promotion can be taken as it is a combination of educational and environmental supports for interventions and situations of living in a suitable environment for health (Meseret and Yihenew 2004:5).

Health promotion is concerned with improving health through seeking improved lifestyles, health services and environments which are not merely restricted to the physical environment but includes cultural and socioeconomic status that substantially determine health conditions. The effective health promotion can be done through community mobilization, health education, advocacy and social marketing, (WHO 2012:15). Update and/or develop IEC/BCC materials and related tools to promote use of high impact MNCH services as per the national health communication material development and implementation guidelines. Community empowerment and demand creation can be done through conducting regular home to home visits to deliver continous counseling on high influence about reproductive health, maternal-newborn and child survival interventions and care as needed through strengthening 1-to-5 HDA network by training network leaders on key MNCH behaviors and practices for awareness creation to support issues related to maternal newborn and child survival, conduct regular mass media activities (FMOH 2015:46).

8.8.1.1.1 Community mobilization

Community mobilization involves convincing the community members to attend or engaged in any activity arranged by health educator to improve awareness on a specific issue at the public level (Sharma and Romas 2012:32). As research findings obtained from the current study indicate, community should be mobilized and guided to be aware of childhood pneumonia prevention and control mechanisms. Health sector integrated with partners can select the suitable approaches and arrange the community mobilization to improve the knowledge of mothers/caregivers those with children under-five age in Sidama Regional State towards pneumonia prevention and control.

For community mobilization activities like organizing a talk in the community, arranging a health fair and bringing together key community leaders for a panel discussion can be used, (Sharma and Romas2012:32). And also it would be useful to follow the local base realistic approaches to screen the participants for the mobilization at the study area.

8.8.1.1.2 Health education

To enhance awareness of community in Sidama Regional State on childhood pneumonia prevention and control strategies, proper health education should be designed and given. Majority of the study respondents had poor knowledge towards the causes and prevention methods of under-five pneumonia. This requires special attention to improve. Health education is a communication of activities aimed with enhancing good health and through averting or reducing of ill-health among individuals and groups by influencing the beliefs, attitudes and behavior of those with influence and of the community at large (Sharma and Romas 2012:18).

According to WHO, health education is not restricted to the distribution of only health-related information, rather it is also an activity to influence the motivation, skills and confidence that necessary to take an action to improve health, as well as the communication of information about the underlying social, economic and environmental conditions those impacting on health. Likewise health education is essential to communicate about the individual risk factors and behaviors to use the health care system (WHO 2012:13). The researcher proposed that health sector and the respected responsible bodies have to provide effective health education to the community in Sidama Regional State utilizing the proper health education materials and approaches considering cultural issues and norms.

8.8.1.1.3 Advocacy

Advocacy has been planned and considered under the current childhood pneumonia awareness and prevention strategies to assure it as prospect. Advocacy can be the communication strategies especially focusing on policy makers and planners, community and opinion leaders to be engaged and attain commitment and support. Moreover it helps for the higher-level commitment and participation in fulfilling a set program agenda, (Meseret and Yihenew 2004:6). Advocacy in health is about shifting a public opinion and mobilizing essential resources to support any issue or policy that affects the health of a community (Sharma and Romas 2012:242). Global and national advocacy campaigns are needed to improve the investments those help to end preventable child pneumonia deaths by 2030 through increasing awareness among all stakeholders about mortality of child due to pneumonia (Every Breath Counts 2017:2). **8.8.1.1.4 Social marketing**

Social marketing is the use of marketable marketing techniques to help in attaining of a behavior that is advantageous for the health of a target population (Sharma and Romas 2012:250). It is proposed that social marketing is one of the vital concepts to be used to increase the awareness of community towards childhood pneumonia prevention and control mechanisms.

8.8.1.2 Strategy two: Protecting children from being diseased from pneumonia through adequate nutrition

Protecting children from childhood pneumonia by creating and promoting a good health practices is one of the key areas of pneumonia prevention and control proposed under the strategy of GAPP, (WHO and UNICEF 2013:7). The responsible bodies should work on critical services that may support the community to protect their children from pneumonia. The GAPPD identified that it is possible to protect children from pneumonia by establishing and sustaining a good health practices since starting from birth and by taking proper actions on activities like exclusive beast feeding for the first 6 months, providing adequate complementary feeding and timely supplementation of vitamin A (WHO and UNICEF 2013:6). The researcher identified that lack of vitamin A supplementation can contribute for the occurrence of childhood pneumonia. According to Farhad and Taheren (2014:57), vitamin A deficiency was one of the significantly associated factors with severe childhood infections.

In addition severe malnutrition was also found to be among the associated factors with childhood pneumonia in the current study area. It is known that poor knowledge and practices on food security and complimentary feeding, as well as exclusive breastfeeding can cause the children to be malnourished. A long term regulation and balance of energy in life can be attained through critical periods of life development and basically through the age of intrauterine life, infancy and childhood. For the prevention of malnutrition it is important to promote a healthy and balanced diet from early childhood (WHO 2016:2). Adequate and balanced nutrition during infancy and early childhood is one of the essential for the growth, health and development of children with their full potential. In contrary poor nutrition or malnutrition can agravate the risk of illness and could be responsible, for the mortality of children in many areas directly or indirectly (WHO 2009:3).

Community should have access for information on the different nutritional factors and timely supplementation of vitamin A; these could influence the occurrence of childhood pneumonia in the study area.

8.8.1.3 Strategy three: Preventing children from being diseased by pneumonia through timely immunization and healthy environment

The integrated global action plan for the prevention and control of pneumonia (GAPP) proposed that it is possible to prevent childhood pneumonia through ensuring world-wide coverage of immunization, HIV prevention and by availing healthy environments (WHO 2009:4). Result of the current study also showed that lack of immunization, absence of latrine, lack of separate house for animals and human being, absence of windows, presence of chimney in the house during the time of data collection, using charcoal as main source of fuel and presence of disease history among children as well as family member contributed for the occurrence of pneumonia among under five children. Community should be sensitized on the benefit of timely vaccination of their children, good housing conditions, safe use of fuel and proper health care seeking for family members when they show any sign and symptom of respiratory diseases. The later action should also be taken if a child is sick of any illnesses- including measles.

The Global Vaccine Action Plan (GVAP), recommended by the World Health Organization meeting in 2012, calls for delivery of vaccines as part of the package of interventions for diseases control including childhood pneumonia (WHO and UNICEF 2013:12). Health sector should ensure the accessibility of vaccine for the eligible children along with effective vaccine management.

The GAPP proposed that reducing indoor air pollution and proper sanitation and hygiene can protect the children from pneumonia (WHO and UNICEF 2009:4). It was reported that minimizing household air pollution through uzing improved stoves can reduce severe childhood

pneumonia (WHO and UNICEF 2013:14). Evidence from the previous study conducted by Gedefaw, Mamo and Getahun (2014:150) showed that the use of charcoal as main source of fuel, mothers carrying a child on back during cooking, rearing of cattle inside the main house and living in crowded house were the factors those significantly contribute the occurrence of childhood pneumonia.

Healthy housing conditions, proper hygiene and sanitation are essential for maintaining good health of a community. Poor housing conditions can expose people for several risks factors of health including indoor air pollution that may cause respiratory problems, crowded housing which rises the risk of exposure to communicable diseases; and insufficient water supply and sanitation facilities that affect food safety and personal hygiene (WHO 2018:2-4).

Community can exercise good health for their own as well as for their children if they have been empowered and familiarized with the preventable challenges of childhood pneumonia. The current awareness and prevention strategies can substansially support in ways that they can be familiarized with the issues of good housing conditions, proper hygiene and sanitation, safe fuel utilization and timely initiation of childhood immunization.

8.8.1.4 Strategy four: Treating children who become ill with pneumonia

Enhancing the community awareness creation approaches may help for timely health seeking and treating of sick child with pneumonia. The current study findings reported that 42.5% of study respondents showed poor practice towards childhood pneumonia prevention and control including timely treating of the sick child. One of the focus areas of GAPP is the treatment of children who are ill from pneumonia with proper treatment. Increased access to proper care, through community based case management of pneumonia, can reduce 35% mortality in children, (WHO & UNICEF 2013:13-15). World Health Organization has recommended that case management is one of the critical pillar strategies of childhood pneumonia control. It involves classifying the severity of disease by using clinical signs and symptoms like fast breathing, chest in-drawing and presence of general danger signs. Likewise it also guides the professionals to apply appropriate treatment which could be home care advice, antibiotics for home therapy and or referral to a higher-level health facility (WHO 2014:4).

8.8.1.5 Strategy five: Improving the utilization of maternal and child health services

Quality and skill care for woman and newborn during and after delivery like all-inclusive family planning and integrating care for HIV/AIDS (PMTCT), malaria and others were among services that can contribute to the prevention and control of pneumonia, (WHO & UNICEF 2013:12). From the current study, findings showed that having large family size, birth interval of less than two years and age of children between 7-36 months were found to be significantly contributing to the occurrence of childhood pneumonia in Sidama Regional State. The accessing and utilizing of maternal health services within the right time has been proposed and community should be sensitized about the integration and utilization of maternal health services and its contribution on the decrease of childhood pneumonia disease and death.

Provision of modern contraceptives to all woman who want to avoid pregnancy; and continuum of care through adolescence, pregnancy, childbirth, postnatal and newborn periods and into childhood based on WHO standards can support to achieve the goal of improving child health. Delivering good quality of care at child birth, timely immunization, initiation of breastfeeding and proper nutrition can help to improve child health (Every woman Every Child 2015:19; FMOH 2015:5-6).



Figure 8. 2: Continuum of care approach: time of care giving and place and approach of care giving (FMOH 2015:36).

8.8.1.6 Strategy six: Engaging multi-sector partnership to facilitate and contribute pneumonia prevention and control

Engaging and involving different partners and sectors can galvanize support to put pneumonia under control. Areas of partnership can be overall child health and nutrition in societies and the wider communicable disease control, ensuring clean air, safe water supply, sanitation and hygiene, and development of financing communities. The possible engaged partners can be governmental, private sectors and nongovernmental organization (Save the Children, UNICEF and Every Breath Counts 2019:7-8).

The Ethiopian health care delivery system is enhanced by the rapid growth of private-for- profit and NGOs sector that have been playing crucial role in expanding the health service coverage and utilization. To catalyze the envolvement of the private sector in health service delivery and increase partnership between the public and private sectors, the FMOH launched the public private partnership (PPP) strategy and supposed to bring significant change (FMOH 2015:18).

Health sector in Sidama Regional State should collaborate with woman, child and youth, education and agriculture sectors to create awareness and exchange experiences. The current study found out that being mothers/caregivers with no formal education to be one of the factors that contribute to childhood pneumonia in Sidama Regional State. These new strategies advise that both inter and intra sector collaboration can help for the proper and timely combat of childhood pneumonia.

8.8.1.7 Strategy seven: Strengthening health system to deliver strong health care in order to combat pneumonia

Health system consists of those organizations, products, manpower and actions whose primary aim is to promote, restore and/or maintain health. The progress towards universal health coverage and improve the quality of primary health care services need the expansion of the coverage of quality and essential health services with an emphasis reducing unfairness and strengthening health care facilities. Likewise the strong and impartial health systems are needed to effectively prevent, diagnose and treat pneumonia and deliver children with their fundamental human right with **SYSTEM BUILDING BLOCKS**

good-quality healthcare (Save the children, UNICEF & Every Breath Counts 2019:3, FMOH 2015:17-8).

Health systems can be strengthened through improving one or more of the functions of the health systems that leads to access better health service coverage with improved quality and or efficiency (USAID & Abt associate 2015:4). World health organization proposed six possible building blocks to strengthen the health systems. These are ensuring good leadership and governance, good health service delivery, a well performing and competant health workforce, a well-functioning health information system, equitable access to essential medical products, vaccines, technologies and a good health financing, (Anwer 2017:21-3 and WHO 2007:3).

Within the strengthened health systems, neonatal and prenatal mortality can be lowered through availing improved health services and training of health workers. Similarly, under-five mortality can also be decreased by empowering the communities and provide engagement through strengthening of health services in communities, (USAID & Abt associate 2015: ix-x).

OVERALL GOALS / OUTCOMES



Figure 8. 3: The WHO health system framework/the six building blocks of health system (WHO 2007:3).

8.9 CONCEPTUAL FRAMEWORK FOR AWARENESS AND PREVENTION STRATEGIES OF CHILDHOOD PNEUMONIA

The conceptual framework has been developed for the implementation of proposed awareness and prevention strategies of pneumonia among under-five children in Sidama Regional State aligning with GAPP framework and also by using current study findings and the existing relevant literatures that are also aligned with the currently proposed strategies.



Figure 8. 4: Conceptual framework developed for implementation of proposed awareness and prevention strategies on childhood pneumonia in Sidama Regional State: 2020

8.9 **PRIORITY AREAS FOR ACTION**

The results of the current study indicated that there are some areas of interventions that can support on the effort of pneumonia prevention and control. The selected priority areas are summarized as below in table 8.2.

Table 8.2: Priority areas of action of awareness creation and prevention strategies on pneumonia prevention and control in Sidama Regional State: 2020

Ser. No Areas of action Purpose Activities Outcome	No. Areas of action Durnosa Activities Outcome	
--	--	--

1	Addressing determinant factors of childhood pneumonia	Creating awareness and empowering the community	•	Educating those illiterate community members Awareness creation on MNCH Address different socio-demographic and economic factors Awareness creation on good housing conditions, safe fuel utilization, hygiene and sanitation	Increased use and implementation of maternal, neonatal and child health services which might indirectly improve child health
2	Improve the KAP of mothers/ caregivers towards childhood pneumonia prevention and control	Increasing the level of KAP among mothers/ caregivers towards in the community towards pneumonia prevention and control	•	Health communication and promotion on pneumonia prevention and control through ✓ Health education ✓ Advocacy ✓ Social mobilization ✓ Social marketing	Improved level of KAP among mothers/ caregivers towards in the community towards pneumonia prevention and control
3	Improve health seeking behavior in the community	Prevent and treat possible health outcomes	•	Aware the community to seek early health care for different health problems Community awareness to immunize their children Treat sick child including pneumonia	Increased awareness and practice of community towards early health seeking
4	Nutritional intervention to child	Prevent malnutrition and improve child health	•	Community awareness on proper child feeding, practices including exclusive and complimentary breast feeding Awareness on vitamin A supplementation Timely deworming the children	Well-nourished children with enabled disease preventing immunity
5	Hygiene and sanitation practice	Improve hygiene and sanitation	•	Awareness on latrine utilization Personal hygiene promotion Feeding practice with the consideration of good hygiene and sanitation Awareness on waste management	Improved hygiene and sanitation practice in the community and supported to prevent disease
6	Demand creation and community participation	Improve community participation on the decision of health issue	•	Capacity building of community HAD training Community mobilization	Demand created in the community with improved participation on the health issues
7	Health system strengthen	Improve health service delivery	•	Consider manpower Avail health logistics Provide quality and equity health service	Improved health service delivery to all community
8	Demand creation and capacity building of health staffs	Improve capacity of health work force	•	Sensitization Training Supportive supervision Mentorship	Improved capacity building within health workforce

8.10 MONITORING AND EVALUATION

The progressive efforts to prevent and control childhood pneumonia need integrated attention. World Health Organization recommends that countries should have primary responsibility for ensuring that data are available and used for action, and for measuring progress (WHO and UNICEF 2013:31).

Only having strategies may not bring the needed changes in awareness creation and prevention for childhood pneumonia. Rather it is essential to establish and availe the monitoring and evaluation systems in place. Current strategies also proposed that putting a strong monitoring and evaluation system in place will be the cornerstone for the success of planned strategies. Monitoring and evaluation is taken as the foundation for national health sector strategic planning as it covers all major disease programs and health systems activities. Thus, the following leading GAPP indicators are adapted and modified to be used to monitor and evaluate the improvement of awareness and prevention strategies on childhood pneumonia. (WHO and UNICEF 2013:32-3 & WHO 2009:6&10).

Table 8.3: Indicators to evaluate progress in achieving key operational targets for awareness and prevention strategies on childhood pneumonia

Indicators	Definitions	Interventions category
Immunization coverage/fully immunized	Percentage of 12-23 months children who received all doses of vaccines	Prevent
Measles immunization coverage	Percentage of 12-23 months children immunized with measles	Prevent
Penta3 immunization coverage	Percentage of 12-23 months children who received 3 doses of pentavalent vaccines	Prevent
PCV10 immunization coverage	Percentage of 12-23 months children who received 3 doses of PCV10	Prevent
Exclusive breast feeding for 6 months	Percentage of 0-5 months infants who are exclusively breastfed	Protect
Continued breast feeding at 1 year	Proportion of 12–15 months children who fed breast milk	Treat
Complementary feeding	Percentage of 6-23 months children who received a minimum acceptable diet	Protect
Vitamin A supplementation coverage	Percentage of 6 -59 months children who received vitamin A 2 annual doses	Protect
Care seeking for pneumonia	Percentage of 0-59 months children with suspected pneumonia taken to an appropriate health provider	Treat
Antibiotic treatment for pneumonia	Percentage of 0-59 months children with suspected pneumonia receiving appropriate antibiotics	Treat
Hand washing with soap	Percentage of households and health care facilities with soap, water and a hand washing facility	Prevent
Hygienic sanitation facility/latrine, waste management facilities/	Percentage of households with a hygienic sanitation facilities	Prevent
Household air pollution	Percentage of households using clean fuels for cooking	Prevent
Antiretroviral prophylaxis among HIV- positive pregnant women to prevent vertical transmission of HIV	Percentage of HIV-infected pregnant women who received antiretroviral to reduce the risk of mother- to-child transmission	Prevent
Reduced number of smokers among family members	Percentage of smokers among family members	Prevent
Community awareness about the prevention and control of childhood pneumonia	Number of men and women who are aware of the methods of childhood pneumonia prevention and control	Prevent
Health seeking behavior	Percentage of households seeking health care for respiratory related issues	Treat
Having well-ventilated and separate house for animal and human being	Percentage of households with well-ventilated and separate houses for animal and human being	Prevent
Availability of information on childhood pneumonia	Percentage of community members who have information on childhood pneumonia	Prevent

Source: GAPPD framework WHO/UNICEF 2013

Monitoring is the

systematic design of information collection

on an ongoing basis to aid in improving the

implementation or operational process for

delivering a quality service or a product

Monitoring is the

systematic design of information collection

on an ongoing basis to aid in improving the implementation or operational process for delivering a quality service or a product Monitoring is the systematic design of information collection on an ongoing basis to aid in improving the implementation or operational process for delivering a quality service or a product.

Monitoring is the

systematic design of information collection

on an ongoing basis to aid in improving the

implementation or operational process for

delivering a quality service or a product.

Monitoring is the

systematic design of information collection

on an ongoing basis to aid in improving the

implementation or operational process for

delivering a quality service or a product.8.11 IMPLIMENTATION STRATEGIES

The main aim of developing the 'Awareness and prevention strategies on pneumonia among under-five children in Sidama regional state' is to help policy makers, programme planners and implementers formulate respected actions to reduce child morbidity and mortality due to pneumonia through increased community awareness on the focus of strategies of childhood pneumonia prevention and control.

8.12 STRATEGIES DISSEMINATION PLAN

The strategies will be published in a peer reviewed, reputable, scientific and open access journal and on the UNISA library website for ease of access. All stakeholders involved in the health sector will have unrestricted access to these strategies. The dissemination will also be through presentations at health seminars, workshops, and national and international conferences.

8.13 CONCLUSIONS

This chapter presented proposed awareness and prevention strategies on prevention and control of pneumonia among under-five children in Sidama Regional State, Ethiopia. The study brings together evidences from the current study, reviewed literatures, existing policies, guidelines and strategies. The strategies have been developed aligning with the goal of GAPP, and are hoped to contribute as reference to achieve the goal of ending preventable death of children due to pneumonia. The respective bodies under FMOH of Ethiopia, especially Sidama regional health bureau including woreda health offices and health facilities are advised to use these strategies as additional resources to formulate appropriate measures, interventions and actions in order to reduce childhood morbidity and mortality due to pneumonia.

References

- 1. Abel FD, Yigzaw K, & Birhanu Z. 2014. Determinants of pneumonia in children aged two months to five years in urban areas of Oromia zone, Amhara region, Ethiopia. *Open Access Library Journal* 1(e1044): 1-10.
- Almamy M, Kante, Hialy R, Gutierrez, Anna M, Larsen, & et al. 2015. Childhood illness prevalence and health seeking behavior patterns in rural Tanzania. *BMC Public Health* 15(951): 1-12.
- Anirudha VM & Vaishali VR. 2017. Study to assess the knowledge, attitude, and practice about acute respiratory infections among school going children and their parents in rural Maharashtra. *International Journal Medical Science Public Health* 6(11): 1584-1587.
- Anurag M, Monika A, Jai VS, Vijay KS & Rajesh S. 2019. Caregiver's knowledge about childhood pneumonia: A study

from rural areas and urban slums of Lucknow National Journal of Community Medicine 10(2): 62-68.

- Anwer A, Eva S, David H, & Lisa M. 2017. Health Systems Strengthening, Monitoring, Evaluation, and Learning Guide: 1-94.
- 6. Badran IG. 1995. Knowledge, attitude and practice: the three pillars of excellence and wisdom: a place in the medical profession. *WHO*, *Eastern Mediterranean Health Journal*, 1(1).
- Batu L, Kenbon S & Daniel A. 2019. Prevalence of community acquired pneumonia among children 2 to 59 months old and its associated factors in Munesa District, Arsi Zone, Oromia region, Ethiopia *Clinics in Mother and Child Health* 16(5): 1-8.
- 8. Bernadeta D & Max R. 2019. Our world in data pneumonia. *Our World in Data:* 1-26.
- Bhutta ZA, Das JK, Walker N, Rizvia A, Campbell HIR. 2013. Interventions to address deaths from childhood pneumonia and diarrhoea equitably: what works and at what cost. *The Lancet, London, England* 381(9875): 1417-29.
- Breanne M, Adriana T, Zulma V, Rueda, Lazaro V & Yoav K. 2017. Atypical bacterial pneumonia in the HIV-infected population. *Bio Med Central* 9(12): 1-7.
- 11. Bryan S, Agbessi A, Nathan P, Miller, Jennifer B & Pamela J, Surkan. 2017. A qualitative exploration of care-seeking pathways for sick children in the rural Oromia region of Ethiopia. *BMC Health Service Research* 17(184): 1-12.
- 12. Child EW. 2015. The global strategy for women's, children's and adolescent's health (2016-2030).
- 13. Daniel G, Fasil T, & Haimanot E. 2016. Determinants of community acquired pneumonia among children in Kersa district, southwest Ethiopia: Facility based case control study. *Journal of Pediatrics and Neonatal Care* 5(2):1-8.
- David AM, Lilu, Ting Shi, Yue Chu, Craig Reed, John Burrows & Et al. 2019. Global, regional, and national estimates of pneumonia morbidity and mortality in children younger than 5 years between 2000 and 2015: A systematic analysis. *Lancet Glob Health*, 7: 47-57.
- Denise F, Polit & Cheryl Tatano Beck. 2010. Essentials of nursing research: Appraising evidence for nursing practice, 6th edition. Philadelphia, Wolters Kluwer.
- Denise F, Polit & Cheryl Tatano Beck. 2018. Essentials of nursing research: Appraising evidence for nursing practice, 7th edition. Philadelphia, Wolters Kluwer.
- 17. Development Iniatives, Justifications work good & World pneumonia day. 2018.The missing piece: Why continued neglect of pneumonia threatens the achievement of health goals. New York.
- Diana M & Heather JZ. 2019. Childhood pneumonia in lowand-middle-income countries: An update. *Paediatric Respiratory Reviews*, 32(2019): 3-9.
- Douglas T, & Chessa KL. 2017. The role of breastfeeding protection, promotion and support in a developing world. Nutrition and Health in a Developing World, 10 (1007): 361-382.
- 20. Ethiopia demographic and health survey (EDHS). 2016: Key indicators report. *Addis Ababa, Ethiopia, and Rockville, Maryland, USA: CSA and ICF.* Addis Ababa, Ethiopia: Central Statistical Agency (CSA), Ethiopia.
- 21. Eduardo JDA, Fonseca L, Maria JG, Maria DF, Maria IL, George HC, et al. 2016. Risk factors for community-acquired pneumonia in children under five years of age in the post-pneumococcal conjugate vaccine era in Brazil: a case control study. *BMC Pediatrics*, 16(157): 1-9.

- 22. Elizabeth WN, Florence NO, & John O. 2018. Cross sectional survey of care seeking for acute respiratory illness in children under 5 years in rural Kenya. *American Journal of Pediatrics*, 4(3): 69-79.
- 23. Elvis E, Tarkang and & Francis BZ. 2015. Application of the Health Belief Model (HBM) in HIV prevention: A literature review. *Central African Journal of Public Health*, 1(1): 1-8
- 24. Enemona EA, Winifred E, Michael C & Revati P. 2019. Understanding the effect of indoor air pollution on pneumonia in children under 5 in low- and middle-income countries: a systematic review of evidence. *Environmental Science and Pollution Research*, 26: 3208–3225.
- 25. Every breathcounts. 2017. Every breath counts: Coalition a public-private partnership to support national governments to end child pneumonia deaths by 2030: 1-14.
- Farhad Hepdarian and Tahereh Ansarinezhad. 2014. Childhood pneumonia and vitamin A. *Reviews in Clinical Medicine*, 1(2): 57-60.
- Farzana Fedous, Fahmida Dil Farzana, Shanawaz Ahmed, Sumon Kumar Das, Mohammad Abdul Malek & Juid Das. 2014. Mothers' perception and healthcare seeking behavior of pneumonia children in rural Bangladesh. *ISRN Family Medicine*, 2014: 1-8.
- 28. Federal Ministry of Healt. 1993. Health policy of the transitional government of Ethiopia. : Health, E. F. M. O. (ed.). Addis Ababa:
- Federal Ministry of Health. 2010. Federal Democratic Republic of Ethiopia Ministry of Health: Health Sector Development Program IV: 2010/11 – 2014/15. *In:* HEALTH, E. M. O. (ed.). Addis Ababa: Federal Democratic Republic of Ethiopia Ministry of Health.
- FMOH .2015a. Health Sector Transformational Plan (HSTP). Addis Ababa: The Federal Democratic Republic of Ethiopia Ministry of Health.
- FMOH .2015b. National strategy for newborn and child survival in Ethiopia 2015/16-2019/20.
- 32. FMOH 2015c. Report Ethiopian Federal Ministry of Ministry of Health. 16.
- 33. Francis Adjei, Osei, Kofi Akohene Mensah, Daniel Asong, Peter Agyel, Baffour Sandra Kwarteng Owusu, Nicholas Karikari Mensah & et al. 2018. Prevalence of pneumonia and risk factors of pneumonia mortality among children under five years in Komfo Anokye Teaching Hospital, Kumasi, Ghana. *African Journal of Current Medical Research*, 10(28):1-6.
- 34. Gasim Omer Elkhailifa Abd-Ekfaraga and Charles O.C. Langoya ;2016. Household air pollution and childhood pneumonia in South Sudan: will clean cooking stoves reduce the incidence and mortality?. *South Sudan Medical Journal* 9(2): 36-39.
- 35. Gedefaw Abeje Fekadu, Mamao Wubeshet Trefe & Getahun Asress Alemie. 2014. Prevalence of pneumonia among underfive children in Este Town and the surrounding rural Kebeles, Northwest Ethiopia. *Science Journal of Public Health*, 2(3): 150-155.
- Ghimire M, Bhattacharya SK & JP, N. 2012. Pneumonia in South-East Asia region: public health perspective. *Indian J Med Res*, 135: 459-68.
- Hasan Ashraf, Mohammod Jobayer Chisti & Nur Haque Alam. 2014. Treatment of childhood pneumonia in developing countries *Clinical Sciences Division, International Centre for Diarrhoeal Disease Research*, 59-88.
- Hemagir K, Sameena ARB, Aravind K, Wahid Khan & Vasanta SC. 2014. Risk factors for severe pneumonia in under five

children, A hospital based study International Journal of Research and Health Science: 2 (1): 47-57.

- Hisato Kosai, Raita Tamaki, Mayuko Saito, Kentaro Tohma, Portia Parian Alday, Tan, A. G. & et al. 2015. Incidence and risk factors of childhood pneumonia-Like Episodes in Biliran Island, Philippines—A Community-Based Study. *PLOS ONE*, 10(1371): 1-9.
- 40. Ikenna K. NDU, Uchenna Ekwochi, Chidiebere D. I. Osuuorah, Kenechi S, Onah, Ejike Obuoha, Odutola I. Odetunde & Et al. 2015. Danger signs of childhood pneumonia: caregiver awareness and care seeking behavior in a developing Country. International Journal of Pediatrics, 2015(167261): 1-7.
- 41. Internal vaccine access center (IVAC). 2016. Pneumonia & diarrhea progress report: reaching goals through action and innovation. *In:* Edition, S. (ed.). Jhons Hopkins, Bloomberg School Of Public health.
- 42. International Vaccine Access Center (IVAC). 2018. Pneumonia and diarrhea progress report. *In:* Health, J. H. B. S. O. P. (ed.). Johns Hopkins Bloomberg.
- 43. James Samwel Ngocho, Plus Gerald Horumpenda, Marien Isaak DE Jonged & Blandina Theophil. 2020. Inappropriate treatment of community-acquired pneumonia among children under five years of age in Tanzania. *International Journal of Infectious Diseases*, 93(2020): 56-61.
- 44. Jayashree Gothanlar, Prakash Doke, Girish Dhumale, Prasad Pore, Sanjay Lalwani, Sanjay Quraishi & et al. 2018. Reported incidence and risk factors of childhood pneumonia in India: a community-based cross-sectional study. *BMC Public Health* 18(1111): 1-11.
- 45. Jean Modeste Harerimana, Leatitia Nyirazinyoye, Dana R. Thomson & Joseph Ntaganira. 2016. Social, economic and environmental risk factors for acute lower respiratory infections among children under five years of age in Rwanda. *Archives of Public Health*, 74(19): 1-7.
- 46. Jena M. 2014. Effectiveness of information booklet on knowledge & practice about prevention of pneumonia among mothers of under five children. *IOSR Journal of Nursing and Health Science (IOSR-JNHS)*, 3(1): 25-30.
- 47. Jenkins, C. D. 2003. Building Better health a handbook of behavioral change, Washington, D.C. 20037, U.S.A., World health organization.
- 48. Jennifer R. Gray, Susank K. Grove & Suzanne Sutherland. 2017. The practice of nursing research, appraisal, synthesis, and generation of evidence.
- 49. Jhon W.Creswell. 2014. Research design: qualitative, quantitative and mixed methods approaches, SAGE publications, Inc.
- 50. Johnson WBR, & Abdulkarim AA. 2013. Childhood pneumonia in developing countries. *African Journal of Respiratory Medicine*, 8(2), 4-9.
- Kapil Goel, Sartaj Ahmad, Gagan Agarwal, Parul Goel, Viljay Kumar, Tachbelle, E. & AL, E. 2012. A cross sectional study on prevalence of acute respiratory infections (ARI) in underfive children of Meerut District, India. J Community Med Health Educ 2: 176.
- 52. Karen Glanz & Donald B. Bishop. 2010. The role of behavioral science theory in development and implementation of public health Interventions. *Annual reviews on Public Health*, 2010(31) 399–418.
- 53. Kothari CR. 2004. Research methodology: methods and techniques, New Delhi, New Age International.
- 54. Krishna Kumar Yadav & Shally Awasthi. 2016. The current status of community-acquired pneumonia management and

prevention in children under 5 years of age in India: a review. *Therapeutic Advances in Infectious Disease* 3(4): 83-97.

- 55. Kumar, R. 2011. *Research Methodology, step by step guide for beginners*, SAGE Publications Ltd.
- 56. Laura M Lamberti, Irena Zakarija-Grkovic, Christal L Fischer Walker, Evropi Theodoratou, Harish Nair, Harry Campbell & AL, E. 2013. Breastfeeding for reducing the risk of pneumonia morbidity and mortality in children under two: a systematic literature review and meta-analysis. *BMC Public Health*, 13(18): 1-8.
- 57. Lema KT, Murugan R, Tachbele E & Negussie BB. 2018. Prevalence and associated factors of pneumonia among underfive children at public hospitals in Jimma zone, South West of Ethiopia. *J Pulmonol Clin Res* 2(1): 25-31.
- Li Liu, Shefali Oza, Dan Hogan, Yue Chu, Jamie Perin, Jun Zhu & et al. 2016. Global, regional, and national causes of under-5 mortality in 2000–15: an updated systematic analysis with implications for the Sustainable Development Goals. *Lancet* 388: 3027–35.
- 59. Lyn Haskins, Merridy Grant, Sifiso Phakathi, Aurene Wildford, Ngcwalisa Jama, Christiane Horwood & AL, E. 2017. Insights into health care seeking behaviour for children in communities in KwaZulu-Natal, South Africa. *African Journal of Primary Health Care and Family Medicine*, 9(1): 1-9.
- 60. Maha Fahad Alluqmani, Abdulrahman Abdullah Aloufi, Amnah Makk Al Abdulwahab, Abdullah Ismail Khalil Alsharif, Abdullah Abdulaziz Ashathri, Alshehri6, M. S. & AL, E. 2017. Knowledge, attitude and practice of mothers on acute respiratory infection in children under five years in Saudi Arabia. *The Egyptian Journal of Hospital Medicine*, 69(2): 1959-1963.
- 61. Manoj Sharma & John A. Romas. 2012. Theoretical foundations of health Education and Health Promotion, Canada, Jones & Bartlett Learning, second edition.
- 62. Mark Corner & Paul Norman. 2015. Predicting and changing health behaviour research and practice with social cognition models, open University press.
- 63. Martin Kolditz, Falko Tesch, Luise Mocke, Gert Hoffken, Santiago Ewig & Jochen Schmitt. 2016. Burden and risk factors of ambulatory or hospitalized CAP: A population based cohort study. *Respiratory Medicine*, 121: 32-38.
- Maximilian Muenchhoff & Philip J. R. Goulder. 2014. Sex differences in pediatric infectious diseases. *The Journal of Infectious Diseases*, 209(3): 120-126.
- 65. Mercy Abbey, Margaret A, Chinbuah, Margaret Gyapong, L. Kay Bartholomew & Bart Van Den Borne. 2016. Community perceptions and practices of treatment seeking for childhood pneumonia: a mixed methods study in a rural district, Ghana. *BMC Public Health* 16(848): 1-10.
- 66. Meseret Yazachew & Yihenew Alem. 2004. Introduction to health E=education, Ethiopia public health training initiative (EPHTI).
- 67. Monir Ramezani, Seyedeh Zahra Aemmi & Zahra Emami Moghadam. 2015. Factors affecting the rate of pediatric pneumonia in developing countries: a review and literature study. *International Journal of Pediatrics* 3(6): 1173-1181.
- Morte, W. W. L. 2019. Behavioral change models, Boston, Boston University school of public health.
- 69. Musa, A. I. 2013. Understanding the intersections of paradigm, Meta-Theory, and Theory in library and information science research: A social constructionist perspective *Samaru Journal of Information Studies*, 13(2): 41-48.
- 70. Nabanita Nirmolia, Tulika G. Mahanta, Manjit Boruah, Reeta Rasaily, Ritap Kotoky & Rituparna Bora. 2017. Prevalence and

risk factors of pneumonia in under five children living in slums of Dibrugarh town. *Clinical Epidemiology and Global Health*, 6(2018): 1-4.

- Nancy K. Janz & Marshall H. Becker. 1984. The Health Belief Model: A Decade Later. *Health Education Quartely*, 11: 1-47.
- 72. Nathan P. Miller, Agbessi Amouzou, Mengistu Tafesse, Elizabeth Hazel, Hailemariam Legesse, Tedbabe Degifie & et al. 2014. Integrated community case management of childhood illness in Ethiopia: Implementation strength and quality of care. *American Society of Tropical Medicine and Hygiene*, 91(2): 424–434.
- 73. Neerupam Gupta & Naine Bhadrala. 2019. Risk factors for acute severe pneumonia in under five children, India. *International Journal of Contemporary Pediatrics Gupta N et al. Int J Contemp Pediat*, 6(3): 949-954.
- 74. Nicholas Walliman. 2011. Research methods: The basics, London and NewYork, Routledg.
- 75. Nirmolia, N Mahantab, T. G, Boruaha, M., Reeta Rasailyc, Kotokyd, R. P. & Borae, R. 2018. Prevalence and risk factors of pneumonia in under five children living in slums of Dibrugarh town. *Clinical Epidemiology and Global Health*, 6: 1-4.
- 76. Noha Fadl, Yasmine Y, Muhammed & Ayat Ashour. 2020. Medical management of pneumonia in children aged under 5 years in Alexandria, Egypt: mothers' perspective. *East Mediterrian Health Journal*, 2020(26): 1-20.
- 77. O. C. A. 2014. Capacity development resource guide: Strategy development. *Health Policy Projec*. Washington, DC.
- Padmanabhan Ramachandran, Krishnamoorthi Nedunchelian, Appasamy Vengatesan & Saradha Suresh. 2012. Risk factors for mortality in community-acquired pneumonia among children aged 1-59 months admitted in an Indian referral hospital. *Indian Pediatrics* 49: 889-895.
- 79. Paul Kibet Keter. 2015. Knowledge, attitudes and practices of mothers in relation to childhood pneumonia and factors associated with pneumonia and seeking health care in Kapsabet District Hospital in Nandi County, Kenya, 1-85.
- Pavani K, Ram, Dhiman Dutt, Benjamin J, Silk, Saumil Doshi, Carole B, Rudra, Jaynal Abedin & AL, E. 2014. Household air quality risk factors associated with childhood pneumonia in urban Dhaka, Bangladesh. *The American Society of Tropical Medicine and Hygiene*, 90(5): 968-975.
- 81. Peter Johan Lor. 2017. Preparing for research: metatheoretical considerations. *Lor International & comparative librarianship*, 1(11): 1-58.
- Preeti Srivastava, Asit Kumar Mishra & Ashish Kumar Roy. 2015. Predisposing factors of community acquired pneumonia in under-five children. *Journal of Lung Diseases & Treatment*, 1(101): 1-4.
- S. Gomathi S, P. Latha Theresa & S. Jasmine Debora. 2017. WASH (Water, Sanitation and Hygiene) *International Journal* of Trend in Scientific Research and Development (IJTSRD), 2(1): 575-79.
- 84. Sidama Zone Health Department(SZHD). 2018. Annual report of Sidama zone health department for the year of 2018. *In:* SZHD (ed.). Hawassa, Ethiopia.
- 85. Save the children. 2017. Fighting for breath: A call to action on childhood pneumonia. 1 St John's Lane London EC1M 4AR, UK.
- 86. Save the children & UNICEF. 2019. Fighting for breath call to action: End childhood pneumonia deaths. London, ave New York: Save the children and UNICEF.

- 87. Save the children, UNICE, for every child and every breath counts. 2019. Fighting for Breath in Ethiopia: A call to action to stop children dying frompneumonia
- Sayyed Rashid Shah & AL-Bargi, A. 2013. Research paradigms: Researchers' Worldviews, Theoretical Frameworks and study designs. *Arab World English Journal* 4: 252 -264.
- Seda Tural Önur, Levent Dalar, Sinem İliaz & Arzu Didem Yalcin. 2016. Pneumonia in HIV-Infected Patients. *Eurasian Journal of Pulmonol*, 18: 11-7.
- 90. Shally Awasthi, Mark Nichter, Tuhina Verma, Neeraj Mohan Srivastava, Monica Agarwal, Jal Vir Singh & et al. 2015. Revisiting community case management of childhood pneumonia: Perceptions of caregivers and grass root health providers in Uttar Pradesh and Bihar, Northern India. *PLoS One* 10(4): 1-18.
- 91. Shally Awasthii, Chandra Mani Pandey, Tuhina Verma, Neha Mishra & Lucknow. 2019. Incidence of community acquired pneumonia in children aged 2-59 months of age in Uttar Pradesh and Bihar, India, in 2016: An indirect estimation. *PLoS ONE*, 14(3): 1-12.
- 92. Shireen Qassim Bham, Farhan Saeed & Manzar Alam Shah. 2016. Knowledge, attitude and practice of mothers on acute respiratory infection in children under five years *Pakistan Journal of Medical Science*, 32(6): 1557-1561.
- 93. Siham MO, Gritly, Mohamed Osman Elamin, Hatim Rahimtullah, Abdikani Y, Haji Ali, Abdi Dhiblaw, Ebtihal A, Mohamed & et al. 2018. Risk factors of pneumonia among children under 5 years at a pediatric hospital in Sudan. *International Journal of Medical Research & Health Sciences*, 7(4): 60-68.
- 94. S. R. C. Howie, J. Schellenberg, O. Chimah, R. C. Ideh, B. E. Ebruke, C. Oluwalana & et al. 2016. Childhood pneumonia and crowding, bed sharing and nutrition: a case-control study from the Gambia. *The International Journal of Tuberculosis and Lung Disease*, 20(10): 1405-1415.
- 95. Singh YK. 2006. Fundamentals of research methodology and tatistics, New Delhi · Bangalore, New Age International (P) limited.
- 96. Sitwala Imenda. 2014. Is there a conceptual difference between Theoretical and conceptual Frameworks? *J Social Science*,, 38: 185-195.
- 97. Smith KR, Mccracken JP, MW, W. & et al. 2011. Effect of reduction in household air pollution on childhood pneumonia in Guatemala (RESPIRE): a randomised controlled trial'. *Lancet* 2011, 378: 1717–1726.
- 98. Societies/FIRS. 2017. *The* Global impact of respiratory disease: Second edition, UK, European Respiratory Society.
- 99. Sommayya Aftab, Ftikhar Ejak, Umar Waqar, Humayun Iqbal Khan, Asif Hanif, Amir Usman & et al. 2016. Risk factors for childhood pneumonia in north eastern Paskistan: A case control study: *Malaysian journal of paediatrics and child Health Online Early* 4(22): 1-10.
- 100. Sullivan D. 2017. Bacterial pneumonia: symptoms, treatment, and prevention.
- 101. Susan Mary Pradhan, Arathip P. Rao, Sanjay M. Pattanshetty & A. R. Nilima 2016. Knowledge and perception regarding childhood pneumonia among mothers of under-five children in rural areas of Udupi Taluk, Karnataka: A cross-sectional study. *Indian Journal of Health Sciences*, 9(1): 30-39.
- 102. Teshome Abuka. 2017. Prevalence of pneumonia and factors associated among children 2-59 months old in Wondo Genet district, Sidama zone, SNNPR, Ethiopia. *Curr Pediatrics Research* 21(1), 19-25.

- 103. Theresa Diaz, Ashasha S George, Sowwya R Rao, Petter S Bangura, John B Baimba, Shannon a Mcmahon & et al. 2013. Healthcare seeking for diarrhoea, malaria and pneumonia among children in four poor rural districts in Sierra Leone in the context of free health care: results of a cross-sectional survey. *BMC Public Health* 13(157): 1-12
- 104. Turner, LM, H., S.B, D. & et al. 2004. Design and implimentation of osteo-porosis prevention using the Health Belief Model. *American Journal of Health Studies*, 19: 115-121.
- 105. Ujunwa FA & Thecla EC. 2014. Risk factors for acute respiratory tract infections in under-fve children in Enugu Southeast Nigeria. *Medical and Health Science Research*, 4, 95-99.
- 106. Ukwaja KN, Talabi AA & Aina OB. 2012. Pre-hospital care seeking behaviour for childhood acute respiratory infections in south-western Nigeria. *Int Health*, 4(4), 289-94.
- 107. UNICEF. 2007. Vitamin A supplementation. A decade of progress. *In:* (UNICEF), T. U. N. C. S. F. (ed.). New York: The United Nations Children's Fund (UNICEF).
- 108. UNICEF. 2014. Ethiopia Central Data Base. October 2014.
- 109. UNICEF. 2015. Water, sanitation and hygiene The case for support *In:* children, U. F. (ed.).
- 110. UNICEF.2016a. Health Pneumonia". N.p 2016. The United Nations Children's Fund (UNICEF): *Web. Available*
- 111. UNICEF. 2016b. The State of the World's Children 2016: A fair chance for every child.
- 112. UNICEF 2016c. UNICEF pneumonia-diarrhea report; One is too many Ending child deaths from pneumonia and diarrhoe. *In:* (UNICEF), U. N. C. S. F. (ed.). New York, USA: United Nations Children's Fund (UNICEF.
- 113. United Nations Development Programme (UNDP). 2009. Handbook on planning, monitoring and evaluation for development result, New York.
- 114. USAID & ABT. 2015. Impact of health systems strengthening on health. *In:* Systems, U. O. O. H. (ed.). Washington, DC The health finance and governance project
- 115. Van Thuan Hoang, Thil Loi Dao, Philippe Minodier, Duy Cuong Nhuyen, Nang Trong Hoang, Van Nghiem Dang & et al. 2019. Risk factors for severe pneumonia according to WHO 2005 criteria definition among children <5 years of age in Thai Binh, Vietnam: A case–control study. *tional Journal of computational intelligence* 9(4), 274-80.
- 116. WHO. 1998. Health promotion Glossary. *In:* World Health Organization, W. (ed.). Geneva, Swizerland.
- 117. WHO. 2000. Nutrition for health and development (NHD): A global agenda for combating malnutrition: World Health Organization 2000: Sustainable Development and Healthy Environments (SDE)
- 118. WHO. 2005. Malnutrition quantifying the health impact at national and local levels: World health organization nutrition for health and development protection of the human environment Geneva 2005 Geneva, Switzerland., WHO Document Production Services.
- 119. WHO. 2007. Everybody's business, strengthening health systems to improve health outcomes WHO's framework for action. *In:* (HSS), H. S. A. S. (ed.). Geneva, Switzerland: WHO.
- 120. WHO. 2009a. Infant and young child feeding model chapter for textbooks for medical students and allied health professionals. *In:* Organization, W. H. (ed.). Geneva.
- 121. WHO. 2009b. Monitoring and evaluation of health systems strengthening, An operational framework. Geneva: WHO.
- 122. WHO. 2012. Health education: theoretical concepts, effective strategies and core competencies: A foundation document to

guide capacity development of health educators, regional office for the eastern mediterranean, Cairo.

- 123. WHO. 2013. Introduction of pneumococcal vaccine PCV10, two dose presentation A handbook for district and health facility staff, Geneva, Switzerland, World Health Organization.
- 124. WHO. 2014. Revised world health organization classification and treatment of childhood pneumonia at health facilities. *In:* WHO (ed.): *Evidence summaries*. Switzerland: World Health Organization.
- 125. WHO. 2015. Immunization in practice. A practical guide for health staff 2015 update, Geneva, Switherland, WHO.
- 126. WHO. 2016a. Oxygen therapy for children. Geneva, Switzerland: World Health Organization.
- 127. WHO. 2016b. Strategic action plan to reduce the double burden of malnutrition in the South-East Asia region 2016–2025. India.
- 128. WHO. 2018a. Household air pollution and health: Fact sheets. *In:* World health organization, W. (ed.)
- 129. WHO. 2018b. Air pollution and child health: Prescribing clean air, Summary. *In:* World health organization, W. (ed.). Geneva, Switherland: Department of public health, environmental and social determinants of health climate and other determinants of health cluster
- 130. WHO. 2018c. Housing and health guidelines, World health organizatiion, Switzerland
- 131. WHO. 2018d. Water, sanitation and hygiene strategy 2018-2025. World health organization, Geneva, Switherland: Department of public health, environmental and social determinants of health.
- 132. WHO.2019a.Factsheet on Pneumonia,. World health organization, Geneva, Switherland
- 133. WHO. 2019b. Fact sheet: Reducing child mortality. World health organization, Geneva, Switherland
- 134. WHO & UNAIDS. 2013. Paediatric HIV surveillance among infants and children less than 18 years of age Geneva 27, Switzerland.

- 135. WHO & UNICEF. 2006. Pneumonia: The forgotten killer of children. The United Nations Children's Fund (UNICEF) September 2006. *In:* WHO/UNICEF (ed.).
- 136. WHO & UNICEF. 2009a. Global action plan for prevention and control of pneumonia (GAPP) *In:* World health organization/The united nations children's fund (UNICEF (ed.), Geneva, Switzerland:.
- 137. WHO & UNICEF. 2009b. Co-trimoxazole prophylaxis for HIV-exposed and HIV-infected infants and children: Practical approaches to implementation and scale up. *In:* UNICEF, W. H. O. A. (ed.). Geneva, 27 Swizerland.
- 138. WHO & UNICEF. 2013. Ending preventable child deaths from pneumonia and diarrhoea by 2025: The integrated global action plan for pneumonia and diarrhoea (GAPPD). *In:* (ed.). World health organization/The united nations children's fund (UNICEF): Geneva, Switzerland
- 139. Worku Awoke. 2013. Prevalence of childhood illness and mothers'/caregivers' care seeking behavior in Bahir Dar, Ethiopia: A descriptive community based cross sectional study, *Open Journal of Preventive Medicine* 3(2), 155-159.
- 140. Yahaya AK, Ekenyong BN, & Obegu P. 2018. Caregivers' knowledge of pneumonia and uptake of vaccination in underfive children in Kaduna State, Nigeria. *Journal of Medical and Biomedical Sciences*, 7(1), 40-48.
- 141. Yordanos M, Abel FD, Abayneh GD, Yohannes AH, Behailu TD, Getu D & et al. 2019. Determinants of under-five pneumonia at Gondar University Hospital, Northwest Ethiopia: Unmatched case-control study. *Hindawi Journal of Environmental and Public Health*, 2019.
- 142. Yoseph Merkeb & Fentahun Adane. 2019. The magnitude and associated factors of pneumonia among under-five children in Ethiopia: A systematic review and meta-analysis (2014-2019). *BMC Pediatrics*, 10, 1-24.
- 143. Yulianda Yeimo, Isna Qadrijati & Bhisma Murti. 2018. Environmental factors associated with pneumonia in children under-five in Nduga district, Papua. *Journal of Epidemiology and Public Health*, 3(3): 307-311.

ANNEXURES

Annexure A: Ethical clearance from UNISA

RESEARCH ETHICS COMMITTEE: DEPARTMENT OF HEALTH STUDIES REC-012714-039 (NHERC)

18 December 2019 Dear Derese Desalegn Buta

HSHDC/945/2019

Student: Derese Desalegn Buta

Student No: 62136313

Supervisor: Prof L Zungu

Qualification: PhD

Decision: Approval

Name: Derese Desalegn Buta

Proposal: Development of awareness and prevention strategies for Pneumonia among Children under five years in Sidama Zone, Southern Ethiopia

Qualification: PhD

Risk Level: Low risk

Thank you for the application for research ethics approval from the Research Ethics Committee: Department of Health Studies, for the above mentioned research. Final approval is granted from 18 December 2019 to 18 December 2024.

Thank you for the application for research ethics approval from the Research Ethics Committee: Department of Health Studies, for the above mentioned research. Final approval is granted from 18 December 2019 to 18 December 2024.

The application was reviewed in compliance with the Unisa Policy on Research Ethics by the Research Ethics Committee: Department of Health Studies on 12/12/2019.

The proposed research may now commence with the proviso that:

- The researcher/s will ensure that the research project adheres to the values and principles expressed in the UNISA Policy on Research Ethics.
- 2) Any adverse circumstance arising in the undertaking of the research project that is relevant to the ethicality of the study, as well as changes in the methodology, should be communicated in writing to the Research Ethics Review Committee, Department

University of South Africs Prever Street, Mocklemat, Ridge, City of Tshutare PPO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 3111 Facshrille; +27 12 429 4130 www.unisa.ac.za







2019 -12- 2 0

Executive Dean College of Human Sciences of Health Studies. An amended application could be requested if there are substantial changes from the existing proposal, especially if those changes affect any of the study-related risks for the research participants.

- 3) The researcher will ensure that the research project adheres to any applicable national legislation, professional codes of conduct, institutional guidelines and scientific standards relevant to the specific field of study.
- 4) You are required to submit an annual report by 30 January of each year that indicates that the study is active. Reports should be submitted to the administrator <u>HSREC@unisa.ac.za.</u> Should the reports not be forthcoming the ethical permission might be revoked until such time as the reports are presented.

Note:

The reference numbers [top middle and right corner of this communiqué] should be clearly indicated on all forms of communication [e.g. Webmail, E-mail messages, letters] with the intended research participants, as well as with the Research Ethics Committee: Department of Health Studies.

Kind regards,

- lee

Prof JM Mathibe-Neke CHAIRPERSON <u>mathijm@unisa.ac.za</u>

der

Prof A Phillips DEAN OF COLLEGE OF HUMAN SCIENCES

Annexure B Letter from SNNPR Health bureau

መንግሥት ጤና በሮ	Ref No
South Nations Nationalities and People's Regional	+3 28-5-12
State Health Bureau	Date

AA. 99 13 mg 00906,9

UPI

7.88: ATSA POLLO AAACH Bookhta

ስላይ አንደተገለጸው የፒ.ኤች.ዲ (PhD) ተማሪ የሆኑት አቶ ደረስ ደሳለኝ "Development of awareness and prevention strategies for pneumonia among children under five years in Sidama zone, Southern Ethiopia" በሚል ርዕስ ዋናት ለማካሄድ አቅዷል። የዋናት ይዘት (Proposal) በድኒሳ ድኒቨርስቲ የዋናት ስን ምግባርና ግምንማ ኮሚቴ ታይቶ የወደቀና ሙሉ ፈቃድ ይገኘ ሲሆን የዋናቱ ሙጤት በክልላችን ስጤና አንልማለተት ስኬት እየተደረገ ሳሰው ተማባር አንድ ግብአት ሲያገለማል የሚችል በመሆኑ በእናንተ በኩል አስራላጊውን ትብብር ሁሉ እንድታደርጉላቸው እንጠይቃለን።

13.89 XST ADA. 2 Phist mapit PAGT90!!

አዕርጣ ወንድሙ ወልደማር የም

1. 000/1/m/2/m/2/0/0

70100

✓ Ams PCPCS theA. A/R/8/19

✓ hf LLA LAA3 - AAA+

Annexure C: Letter of permission from former Sidama zone health department to conduct research

	T.Kiirom [15]
Sidaamu Zoone Gashshooti Giddo Fayyimmate Biddishsha	Barra 1/7/Lon
በሲዳማ ዞን መስተዳድር ጤና መምሪያ	ቀን

ለበንሳ ወረዳ ጤና ጥ/ጽ/ቤት <u>ዳዬ</u> ለቦና ዙ/ወረዳ ጤና ጥ/ጽ/ቤት <u>ቦና</u> ለአለታ ወንዶ ከተማ ጤና ጥ/ጽ/ቤት <u>አለታ ወንዶ</u> ለሎካ አባያ ወረዳ ጤና ጥ/ጽ/ቤት <u>ሐንጣጤ</u> ለዳሌ ወረዳ ጤና ጥ/ጽ/ቤት <u>ይርጋዓለም</u> ለወንሾ ወረዳ ጤና ጥ/ጽ/ቤት <u>ቦኳሶ</u>

<u> ጉዳዩ፡- ለጥናት የሚደረግ ትብብርን ይመለከታል</u>

በመባቢያዉ እንደተገለጸዉ በUNISA የፒ.ኤች.ዲ (PhD) የሆኑት ተማሪ አቶ ደረስ ደሳለኝ "Development of Awareness and Prevention strategies for Pneumonia among children under five years age in Sidama Zone, Southern Ethiopia" በሚል ርዕስ ጥናት ለማካሄድ እንዳቀዱና ይህም በዩኒቨርስቲዉ ስነ-ምግባርና ግንግማ ኮሚቴ ታይቶ እንኤፀደቀና ሙሉ ፊቃድ እንዳገኝ በመጥቀስና በዞናችንም አስፈላጊዉን ትብብር እንድናደርግላቸዉ ከደቡብ ብ/ብ/ሕ/ከ/መንግስት ጤና ቢሮ በቁጥር የዉ6-19-44 በቀን 28/5/2012 ዓ.ም በተጻፌ ደብዳቤ ተገልጾልናል።

በመሆኑም የጥናቱ ዉጤት በዞናችንም ለሚደረንዉ ጤና አንልግሎት ስኬት እንዴ ግብዓት ሊያንለግል ስለሚችል በእናንተ በኩል ተንብዉን ድጋፊና ትብብር እንድታደረጉሳቸዉ እንጠይቃለን፡፡

ግልባጭ ❖ ለአቶ ደረሰ ደሳለኝ <u>ባሉበት</u>

Allulu Yoheanno One State of the State of t

Annexure D: Information sheet and consent form Information sheet

This sheet is to be read for the respondents of the study. My name is ______. I am working as data collector for Mr Derese Desalegn /PhD student at UNISA/. We are conducting this survey inbehalf of Mr Derese Desalegn collaboration with UNISA and SNNPR Health Bureau.

Purpose/aim: To develop awareness and prevention strategies on pneumonia in Sidama Regional State, Ethiopia through assessing the prevalence of childhood pneumonia and its associated factors and exploring knowledge, attitude and practice of mothers towards childhood pneumonia prevention and control.

Benefit of current study

These will be an important contributions for policy makers and organizations those involved on care and support for under-five children' activities.

Right to participate/refuse and any harm due to participation

Your involvement in this study will be voluntary base. If you decide not to participate, there will be no negative consequences for you. If you do agree to participate there will no incentives from this study, however your participation in this study is very important for achievement of the study goal and for paving the way for the integration of under-five pneumonia health service in the early detection and giving appropriate care for children thereby increasing the quality of care for these people.

There is no any risk that will occur to you because of your participation in this study.

Confidentiality Issue

All the responses given by you and results obtained will be kept confidential by using coding system whereby no one will have access to your response. You are not expected to give your name or phone number. Without permission from you and legal body, any part of this study will not be disclosed to third person. You have full right to refuse and withdrawal to participate in this study if you don't wish any time.

Time needed

The interview period might take about 30 minutes. If you are willing to participate in this study, you need to understand and sign the agreement form, and then you will be asked to give your responses by data collectors.

If you want to know further information you can communicate the researcher through:

Name of Investigator: Derese Desalegn Tel: +251913436793

Name of Advisors: Lindiwe Zungu (Prof) Email: zunguli@unisa.ac.za

Are you voluntary to participate in the interview? Yes____No___ Consent to participate in this study

Consent to participate in this stu

Respondents consent

I, ________ (participant name), confirm that the person asking my consent to take part in this study has told me about the nature, procedure, potential benefits and anticipated inconvenience of participation. I have read (or had explained to me) and understood the study as explained in the information sheet. I have had sufficient opportunity to ask questions and am prepared to participate in the study. I understand that my participation is voluntary and base that I am free to withdraw at any time without penalty (if applicable).

I am aware that the findings of this study will be processed into a research report, journal publications and/or conference proceedings, but that my participation will be kept confidential unless otherwise specified.

I agree to the respond/recording of the "Awareness and prevention strategies on pneumonia among under five children in Sidama Regional State, Ethiopia".

I have received a signed copy of the informed consent agreement.

Research project and I consent to participate voluntarily in the research project. I understand that I am at autonomy to withdraw from the project at any time.

Participant Name & Suriname	Signature	Date	
Data collectors Name	Signature	Date	
Researcher's Name & Surname	Signature	Date	

Annexure E: Data collection tool for quantitative part of study (English version)

001. Respondents ID Number ____

002. Address: Woreda kebele Village

003. Code of the child (Take the youngest child for the assessment even they are twins)

S/No	Questions	Choice of responses	Skip to
	1. Socio-demographic and economic information		
101	Relation of the respondent to the child?	 Mother Father Care givers 88.Others/specify 	
102	Usual residence of the respondents	 Urban Rural 	

103	Marital status of the respondents	 Single married Widowed Divorced Others/specify 	
104	Total number of family members?	in number	
105	Do you have your own house?	1. Yes 2. No	
106	What is your religion?	1. Protestant 2. Orthodox 3. Muslim 88. Others(specify)	
107	What is your ethnic group?	1. Sidama 2. Amhara 3. Wolaita 4. Oromo 88. Others(specify)	
108	What is your educational level?	1. Illiterate2. Elementary school(1-8)3. Secondary school 9-12)4. Above secondary school	
109	How old are you?	years	
110	What is your job?	 House wife Governmental employer Merchant Daily labourer Others/specify 	
111	What is the educational level child's father?	1. Illiterate 2. Elementary school(1-8) 3. Secondary school 9-12) 4. Above secondary school	
112	What is the occupation of child's father?	1. Farmer 2. Governmental employer 3. Merchant 4. Daily labourer 88.Others/specify	
113	What is the main source of income of household?	1. Privately owned farm 2. Daily labourer 3. Salary government or NGOs 4. Trading 88.Others/specify	
114	Wealth index for rural dwellers:		
114.1	Do household owned agricultural land?	1. Yes 2. No	114.3
114.2	How many hectares of agricultural land do household owned?	hectare	
114.3	Do you have your own house?	1. Yes 2. No	
114.4	Is your house roufed with corrugated iron?	1. yes 2. No	
114.5	Do you have a functional radio?	1. Yes 2No	

114.6	Do you have a bicycle?	1. Yes 2. No	
114.7	Do you have a blanket?	1. Yes 2. No	
114.8	Do you have cotton mattress?	1. Yes 2. No	
114.9	Do you have a motor cycle?	1. Yes 2. No	
114.10	Do you have a sofa?	1. Yes 2. No	
114.11	Do the household use electric power?	1. Yes 2. No	
114.12	Do you have a TV?	1. Yes 2. No	
114.13	Do you have a mobile Phone?	1. Yes 2. No	
S/No	Questions	Choice of responses	Skip to
	2. Childs' Characteristics		
201	What is the age of your child	months	
202	Sex of the child	 Male Female 	
203	Where was your child born?	1. Health institution 2. At Home 88. Others/specify	
204	Number of total children mother ever do have until the day of data collection	1. One 2. Two 3. Three 4. Four 88 Others/specify	
205	Birth order of the sample child?	1. First 2. Second 3. Third 4. Fourth & above	
206	The birth interval of children(the sample child with its older one)	months/year	
207	Is child currently diseased	1. Yes 2. No	
208	If yes what is? The respondents perception	 Common cold Pneumonia Malaria Measles 88. Others/specify 	
209	Has your child had cough with or without fever for less than two weeks duration?	1. Yes 2. No	
210	Assess the child for pneumonia (assess respiratory rate per minute and classify the finding, PR/minute) > 50 (infants 2-11 months) > 40 (children 12-59 months) Circle if any of the following Chest In drawing or Stridor or any danger signs (Not able to eat, drink or breast feed or Lethargic or unconscious or Vomits everything or r Convulsions)	 No pneumonia Pneumonia Severe pneumonia or Very sever disease 	NB: Follow the Assessment and classification sheet

211	Did child ever have known other illness historical?	1. Yes 2. No 3. I don't know	
212	If yes for the above Q211 what is it? May be more than one answers	1. Measles 2. Malaria 3. Tonsillitis 4. Diarrheal 88 Others/specify	
213	Has ever your child had measles last year	1. Yes 2. No 3. Un kwon status	
	3. Information on environmental factors		
301	Do you have latrine? check	1. Yes 2. No	
302	If yes for Q301 which type they do have?	1. Simple pit latrine 2. VIP 88. Others/specify	
303	Where do you dispose solid waste?	 Backyard simple Blow a fuse at any where Bury 88. Others/specify 	
304	Is there separate house for human beings and domestic animals? Check	1. Yes 2. No	
305	Type of roof materials of the living house? check	1. Grass 2. Corrugated iron 3. Bamboo 88. Others/specify	
306	Type of floor material of the living house? check	1. Mud 2. Cement 3. Wood 88. Others/specify	
307	Does the living house have a window? Check?	1. Yes 2. No	
308	Number of rooms in the living house. Check	1. One 2. Two 3. Three 4. Four 5. More than 4	
309	Is there any animal living with the same house?	1. Yes 2. No	311
310	If yes for Q309 which animal?	1. Cow/bull 2. Sheep 3. Goat 4. Horse 5. Donkey 88 Others/specify	
311	What kind of fuel is used mostly for cooking in your home?	1. Charcoal 2. Wood 3. Electricity 4. Kerosene 5. Animal dung 88 Other specify	
312	Do you use fire charcoal or ' <i>Kesel</i> ' for other purpose / for coffee, to keep the temperature warm inside your living home?	1. Yes 2. No	
313	If yes, how often do you fire charcoal?	 Once per a day Twice per a day Three times and above per a day Once per every two days 	

			88 Other/specify	
214	If you to Q200 shows where do not			
514	II no to Q509 above, where do you		2 Kitchen	
			3 Outdoors	
			88. Others/specify	
315	Is there a hood or chimney in the h	ouse or kitchen during data	1. Yes	
	collection? Check	-	2. No	
216	Number of windows in the kitchen		1. One	
510	Number of windows in the kitchen		1. One 2. Two	
			3. Three	
			4. None	
			88 Others/specify	
317	Where is the usual location of the	child during cooking?	1. at the mother's back or beside the mother	
		6 6	2. Outside of the cooking house	
			88. Others/specify	
318	The energy source of lighting in ye	our home	1. Masho	
			2. Kuraz	
			3. Wood	
			4. Fanos 5. Electricity	
			5. Electricity 88 Others/specify	
319	Is there any smoker in the house?		1. Yes	201
320	If yes for O319, who is an active s	moker inside the house?	2. NO 1 Eather	521
520	If yes for Q319, who is an active s	moker made the nouse?	2 Mother	
			88. Others/specify	
321	Do any family members have a his	tory of HIV/AIDS? Explain it	1. Yes	
	clearly		2. No	323
			3. Unknown status	
322	If yes for Q321 who has/have got i	t?	1. Father	
			2. Mother 3. Brother/cister	
			88. Others/Specify	
			•••• • • • • • • • • • • • • • • • • •	
323	Do any family members have a his	tory of respiratory diseases	1. Yes	101
	(bronchitis, TB of lung, Asthma et	c.)? Explain it clearly	2. No	401
324	If yes for Q323 who has/have got	+9	3. Unknown	
524	If yes for Q525 who has/have got h		2 Mother	
			88. Others/specify	
	4. Nutritional related factor			
401	What was the weight of the child a	t the time of delivery?	Kg	
	Measurements	Reading 1	Reading 2	Average
		Ũ	C	e
402	Weight(Kg)			
403	Height(cm)			
404	Mid upper Arm			
	circumference(cm)			
405	Check for bilateral pitting edema			
10.5				
406	Check for palmar pallor	Is there severe palmar	Isthere some palmar pallor?	
		panor :		
1		1	1	

407	When did you start initiation of breast feeding?	 Immediately after birth Within one hour After 1 hour and within 24 hours After a day 	
408	Breast feeding status of the child during the first 6 months of life.	 Exclusive breastfeeding Partial Breast feeding Not Breast fed at all 88.Others/specify 	
409	For how long have you breastfed your child?	 Less than 6 months up to 12 months for two years More than two years 	
410	If your child is still breast feeding how many time per day (per 24 hour) do you breast feed your child.	 Less than 5 times Up to 10 times More than 10 I don't know 	
411	Have you started feeding your child other than breast when it was greater than 6 months?	1. Yes 2. No	413
412	What was your child's age when you were feeding him/her solids for the first time?	 Less than 6 months At 6 months More than 6 months 	
413	What do you use to feed the milk to the child?	 Bottle Cup Spoon By hand 88 Others/Specify 	
414	Within the last six months did the child have received de-worming (for children aged 24-59 months) brief it clearly!	1. Yes 2. No	
415	Within the last six months did the child receive vitamin A (for children aged 6-59 months) brief it clearly!	1. Yes 2. No	
416	Within the last six months did the child received Zinc (for children aged 6-59 months) brief it clearly!	1. Yes 2. No	
	5. Information on Immunization status of the child		
501	Did your child ever had vaccinated for any vaccinations?	1. Yes 2. No	503
502	If yes circle on the following options if the card shows a dose is given—check EPI card if available	 Penta" 1 Penta" 2 Penta" 3 PCV"1 PCV"2 PCV"3 Rota1 Rota 2 Measles 	
503	Did your child had a PCV_{10} vaccination, that is, an injection given in the right thigh or buttocks sometimes at the same time as polio drop?	1. Yes 2. No	
504	How many times was a PCV ₁₀ vaccination given?		
505	A measles injection that is, a shot in the arm at the age of 9 months or older - to prevent him/her from getting measles?	1. Yes 2. No	
506	Why did child not receive the vaccine? If it was not given for its age.	 Time shortage for mother Lack of knowledge Unavailable of service Fear of side effect 88. Other/specify 	
6. KAP of t	he respondents		
6.1 Informa	ition on Knowledge		
601	Do you know what pneumonia is?	1. Yes 2. No	

602	If yes Q601 (in your opinion) what is pneumonia? (Please write)		
603	What is the name given to pneumonia in your local language?		
604	From which sources did you first get information on pneumonia?	 HEWs Health workers TV or radio Older persons in family/community Newspapers, publications and journals 88 Other (Specify) 	
605	In your opinion, what are the signs and symptoms of pneumonia? (Can be many answers)	 Fever Chills Headache Cough Chest in drawing Wheezing Shortness/difficulty in breathing Chest pain/sharp/stabbing pain Convulsions Rapid/fast breathing Fatigue/weak Loss of appetite Excessive sweating and clammy skin Body/muscle pain Do not know Other/specify 	
606	What do you think are the causes of pneumonia?	 Cold temperature/weather change Dust/germs Bacteria Virus Fungi Witchcraft Do not know Others/specify 	
607	Have you seen or heard any information on pneumonia vaccine?	1. Yes 2. No	
608	What measures do you think should be taken to prevent childhood pneumonia? (can be many answers)	 Vaccination/immunization Proper nutrition (balanced diet) Breastfeed exclusively Clothing warmly during cold season Proper house ventilation Washing hands before feeding Do not know Other/specify 	
609	What do you think, how do people get infection of respiratory diseases?	 Air Contact way Dirty hands 88. Others/Specify 	
6.2 Attitude	e related questions		
610	What do you think is the risk of childhood pneumonia?	 Not dangerous Dangerous Highly dangerous Do not know 	
611	A child with pneumonia should be brought immediately to health institution	 Strongly disagree Agree Indifferent Disagree strongly disagree 	
612	Traditional healers sometimes can treat childhood pneumonia	 Strongly disagree Agree Indifferent Disagree 	

r		
		5. strongly disagree
613	Mothers can treat cough or common cold at home	1. Strongly disagree
		2. Agree
		3. Indifferent
		4. Disagree
		strongly disagree
614	Reducing indoor smoke and dust may help to decrease pneumonia	1. Strongly disagree
	occurrence	2. Agree
		3. Indifferent
		4. Disagree
		5. strongly disagree
615	Exclusive breast feeding and EPI may support to decrease	1. Strongly disagree
	childhood pneumonia morbidity	2. Agree
	······································	3. Indifferent
		4. Disagree
		5 strongly disagree
616	We can by antibiotics for child with pneumonia from shop/any	1 Strongly disagree
010	pharmacy without visiting health facility	2 A gree
	pharmacy without visiting health facility	2. Agree 3. Indifferent
		J. Disagroo
		4. Disaglee
(2 Drug at		5. strongly disagree
0.5 Fracu	the second parents towards pheumoinal prevention and controls	1 V
617	Has your child ever suffered from pneumonia?	1. Yes
		2. No
		3. Do not know
618	If you had a child suspected of pneumonia, where would you first	1. Take child to the closest health
	seek treatment/advice?	facility/Hospital or Health Center
		2. Traditional practitioner
		3. Visit a pharmacist
		4. Religious leader Get
		5. medicine from a shop
		88.Other/specify
619	When would you visit the health Facility when a child is suspected	1. Immediately/within 24hrs
	of pneumonia?	2 After 24 hours
		2. <i>Intel</i> 24 notifs
620	At home, prior to hospital/health center attendance, what would	1. Giving drugs e.g pain killers
	you do if you had a sick child with fever and/or suspected of	2. Feed child with liquid food/fluids
	pneumonia?	3. Breastfeeding
		4. Nothing
		88. Other/specify
621	Have you ever vaccinated your children including siblings?	1. Yes
		2. No
622	Do you have any ventilation system to remove smoke and steam	1. Yes
	which may help to reduce pneumonia infection?	2. No
623	Do you care about reducing indoor smoke and dust to reduce	
025	neumonia infection?	$2 N_0$
624	Do you give Herbs like Tenedem Demokese and Dinacteons to a	1 Ves
024	child while getting pneumonia at home?	1. 105 2 No
625	Do you alothe your abild as parits status to present and 1 (2)	2. INU 1. Vec
025	Do you clothe your child as per its status to prevent cold or hot?	1. 1 es 2. N-
1		2. INO

Thank You very much for giving me your precious time!

Annexure F: Information sheet, consent form and data collection tool for quantitative part of study (Sidaamu Afoo) version

Sumuu yaatenna mashalaqqe aa la'anno forme / Information sheet/

Keere hosinooni/galtinooni? Su'maya _______yinanie, tenne yannara Wodidi Afrikira afantanno Yuniverisitera/UNISA/ho 3ki Digire/PhD/ rosanohu Kalaa Derrese Dessaalenyihura xinxallote taje gamba assitano gaamo millaati. Gamba asseemo tajeno Sidaamu Dagoomu Mootima Qoqqowi giddo ontu dirri woro noo qaaqullira foolate qarri xisso abbitannori hiikkuriitiro/maatiro afatenna tenne xisso gargadhate widooni amuwahono ikko losaanonsara noonsa egenonna lao badate dandiisitanota ikkitana jeefoteno haaro straateje qineesate assinani xiinxalora ikkitano taje gamba assateeti.

Furcho: Tini mashalaqqenna umi kaisha xa'mate forme hasiisinohu, xiinxalote umona ho'ro, kulateeti, konneno assitanori, tenne xiinxallo gaamo miilaati.

Tenne xiinxalloti umi hasatto

Sidaamu zoone giddo, ontu dirri woroo noo qaaqullira foolate qarri/Saamba michete/ xibba abitanori maatiro afatenna tenne xisso daafo amuwahono ikko qaaquwu awuutora noonsa egenonna lao hattono gargadhate assitano sharronsa bade afatenni aantete tenne xisso gargadha dandiisitanota haaro hayyo kalaqateeti. Xiinxallote miillano mittu tenne xiinxallo assanohu, tonaa lamu mashalaqe ganba assitano neerisse, sasu kaa'lote laoshe assitano faayyimate ogeyena mitte supervayizere afantanno.

Tenne xiinxalloti ho'rose

Xiinxallote ho'ro mangistete giddona gobbani loossano Fayyimate poolise fushitanori, piroogiraame kalaqanorina, daga/dagoomu balluunku afe qarrasi tidhitano gede raga worate kaa'litano. Kunino fayyimate ekistenshiine pirogiraame woyeesatenna woleno foolate qarri xibbuwa gargadhatena qorophate kaimu hedo shiqishate kaa'litano yine hendanni.

Beeqaano doorate ikkito

Ninke ate tenne xiinxallora dooromohehu taje gamba assinani qarqarira teesaancho ikkakiniitila, atino beeqate/tumo qolate fajjitankero giddokini halaalu aana xaadano mashalaqe calla aattae gede huucireemmo. Tenne xiinxallo assinanihu amate/lassaanchotena qaaqulu annaati.

Tenne xiinxallora beeqatenni hoogatari/gawajjamattari

Xiinxallote giddo beeqatenni hoogata/ori mitturino nookkitana hoogitarono sajjoo daqiiqakki callaati. Mashalaqekkinno konne xibba hunate assinani sharrora lowo geeshsha kaa'litanota ikkitana beeqakininno gawajaaheri mitturino nookita kulate baxeemo.

Tenne xiinxalloni beeqateni afi'nanni ho'ro

Tenne xiinxallora beeqateni afirato/ta ho'ro nafu hooguro, beeqakini qaaquuleho samba michete xibba abbitanore bande anfe hunate assinani xiinxallora kaa'lo assatatana, tini kaa'lo addi doogoni ateno ikko dagooma kaa'litanota ikkase kulate baxeemo.

Beeqqateni afi'nani batooshe

Tenne xiinxallora beeqakira uyinanihu baxxino batooshi dino, kayi beeqaki dagooma kaa'litanota ikkaseni lowo geeshi galati heeranonke.

Mashalaqqete maxooshe

Tenne xiinxaloni afi'nanni mashalaqe maxooshuni amandanitana, su'mikino mashalaqeki ledo ofolanokita, ayiiranno ootta/o qolo daafira kuleemokitana kompiyuutereteno maxooshu kiironi maxeemota kulammohera baxeemo. Mashalaqe ganba asse gudumohu gedensaanni xa'mote woraqata baalanata Xiinxalaanchoho eemositana isino ate ayiima afikini calla maaxooshi kiiro garini loosanota xawiseemohe.

Dihasireemmo/a yaate hasatto

Ayee yannarano hasirito/taro xiinxalote giddoni fula dandaatatana, wo'mantano ikkito gama xa'mo diqoleema yiitaro dandatata kuleemohe.

Konne assakini qaaqiki foolate qarri xibbini xiwamiro haasiratta/o horro gattanohekita kulamohera baxeemo.

Hasiisiro xaandanni manni su'ma

Tini xiinxallo UNISA yuniverisite xiinxalote amanyoote agartano gaamowa la"ante, sumuu yinnonita ikkitana xiinxalote daafira lexxote mashalaqqe afate hasirittoro/taro konni woroo noo manni wiini afira/to/tata kulate baxeemo.

Derrese Desaaleny: Xinxalaanchoho UNISA Yuniveeristera, Bilbilla: +251 913436793

Profeeseri Lindiwe Zungu zunguli@unisa.ac.za

Mashalaqqe aate sumuu yaata/o? Ee yeema/o_____Dee'ni/dihasireemma/o___

Tenne xiinxallora beeqate/mashalaqqe aate/ sumuu yaate taje/ Consent to participate in this study

Ani su'ma'ya ______ yaamameema/o, tenne xiinxallora hasiisano mashaqe fajjo'yanni eema/o gede hasiisanno xawishsha taje gamba assanohuwiini ikkadu garinni afiroomaha ikkanna, tenne xiinxoliti kaimise mayiraatirona horoseno maatiro hattono mashalaqqe aa hooga dandeemata beeqa'yannino anera baxxitino gawajjo iilitanoeti nooki gede seeke huwatooma/o.

Tenne xiinxaloni anfanni gumi xaphi yii gedensaani dagate wodhosi agare iilanoha ikkasino, tenneni ka'neno qaaquuleho saamba michete xisso gargarate dandiisitano hayyo kalanqanita qummi assinoonie.

Tini xiinxallo uuyitano horo hedote giddo woratenni anino umi'ya fajjoni hasiisano mashalaqqe aate sumuu yoomata/o xawisate baxeemma/o

Beeqaanchu/ote Su'ma	Malaate	Barra
Taje gamba assanohu su'ma	Malaate	Barra
Harunsaanchu su'ma	Malaate	Barra
Xiinxalaanchu Su'ma	_Malaate	Barra

Qaaqquuleho Saamba michete xisso taje gamba assate qixxaabino xa'mo (data collection tool)

001 Xa'mamaanchu kiiro____

002 Teessosi/se: Worada_____Ollaa____Qacha____

003 Xiinxallote hanqafami qaaqi badooshu kiiro_____(Ilamate shiimidiha adhi)

T.kiiro	Xa'mo	Dawaro	Kubbi			
	1. Dagoomu, ikkonoometenna(honqooqimatenna) hajo					
101	Ati Qaaqqoho/te maati?	1. Ama2. Anna3. Awuuto88. Wole xawisi				
102	Teesso'ne mamaati?	1. Katama 2. Gaxare				
103	Adhanana adhamate gara	 Adhaminokkire Adhame ledo heeranore Galte shiidheeti/shiireehu Adhame tidhantinote/minoho 88. Wole xawisi 				
104	Mini'ne giddo xaphoomunni meu manni hedhinanni?	kiirotenni				
105	Umi'ne mini noo'ne	1. Ee 2. Dinonke				
106	Amma'noki maati?	1. Protesitaante 2. Ortodokise 3. Isilaama 88. wole xawisi				
107	Ilamaki maati?	1. Sidaamaho 2. Amaharaho 3. Wolayitaho 4. Oromote 88. wole xawisi				
108	Rosiki deerri mageeshaati?	 Horonta dirossino Umi dirimi roso (1-8 kifile) Layinki dirimi roso (9-12 kifile) Layinki dirimi rosi aleenni 				
109	Diriki me"eho?	dirunni				
110	Loosiki qeechi maati?	 Mini amaati Mangisitete loosaasinchooti Daddalaanchote Barru loosaasinchooti 88. wole xawisi 				
111	Qaaqqu/ote anni rosu deeri mageeshaati?	 Horonta dirossino Umi dirimi roso(1-8 kifile) Layinki dirimi roso (9-12 kifile) Layinki dirimi rosi aleenni 				
112	Qaaqqu/ote anni loosi maati?	 Loosire galinoho Mangistete loosaasinchooti Daddalaanchoho Barru loosaasinchooti 88.Wole xawisi 				
113 114 Jirote	Maate'neti eote buicho maati?	 Baatto loosi'rateni Barru looso loosatenni Mangisitete loosini Daddalunni 88. Wole xawisi 				

114.1	Baatto noo'ne loosidhinanniti?	1. Ee 2. Dinonke	
114.2	Mageeshi baatto noo'ne	 <= 1 hekitaare 2- 3 hekitaare 4-5 hekitaare > 5 hekitaare 	
114.3	Minu umi'nehu noo'ne?	1. Ee 2. Dinonke	
114.4	Qorqorrote mini noo'ne?	1. Ee 2. Dinonke	
114.5	Raadoone noo'ne?	1. Ee 2. Dinoe	
114.6	Bisikiliite noo'ne?	1. Ee 2. Dinoe	
114.7	Hashi diphano noo'ne?	1. Ee 2. Dinoe	
114.8	Fuutunniti firaashe noo'ne?	1. Ee 2. Dinoe	
114.9	Dhoqi dhoqe noo'ne?	1. Ee 2. Dinoe	
114.10	Ofolinani soofi noo'ne?	1. Ee 2. Dinoe	
114.11	Mine'ne caabbichu noo'ne?	1. Ee 2. Dinoe	
114.12	Telezhiine noo'ne?	1. Ee 2. Dinoe	
114.13	Bilbilu/mobayilete silke noo'ne?	1. Ee 2. Dinoe	
T.kiiro	Xa'mo	Doorsha	Sai
Gafa 2 Qa	aqqu/ote orte lainohunni		
201	Qaaqqoho/ote me''e dirosiiti (aganunni)	aganaati	
202	Qaaqqu koo/tee	1. Koo 2. Tee	
203	Konne gedensiidi qaaqoki mama ilita?	 Fayimate uurrinshara Mine 88. Wole/xawisi 	
204	Xaphoomuni meu qaaqqulli noo'ne?	1. Mittu 2. Lamu 3. Sasu 4. Shoolu 88.Wole xawisi	
205	Kuni meekkiki qaaqooti?	 Umikiho Laayinkiho Sayikiho Shoolkina ale 	
206	Qaaqqokki mageeshi yanna keeshite ilootta	 Lamu diri woro Lamu diri ale 	
207	Xaa yannara qaaqiki xisaminoni?	1. Ee 2. Dee'ni	
208	Xisaminoha ikkiro xissosi maati?	 Ganshoho Saamba machete Shekeerete Huffanate 	

		88. wole /xawisi	
209	Qaaqikira 2 lamala woro ikkitano buusano noosini?	1. Ee 2. Dinosi	
210	Foolasi amanyoote keenateni samba miche heerasina hoogasi buuxi(PR/minute) >60 (2 agani worihura) > 50 (2-11 agani qaaqira) > 40 (12-59 agani qaaqira) Qoqqowi! Gagasu giddora hige eino /foolanno woyite fiiqi fiiqi yaano / Busha malaata nooni (Ita woyi aga hooga, qana hooga, dalaga, huxira, afooho wodhi baala tufa)	 Saamba miche dinosi Samba miche noosi Kaajado samba miche noosi woyi Lowo geesha kaajado xisso noosi 	
211	Qaaqiki konni albaani wole xissoni xissame egeninoni?	1. Ee 2. Dee'ni 3. Di afooma	
212	Xa'mo 211 dawaro ee ikkituro maati xisso?	 Huffana Shekeere Kokkerichi Deeu xisso Wole/Xawisi 	
213	Qaaqiki konni alba Hufannate xissoni xissame egenino?	1. Ee 2. Di egenino 3. Di afooma	
Gafa 3: (Qarqarate leeltannori samba-michete xissora reqecci assitanno korka	atubba	
301	Shumate mini noo'ne?	1. Ee 2. Dino	
302	Xa'mo 301 dawaro ee ikkituro hiiteneeti?	2. Dino 1. Mullicho qarqarate haqqenei loonsoonite/Simple pit 2. Woyeesine loonsoonite/VIP 88 Wole/Xawici	
303	Moola ishine hiiko huntinani?	1. Gate hunanni 2. Anfiwa giirani 3. Bale umme hunani 88 Wole/zawisi	
304	Mannunna saada galtanowi baxxinno? Lai	1. Ee 2. Dibaxxinno	
305	Minu iimiido mayiinni fukkaminoho? Lai	1. Buuyotenni 2. Qorqqorotenni 3. Sheekuni 88 Wole xawisi	
306	Minu uulayidosi mayiinni loosaminno? Lai	1. Sabbuni 2. Simmintotenni 3. Haqqunni/leemichoteni 88.Wole xawisi	
307	Mineho masikoote noosi?	1. Ee 2. Dinosi	
308	Mineho me"e kifilla noosi?	 Mitte Lame Sase Shoolenna ali Dinosi 	
309	Gallani mine galtano saada nooni?	1. Ee 2. Dino	312
310	Xa'mo 307 dawaro ee ikkituro hiiti saada	1. Lalu 2. Gereewu 3. Meu 4. Faradu 5. Hare 88 wole/xawisi	

311	Mine sagale loosirate maa horoonsidhinanni?			1.Kasal2.Haqq3.Koree4.Laam5.Saada88Wole xaawi	e e ente ba ute obba si	
312	Mine iibishirateno ikko loosirate kasale mini gi	iddo giidhitinanni?	,	1. Ee 2. Dee'r		314
313	Ee yoottaro xa'mo 312 me''e higgine giidhitan	ani?		1. Barrunni 2. Barrunni 3. Lamu bar 4. Lamalate 5. Lamalate 88 Wole xaw	lamenna hakkuyi ale mitte hige ri giddo mitte hige giddo lame hige giddo mitte hige ⁄isi	
314	Sagale duucha woyite hiikko loosidhinanni? La	i		 Gallanni Kushiinal Minu gob 88 Wole xawis 	mine ho bbanni i	
315	Taje gamba assini woyiite mini giddo wiliile no	o? lai		1. Ee no 2. Dino		
316	Kushiinu nooro me''e masikoote noosi? Lai		1. Mitte 2. Lame 3. Sase 4. Horor	nta dino		
317	Sagale loosatta woyite qaaqqu hiiko heeranno?		1. Amat 2. Sagal 88. Wole xawis	e badhera woyi mulese e loonsanni mini gobbaanni i		
318	Caabbichu wolqa mamiinni afidhinanni?			1. Maas 2. Kurra 3. Haqq 4. Korre 88 .Wole/Xawis	hotenni/Faanosetenni azetenni etenni entetenni si	
319	Mini'ne giddo sigaara wiliishanohu nooni?			1. Ee 2. Dino		321
320	Aliidi xa'mo 319 dawaro ee ikkituro ayeeti wil	iishanohu?		1. Anna 2. Amat 88. Wole/xawis	ho e i	
321	Mine'ne HIV/ADIS xissoni xisaminohu nooni?			1. Ee 2. Dino 88. Dianfooni		323
322	Aliidi xa'mo 321 dawaro ee ikkituro ayeeti?			1. Anna 2. Amat 3. Rodo 88. Wole/xawis	ho e oho/rodoote i	
323	Mine'ne foolate qarri xissoni xisaminohu brochitus, asthma)	nooni? Xawisi	(TB,	1. Ee 2. Dino		401
324	Aliidi xa'mo 323 dawaro ee ikkituro ayeeti xissaminohu?			1. Anna 2. Amat 88. Wole/xawis	ho e i	
Gafa 4: Q	aaqqulleho saga'late ledo amadantanno korkaa	tuba				
401	Qaaqqikki ilami woyite mageeshi kiilosi me"ete	?		Kg		
	Bikkaanchu	Bikka 1	Bikk	a 2	Mereerima guma	
402	Ayirinyesi (Kg)					
403	Hojjasi (cm)					
404	Cigilete mereeri bikko (MUAC)					
405	Lamente lekara darshe herasi buxi					

406	Angasi hoowo mundeete anje noosiha ikkasi Ee lowo buuxi geesha noosi	Mereerimuni noosi	
406	Ilami woyite unuuna qansa mamoote hanafitta?	 Ilamanni heerenna Ilaminkunni mitte saate giddo Ilaminkunni mitte saatenni kayise 24 saate mereero Mittu barri gedenssanni 	
407	Qaaqikira ilaminkunni lewu agani geeshsha maa itisita woyi haayikisita?	 6 agani geesha unuuna calla qanino Unuunanna wolere horonsirinno Unuuna diqaninno 	
408	Qaaqqokki unuuna mageeshi yanna qansootta?	 6 agani woro 12 agani geesha Lame diro Lamu diri ale 	
409	Qaaqikki xaa geesha unuuna qananni nooro 24 saate giddo me''eege qansatta?	 <=5 hige qanseemma 6-10 hige qanseemma >= 11 ale qole qanssemma Diafoomma 	
410	Qaaqikkira wole sagale aa hanafoota? 6 agani ali qaaqqira xa'minanni xa'mo	1. Ee 2. Dee'ni	412
411	Ee yootaro,qaaqqikki diri me''eho umo sagale hanafitta yanna?	 bee in 6 agani woro 6 aganira Lee diro ale higeennasi 	412
412	Qaaqqokki itisate maa horoonsirata?	 Xuunxo Kubbaayya Maanka Angatenni 	
413	Sai 6 agani giddo godowu giddo noo da'muulla shitanno kiniine qaaqqikki egemminno (24-59 agani mereero noo qaaqqulle calla xa'ma)	1. Ee 2. Dee'ni	
414	Sai 6 agani giddo qaaqqikki Vayitaamine(Vit A) adhinno (24-59 agani mereero noo gaaggulle calla xa'ma)	1. Ee 2. Dee'ni	
415	Sai 6 aganni giddo deeo hooltannota zinkete kiniine qaaqqikki adhinno (24-59 agani mereero noo gaaqqulle calla xa'ma)	1. Ee 2. Dee'ni	
Gafa 5: Qa	aqqu kittiwaate adhate gara lainohunni		
501	Qaaqqikki ayi dani dhibba gargartanno kittiwaate ikko qaaqqulle mittenni gamba asine uyinannita kitiwaate adhe egenninno?	1. Ee 2. Diadhinno	
502	Ee yitinnoro qaaqqunnita kitiwaatete kaarde la''anni qaoqqowi	 Penta umita Penta layinkita Penta sayikkita PCV umita PCV layinkita PCV sayikkita Rota umita Rota layinkita Huffananita 	
503	Qaaqqikki gurayidi giwoodira woyi tafira marfetenni qansannita PCV10 kittiwaate pooliyote kittiwaate qaaqqoho afooho coccoinshanni yannara ledo uyinannita adhinno?	1. Ee 2. Diadhinno	
504	Me''e hingeeti PCV10 kittiwaate uyinoonnihu?		
505	9ki aganira uyinannita hufannate kitiwaate qaaqqu adhinno?	1. Ee 2. Diadhinno	
506	Qaaqqu kittiwaate mayiira diadhinno?	 Amate yanna hoogeenna Afa hoogatenni Kittiwaate hooggenna Kittiwaate adhineenna daanno qara waajjatenni Wole xawisi 	
Gala 6. A	muwano samba micne lainonuni noonsa egeno, laona gargadhat	e assano snarro/KAP/	

6.1 Egenon	sa lainohuni/Knowledge		
601	Saamba miche maatiro afoota?	1. Ee 2. Di afooma	603
602	Aliidi xa'mo 601 dawaro ee ikkituro ate hedoni samba mic maati? Boreessi!		003
603	Qarqari'nera qaaquulu samba miche maayine woshinani?		
604	Afootaha ikkiro mashalaqqe mamiichini afirita?	1. Fayimate Ekistenshiine wiini 2. Televiizhinete/Radooneteni 3. Qarqarinkera jajjabu mannu cooyirana 4. Gazeexuni/babbaxitino interneeteni 89. Wole/xawisi	
605	Ate hedoni Saamba michete xissoti malaata maati?	1. Iibabbo 2. Qiida gana 3. Umu damuume 4. Buusano 5. Gagasu giddora hige goshooshama/Chest in drawing 6. Gagasu fiiqi fiiqi yaa/Wheezing 7. Fooate dandaa hooga/Shortness/difficulty in breathing 8. Huxira/Convulsions 9. Mudde mudde folaa/Rapid/fast breathing 10. Daafuraawa/Fatigue/weak 11. Sagale giwisa/Loss of appetite 12. Hunkiisha/Excessive sweating and clammy skin 13. Di afooma 89. Wole/xawisi	
606	Ate hedoni samba michete xissora kaimu maati?	1. Qidaado ayyare /Cold temperature/weather change 2. Buko/Dust/germs 3. Bakiteeriyaho/Bacteria 4. Vaayireseteni/Virus 5. Fangaseteni/Fungi 6. Bitate/Witchcraft 7. Di afooma 89. Wole/xawisi	
607	Saamba miche gargatano kitiwaate noota afoota?	1. Ee 2. Di afooma	
608 609	Ate hedoni samba michete xisso gargadhate maa assa hasiisar Foolate qarri xissuba mannaho sa'anohu mayiiniiti?	 Kitiwaate qasiisa Taaltino sagale itisateni/balanced diet G agani geesha unuuna calla qansateni/Breastfeed exclusively Qiidu yannara iibitano uddano udisiisateni/Clothing warmly during cold season Mini giddo ayyare ikkado eano gede assateni/Proper house ventilation Qaaquwa itisate alba anaga seekke haashirateni/Washing hands before feeding Di afooma 88. Wole/xawisi Ayyarete/Air Kikkisamate/Contact way Xurtino anganni/Dirty hands 88. Wole/xawisi 	
6.2 Xissote	noonsa lao/ Attitude related questions	00. WOR/Adwisi	
610	Saamba michete xisso ma garaini la"atta?	 Qara di afidhinote/Not dangerous Qara afidhinote/Dangerous Lowo geesha qarra afidhinote/Highly dangerous 	

		4. Di afooma	
611	Saamba mciheteni xissamino qaaquwa ollonke hakimete	1. Lowonta sumuu yeema/o	
	minra massa hasiisano	2. Sumuu yeema/o	
		3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5. Horonta sumuu diyeema/o	
612	Budu xagisaano mite woyite samba michete xisso xagise	1. Lowonta sumuu yeema/o	
	hursa dandiitano.	2. Sumuu yeema/o	
		3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5. Horonta sumuu diyeema/o	
613	Amuwu mini giddo qaaquwaho buusano/gansho hakkama	1. Lowonta sumuu yeema/o	
	dandaano	2. Sumuu yeema/o	
		3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5. Horonta sumuu diyeema/o	
614	Mini giddo sigaaru wiliilenna babbaxitino buku	1. Lowonta sumuu yeema/o	
	heedhanoki gede assa samba michete xisso gargarate	2. Sumuu yeema/o	
	kaa'litano	3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5. Horonta sumuu diyeema/o	
615	6 agani geesha amate unuuna calla qansanna kitiwaate	1. Lowonta sumuu yeema/o	
	garuni aa samba michte xisso gargarate kaa'litano	2. Sumuu yeema/o	
		3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5. Horonta sumuu diyeema/o	
616	Saamba michete xissora qaaquuleho kiniine ayee	1. Lowonta sumuu yeema/o	
	suuqenino hidhate dandiinanila/hakimete leelisha	2. Sumuu yeema/o	
	hasiisukini	3. Baxxitino hedo nooe	
		4. Sumuu diyeema/o	
		5 Horonta sumuu diyaama/a	
		J. Horonta suntuu uryeenna/o	
6.3 Garga	dhate assano sharro lainohuni/Practice towards pneumon	a prevention and controls	
6.3 Garga	dhate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino?	ia prevention and controls 1. Ee	
6.3 Garga	dhate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino?	ia prevention and controls 1. Ee 2. Di egenino	
6.3 Garga 617	dhate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino?	ia prevention and controls 1. Ee 2. Di egenino 3. Di afooma	
6.3 Garga 617 618	dhate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero	3. Floring sumul diversion ia prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema	
6.3 Garga 617 618	 Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata 	a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema	
6.3 Garga 617 618	 Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata 	 a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 	
6.3 Garga 617 618	 Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata 	 a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 4. Ammanote annuwi huucitano gede aseema 	
6.3 Garga 617 618	 Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata 	 a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 4. Ammanote annuwi huucitano gede aseema 5. Xagicho suuqeteni hireema 	
6.3 Garga 617 618	 Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata 	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Wole/xawisi 	
6.3 Garga 617 618 619	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate	 a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 4. Ammanote annuwi huucitano gede aseema 5. Xagicho suuqeteni hireema 88. Wole/xawisi	
6.3 Garga 617 618 619	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata?	 a prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 4. Ammanote annuwi huucitano gede aseema 5. Xagicho suuqeteni hireema 88. Wole/xawisi 1. Xissamihuni 24 saate giddo 2. 24 saate gedensaani 	
6.3 Garga 617 618 619	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata?	3. Florinta sumul diversitato ia prevention and controls 1. Ee 2. Di egenino 3. Di afooma 1. Mule noo Fayyimate uurinshawa maseema 2. Budu xagisaanchiwa maseema 3. Xagichu minini xagicho hireema 4. Ammanote annuwi huucitano gede aseema 5. Xagicho suuqeteni hireema 88. Wole/xawisi 1. Xissamihuni 24 saate giddo 2. 24 saate gedensaani	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain 	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Xagicho suuqeteni hireema Wole/xawisi 1. Xissamihuni 24 saate giddo 2. 24 saate gedensaani 1. Xisso gabbi assitano xagga eema/Giving pain killers	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Xagicho suuqeteni hireema Wole/xawisi 1. Xissamihuni 24 saate giddo 2. 24 saate gedensaani 1. Xisso gabbi assitano xagga eema/Giving pain killers 2. Sagale itiseema/aganore eema/Feed child with	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids 	
6.3 Garga 617 618 619 620	Inate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding 	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 	
6.3 Garga 617 618 619 620	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 	
6.3 Gargar 617 618 619 620 621	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 	
6.3 Garga 617 618 619 620 621	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani 1. Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 1. Ee Di afidhino 	
6.3 Garga 617 618 619 620 621 622	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Kagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 1. Ee Di afidhino Ee 	
6.3 Garga 617 618 619 620 621 622	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 1. Ee Di afidhino Ee Dino 	
6.3 Garga 617 618 619 620 621 622 623	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no? Mini giddo wiliilenna buko heedhanoki assateni samba	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Xagicho suuqeteni hireema Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema Budu xawisi Ee Di afidhino Ee Dino 	
6.3 Garga 617 618 619 620 621 622 623 623	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no? Mini giddo wiliilenna buko heedhanoki assateni samba michete xisso ajishate looso loosinooni?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Xagicho suuqeteni hireema Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema Bo Vole /xawisi Ee Di afidhino Ee Dino 	
6.3 Garga 617 618 619 620 621 622 623 624	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no? Mini giddo wiliilenna buko heedhanoki assateni samba michete xisso ajishate looso loosinooni? Bababxitinota budu xagga/ lawi: Tenadam, Damakesa,	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema Xagicho suuqeteni hireema Wole/xawisi 1. Xissamihuni 24 saate giddo 24 saate gedensaani 1. Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema Be Di afidhino 1. Ee Di afidhino 1. Ee Di assinooni Ee 	
6.3 Garga 617 618 619 620 621 622 623 624	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xissamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no? Mini giddo wiliilenna buko heedhanoki assateni samba michete xisso ajishate looso loosinooni? Bababxitinota budu xagga/ lawi: Tenadam, Damakesa, Dingetegna/qaaquule uuyite egenoota?	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 1. Ee Di afidhino Ee Di assinooni Ee Di assinooni 	
6.3 Garga 617 618 619 620 621 622 623 624 625	Ihate assano sharro lainohuni/Practice towards pneumon Qaaquuliki samba michete xissoni xissame egenino? Qaaquuliki Saamba michete xissoni xisamino lawihero woyi xisamiro balaxxe maa assata Qaaquuliki ssamba michete xissoni xisamiro fayimate uurinshawa mamoote haadhe harata? Qaaquuliki samba michete xissoni xisamiro hakimete minira haadhe harata geesha mini giddo maa assatasi? Qaaquuleki kitaawe afidhano asoota? Mini'ne giddo wiliilena buko fulate kaa'litanoti maskoota no? Mini giddo wiliilenna buko heedhanoki assateni samba michete xisso ajishate looso loosinooni? Bababxitinota budu xagga/ lawi: Tenadam, Damakesa, Dingetegna/qaaquule uuyite egenoota? Qaaquuleki ayyare qiidano woyiteno ikko iibitano woyite	 a prevention and controls Ee Di egenino Di afooma Mule noo Fayyimate uurinshawa maseema Budu xagisaanchiwa maseema Xagichu minini xagicho hireema Ammanote annuwi huucitano gede aseema Xagicho suuqeteni hireema 88. Wole/xawisi Xissamihuni 24 saate giddo 24 saate gedensaani Xisso gabbi assitano xagga eema/Giving pain killers Sagale itiseema/aganore eema/Feed child with liquid food/fluids Unuuna qanseema/Breastfeeding Mittoreno di aseema 88. Wole /xawisi 1. Ee Di afidhino Ee Di assinooni Ee Dino 	

Yannaki ootae/oe daafo lowo geeshsha galaxeemmohe!

Annexure G: FGD guide and Consent form for Qualitative data collection (English version)

FGD introduction

Good morning/afternoon and welcome to our session. Thanks for taking the time to join us and to talk about what is childhood pneumonia and how it can be prevented and controlled among the community. My name is ______ and assisting me is ______. We are asked by Mr Derese Desalegn to conduct this FGD with you. The goal of this study is to further develop awareness and prevention strategies on pneumonia among under-5 children in Sidama Regional State, Ethiopia. Pneumonia is number one cause for morbidity and mortality in children. The government tried to prevent and control it through different strategies. We would like to talk with you today to explore your knowledge, attitude and practice regarding childhood pneumonia prevention and control.

Focus groups being used more and more to gain information and to further inform future trials, study planning and the likes. You are the experts; we may learn a lot from you positive or negative. Mothers/Caregivers of individuals with under-5 year children has been invited to participate because you have important knowledge about under-5 pneumonia experiences, needs or perspectives, that we hope to learn more about as a result of our discussion today in this focus group. We are not trying to get everyone to agree or achieve consensus, rather we are gathering information. It is okay if you have different opinions and ideas than the other persons in the group. It is our goal to develop a community-centered awareness and prevention strategies on childhood pneumonia which may help you to prevent and control pneumonia among under-5 year children.

Consent to Participate in Focus Group Discussion

You have been asked to participate in a focus group arranged by Mr Derese Desalegn, PhD student at University of South Africa, department of public health. The purpose of the group is to try and understand what childhood pneumonia is and how can we prevent and control it among the community. The evidence learned in the focus groups will be used to develop awareness and prevention strategies on pneumonia among under-5 children in Sidma Regional State, which may help the community to prevent and control the pneumonia. You can decide whether or not to participate in the focus group and break at any time. Though the focus group will be tape recorded but your reactions/responses will kept anonymous and no names will be mentioned in the report. There is no right or wrong answers to the focus group questions. We want to hear many diverse lookouts and would like to hear from everybody. We hope you can be truthful even when your answers may not be in covenant with the rest of the group. In respect for each other, we ask that only one individual speak at a time in the group and that responses made by all participants be kept confidential. I understand this information and agree to participate fully under the conditions stated above:

Signed Date

Focus group logistics and Ground rules

a) Logistics

- FG may take about an hour and 20 minutes (1:20)
- Feel free to move around
- You can look where the restroom and Exit is
- You can help yourself to refresh

b) Ground Rules

- Everybody should participate and only one person better talks at a time
- It is important for us to hear everyone's thoughts and feelings
- There is no right or wrong answers to questions just ideas, experiences and opinions, which are all valuable

• The session will be audio recorded to help us gather more detailed information about your responses than the handwritten notes that will be taken by investigators, and it will allow us to double check our data for accuracy (if session is audio recorded, which is depending on consent from individual participants).

- Stay with the group, please don't have side conversations and speak clearly to increase recording quality
- Turn off or silence cell phones.
- Each of you has to summarize your opinions in maximum of 7 minutes
- Enjoy the discussions

Do you have any questions before we get started?

Questions for FGD – Mothers or caregivers

1. What do you know about childhood pneumonia and its causes? From where do you get information about the childhood pneumonia
- 2. What do you know about signs and symptoms of childhood pneumonia?
- 3. What is/are treatment for childhood pneumonia?
- 4. What did you do while your child had get pneumonia?
- 5. How can we prevent childhood pneumonia early?
 - Regarding environmental factors
 - Behavioral factors
 - Child related factors
 - Nutritional factors
 - Child's vaccination status
- 6. What will be the challenge to do not prevent childhood pneumonia among the community?

Any questions or comments

Thank you so much for coming and sharing your thoughts and opinions with us. We kindly inform you this is the end of today's group discussion. Have a nice time and thank you once again!

Annexure H: FGD guide and consent form for qualitative data collection (Sidaamu Afoo version)

Gaamote Hasaawira beeqqate sumuumenna kaimu Xa'mo

Keere galtinooni/hosinooni, hawalle hasaawinke battala keerunni daggini yaate banxeemo. Anehu su'mi'ya_______ikkana kaa'laanchi'yahu kayiini______. Techo ki'ne ledo heeranonkeha gaamote hasaawa qineesaate gaamamoomoha ikkana doorinonkehu Kalaa Derrese Dessaalenyiho, Wodiidi Afirikira afantano Yuniversitera 3ki digire rosaanchooti. Rososi gudate xiinxallo assani nooha ikkana techohu gaamote hasaawinkeno tenne xiinxallora ikkitano taje gamba assateeti. Xiinxallo assinannihuno ontu diri wori qaaquuli Saamba michete xisso aana ikkana ki'newiini tini xisso maatiro, hiitooni gargadhinannirona ittisa dandiinanniro mashalaqqe gamba assine albilicho tenne xisso hunanni hayyo kalaqateeti.

Gaamote hasaawi mashalaqe gamba assate, hedo sooriratenna afi'nanni mashalaqe kaimini albillicho hayyo kalaqate kaa'lanno. Ki'ne baalanku doorantinoonihu onti diri wori qaaquulli noo'neereetinna hala'ludu rosichi, egenonna afa hattono hayyo noo'nereeti. Konni kaiminino ki'newiini lowo rosicho afi'neemo yine hendeemo. Hasaawu yannara annu annunku kayisinanninke woyi uyiitinaninke hedo lowo geeshsha hasiisanotenna mittu hedo wolunnite wiini ajjanokite. Togo yineemo woyiiteno ayeetino kaphonna halaaleho yine bandannina agurranni hedo dino, kayiinilla baalanti ka'anno hedo umise deerrini hasiisanote. Konni daafira ayiino hedosi waajikkinina saalikini beeha dandaanonkella. Ayeetino hedo ayiiradote, mittu coyiranno woyite wolu maciisha hasiisano kayini hasaawu birxichi woyi hedo heedhusihu anga kayise adhe hedo aate dandaano.

Gaamote hasaawira sumuu yaatenna giwate kkito

Ki'ne baalunku Kalaa Derrese Dessaalenyihu harisanni noo xiinxallora ikitano mashalaqqe gamba assinani hasaawiraati koyiisantinoonihu. Konni gaamote hasaawihu kaimu onti diri wori qaaquulira samba michete xissohu kaimu maati, hitooni taraawanonna hiissine gargadhinanni yitanotera ikkado mashalaqqe gamba assine kuri kaiminnino albillicho tenne xisso gargadhatenna hunate kaa'litano hayyo/Strategy/ kalaqateeti. Konni hasaawira beeqitinannihu baxxinennna fajjitine calla ikkana hasidhini sa'aatera agurano dandiitinannilla. Taje gamba assinannihuno hinkiilanniiti. Ikkolana ki'newiini gamba assinanni taje ayeerano ewelimma'ne afantanno garini dikullannite. Jeefote xaphi yitino ripoorteno annu annu su'mi kulammikinilla odeesinanni. Konni daafira aye gedee huluulono heedhannera di hasiisanno. Maaxinikini hedo'ne xawisate dandiitinanniha ikkanna wolu yiinota la''e adda asse adha di hasiisanno. Kayini annu annu hedo garunni maciishantannonke gede maciishamannilla hasambeemo.

Konni hasaawira beeqqateni afidhinanniti baxxitino horono hooguro ki'newiini afi'nanni hedo gobbate deerini qaaquullu saamba miche xisso hunate assinani sharrora lowo irko assitano, wole widooni beeqqa hooginirono gatanno'neri dino. Kayiini beeqitinara lowo geesha banxeemmo.

Konni hasaawira beeqqate sumuu yitinanni?

Beeqaanote malaate_____Barra_____

Hasaawaho hasiisanno wodho

- Xaphoomunni hasaawaho 1:20 hasiisankera dandiitano
- Daafurihu suutuni kae milli yee higa dandaano
- Shumateno ikko wole hajara mule noo injo horoonsira dandaano
- Hasaawaho baalunku taru tarunni beeqqa hasiisanno

- · Ayeeti hedo aayiradote yanna uyiine maciisha hasiisano ikkinina tini halaaleho ti'i kaphoho yinaniri dino
- Midaado midaado hinge hashaashama di fajjinani
- Bilbila'ne cancannoki gede asse
- Mittu mittunku hedosi batiriro 7 daqiiqi giddo xaphi assa noosi

Huluullo kalaqi'no'Neri nooro xa'ma dandiitinanni

Hasaawaho kaimu xa'muwa

- 1. Saamba michete xisso maati yitine hedinanni? Maayinni daggano? Mashalaqqe mamiii afidhini?
- 2. Qaaquuleho tenne xissoti malaata maati?
- 3. Maayini xagisate dandiinanni yitine heddinanni?
- 4. Qaaquullu saamba michete xissoni xissamiro maa assitinanni
- 5. Tenne xisso balanxe gargadha dandiinannihu mayiiniiti?
- 6. Xisso tenne gargadhate qarra ikkanori maati yitine heddinanni?

Uyiitinaniti wole baxxitino hedo noo'neni?

Yanna'ne uuyitine beeqqitinooni daafo lowo geeshsha galanteemmo, Keerunni!

Annexure I: Comment and feedback form for Delphi group

COMMENT AND FEEDBACK FORM FOR DELPHI GROUP

Dear Sir/Madam it is a great honour to send you this comment and feedback form. As we discussed via e-mail or phone call I have been developing awareness and prevention strategies on pneumonia among under-five children in Sidama Regional State, Ethiopia as part of my PhD dissertation work at UNISA. Your comments and suggestions as part of Delphi panellist group are expected and will be helpful for the development of these awareness and prevention strategies by integrating with current research findings and the reviewed literatures. So please read the draft document and share me your comment in the following comment and feedback form.

Issues to be considered	Comments	Suggestions	Remark
General opinion on the proposed strategies			
Purpose and objective of strategies			
Integration of findings with the theoretical foundations			
Comprehensiveness of the proposed strategies			
Feasibility of Framework for proposed strategies			
Detail of proposed priority areas for action			
Appropriateness of proposed monitoring and evaluation mechanism and			
indicators			
Any additional points to be considered			

Thank you very much for your consideration! Derese Desalegn (PhD student at University of South Africa) Cell phone no +251913436793 E-mail 62136313@mylife.unisa.ac

Annexure J: Bibliography of Delphi technique individuals

S.No	Profession	Sex	Specialty/education	Position and working place
1	Chief public health expert	М	MPH in General public health	CEO, Hospital
2	Chief public health expert	F	MPH in Reproductive health	NGO, Cluster manager
3	Medical doctor	М	MD	RHB, clinical service
4	Public health specialist	М	MPH in Epidemiology	University, lecturer
5	Nutritionist	М	MSc in Human nutrition	RHB, MNCH coordinator
6	Medical doctor	М	MD, MPH candidate	RHB, clinical case team
7	Public health specialist	М	PhD, Public health	University, lecturer
8	Chief public health expert	М	MPH in health service management	RHB, disease prevention and health
				promotion officer
9	Chief public health expert	М	MPH in health education and behavioral	University, lecturer
			change	

J. Psycho	blogy and Mental Health Care			Copy rights @ Derese Desalegn Buta,
10	Public health specialist	М	PhD, Associate professor	University, lecturer

Annexure K: Direction to assess and classify the childhood pneumonia adapted and modified by the PI based on IMNCI Chart booklet

Direction to assess and classify the childhood pneumonia adapted and modified by the PI based on IMNCI Chart booklet

Ask the	Check for	Assess	Classify				
respondents	general danger	Look, Listen,	Age	We can say			
	signs by DC	Feel, DC		fast	Signs	Classification	Remark
	B			breathing			
Does the child	Not able to	Count the					
have cough or	eat, drink or	breaths in	If the child	50 breaths	Any danger	Severe	Counsel mother
difficult	breast feed	one minute	is 2 months	per minute or	signs, or	pneumonia or	to take the child
breathing?	Lethargic or	Look for	up to 12	more	Chest in	Very severe	to the nearest
For how long	unconscious	chest in	months		drawing or	disease	health Hospital
	Vomits	drawing			Stridor in		or health center
	everything	Look and			calm child		urgently and
	Convulsions	listen for					give the remark
	• Convuisions	stridor	If the child	40 breaths	Fast breathing	Pneumonia	paper
			is 12	per minute or	No signs of	No	If cough for more
			mentheun		INO SIGNS OF	NO	n cough for more
			months up	more	pneumonia or	pneumonia:	than 2 weeks
			to 5 years:		very severe	Cough or cold	counsel them to
						Ŭ	take it health
					disease		facilities, unless
							consul them way
							of prevention

Annexure L: Langauge editorial certificate





Address: -Yeka Sub-City Addis Ababa, Ethiopia

1 June 2021

TO WHOM IT MAY CONCERN

This certificate serves to confirm that I have edited and proofread the Ph.D. thesis entitled "DEVELOPMENT OF AWARENESS AND PREVENTION STRATEGIES ON PNEUMONIA AMONG UNDER FIVE CHILDREN IN SIDAMA REGIONAL STATE, ETHIOPIA", submitted by Mr. DERESE DESALEGN BUTA to the University of South Africa for the award of Doctorate of Literature and Philosophy.

I found the work significant and intriguing to read. Much of my editing basically dealt with dows obstructionist technical aspects, including the grammar of the language which could have s to activ otherwise compromised smooth reading and ended up changing the intended meaning of

the study. I believe the work, after the thorough editorial work, will be found to be of an acceptable standard.

Sincerely,

Solomon GebreMedhin Feyisa

Contacts: tennishsole@gmail.com; Mobile: +251 911 829 507



Mob. +251 911 461804 Tel. +251 116 662110 / +251 116 662113 E-mail:kaleyesusb@gmail.com Addis Ababa, Ethiopia

Activate Windo Go to PC settings to

Annexure M: A sample transcript of FGD1 based on sub-themes

Sub-themes Common responses of participants' General opinion and knowledge of mothers towards causes of average of mothers towards causes of childhood pneumonia

... "The childhood pneumonia can occur within the sudden interaction of hot and cold weather condition. When mothers did not dress their children and probably those exposing the chest of child to cold weather, the child may get pneumonia..." (FGD1 participant 2). ... "The cause for childhood pneumonia cannot be witchcraft, rather it could be poor handling practice that community showed..." (FGD1 participant 3) ... "Pneumonia can be transmitted via air droplet; hence it is the disease of interaction between cold and hot Knowledge of mothers related temperature..." (FGD1 participant 4). with transmission of childhood pneumonia Knowledge of mothers towards ... "We have to immunize our children properly to prevent them from different diseases including childhood pneumonia. prevention childhood Some mothers may delay immunizing their kids and those children can be challenged with diseases..." (FGD1 participant of 5). Pneumonia can be prevented with proper clothing of the child [Du'mu uduunichinni gagasosi seekkine funge amdda pneumonia hasiisanno/qiidu afirannosiki gede/] since from the time of delivery. Attitude towards severity and ... "Pneumonia is highly dangerous which can disturb the health of our children badly. I never forget the history when my child had diseased with pneumonia while his father was far away from us and I had been through a lot of challenges. dangerous of childhood pneumonia The child had become critical, he could not eat and even he could not breathe properly and become lethargic until I took him to health facility. So, I believe that pneumonia is the worst disease for children..." (FGD1 participant 6). Attitude towards prevention and No as such important ideas raised from FGD1 control mechanisms of childhood pneumonia Weather mothers practicing or ... "Childhood pneumonia can be managed by proper clothing of the children when they became sick. I believe that most of the pneumonia cases among under-five children could happen when their mothers did not follow the proper clothing not those childhood pneumonia prevention and control approach, especially when there might be the interaction of hot and cold weather. So, to minimize or manage challenges such like this, it is better to dress children with right clothes according to the weather conditions..." (FGD1 participant mechanism 2).

Annexure N: Sample FGDs discussants on discussion (Pictorial presentation)





Photo credit @Fitsum Dangura & @Researcher

Annexure O: Parental Consent form

I, the Undersigned,

Hereby give my consent that my child, (Child's name) _____ may participate in the research study being conducted by Derese Desalegn Buta, in partial fulfillment of his PhD degree in Public health at UNISA. I understand that the study is focused on my child's experience on the issues of childhood pneumonia.

I acknowledge that I gave my consent willingly and without being unduly influenced to do by any other person.

 Signature_____

 Name______

 Tel No______